

ITEM: 3.23 (ID # 19992) MEETING DATE: Tuesday, May 23, 2023

FROM: FACILITIES MANAGEMENT AND RIVERSIDE UNIVERSITY HEALTH SYSTEMBEHAVIORAL HEALTH:

SUBJECT: FACILITIES MANAGEMENT (FM) AND RIVERSIDE UNIVERSITY HEALTH SYSTEM-BEHAVIORAL HEALTH: Riverside University Health System Behavioral Health Indio Inpatient and Crisis Facility Anti-Ligature Compliance Project – California Environmental Quality Act Exempt, Approval of In-Principle, Preliminary Project Budget, and Approval of Professional Services Agreement for Architectural Services with Prest Vuksic Greenwood Architects, District 4. [\$1,117,895 - 100% RUHS - Behavioral Health Department - General Fund 10000]

RECOMMENDED MOTION: That the Board of Supervisors:

- 1. Approve the Riverside University Health System Behavioral Health Indio Inpatient and Crisis Facility Anti-Ligature Compliance (RUHS-BH Indio Anti-Ligature Compliance) Project for inclusion in the Capital Improvement Program (CIP);
- 2. Find that the Project is exempt from the California Environmental Quality Act (CEQA) pursuant to State CEQA Guidelines Section 15301, Class 1 Existing Facilities Exemption and Section 15061 (b)(3), "Common Sense" Exemption;

Continued on Page 2

ACTION:Policy, CIP

hang, Director Shang 3/16/2023 Rose Salgado, Director of Facilities Management 3/16/2023

MINUTES OF THE BOARD OF SUPERVISORS

On motion of Supervisor Spiegel, seconded by Supervisor Washington and duly carried by unanimous vote, IT WAS ORDERED that the above matter is approved as recommended.

Ayes:

Jeffries, Spiegel, Washington, Perez, and Gutierrez

Navs:

None

Absent:

None

Date:

May 23, 2023

XC:

FM, RUHS-BH

3.23

Kimberly A. Rector

RECOMMENDED MOTION: That the Board of Supervisors:

- Approve in-principle the RUHS-BH Indio Anti-Ligature Compliance Project located at 47915 Oasis Street in Indio, California; to perform interior remodel work in three wings consisting of anti-ligature, plumbing fixture and accessory replacement, miscellaneous remodel upgrades, and accessibility compliance within the remodeled areas;
- 4. Approve a preliminary project budget, in the not to exceed amount of \$1,117,895, for the Project;
- 5. Authorize the use of RUHS-Behavioral Health Department General Fund 10000, in the not to exceed amount of \$1,117,895, including reimbursement to Facilities Management (FM) for incurred project related expenses;
- 6. Approve the attached Professional Services Agreement for architectural services between the County of Riverside (County) and Prest Vuksic Greenwood Architects (Prest Vuksic) of Palm Desert, California, in the not to exceed amount of \$176,100, and authorize the Chairman of the Board (Chairman) to execute the agreement on behalf of the County;
- Authorize the Director of Facilities Management, or her designee, to administer the Professional Services Agreement with Prest Vuksic in accordance with applicable Board policies;
- 8. Delegate project management authority for the Project to the Director of Facilities Management, or her designee, in accordance with applicable Board policies, including the authority to utilize consultants on the approved pre-qualified list for services in connection with the Project, and are within the approved project budget; and
- Authorize the Purchasing Agent to execute pre-qualified consultant service agreements not to exceed \$100,000 per pre-qualified consultant, per fiscal year, in accordance with applicable Board policies for this project, and the sum of all project contracts shall not exceed \$1,117,895.

FINANCIAL DATA	Current Fiscal Year:		Next Fiscal Year:		Total Cost:		Ongoing Cost	
соѕт	\$	447,158	\$	670,737	\$	1,117,895	\$	0
NET COUNTY COST	\$	447,158	\$	670,737	\$	1,117,895	\$	0
SOURCE OF FUNDS: RUHS-Behavioral Health Budget Adjustment: No								
Department - General Fund 10000 – 100%				iscal Year: 22	/23-23/24			

C.E.O. RECOMMENDATION: Approve

BACKGROUND:

Summary

The RUHS-BH Indio Anti-Ligature Compliance Project is located at 47915 Oasis Street in Indio, California. The facility includes an inpatient Psychiatric Health Facility (PHF), Crisis Stabilization Unit (CSU) and Crisis Residential Treatment (CRT).

In an effort to address anti-ligature concerns in all RUHS-BH facilities, this proposal is focused on the county owned facility in Indio that is operated by the Telecare PHF, CSU and RI CRT. The facility service provider and County staff agreed to implement a phased approach to address anti-ligature safety concerns and some of the first priority items have already been implemented.

On June 29, 2021, Item 3.16, the Board of Supervisors (Board) approved a pre-qualified list of architectural and engineering firms to be retained on an as-needed basis. Due to their Health Care Access and Information (HCAI) experience, FM selected Prest Vuksic from the pre-qualified list to provide architectural services to the Project. FM is seeking the Board to approve the professional services agreement with Prest Vuksic in the not to exceed amount of \$176,100 to move forward with the design phase and meet project schedule commitments.

The RUHS-BH Indio Anti-Ligature Compliance Project will complete the final phase of the antiligature compliance work needed for the units within the Indio campus. The scope of the Project includes but is not limited to: interior remodel of various patient occupied areas in three wings within the facility, plumbing fixtures and accessory replacement in shower and restrooms, and mechanical and electrical engineering, and accessibility compliance within the remodeled areas.

(Continued)

BACKGROUND: Summary (Continued)

With certainty, there is no possibility that the RUHS-BH Indio Anti-Ligature Compliance Project may have a significant effect on the environment. The Project, as proposed, is limited to improvements within an existing facility. The use of the facility would continue to provide public services and would not result in a significant increase in capacity or intensity of use. Therefore, the RUHS-BH Indio Anti-Ligature Compliance Project is exempt as the project meets the scope and intent of the "Common Sense" Exemption identified in Section 15061 (b)(3), and Class 1 - Existing Facilities Exemption identified in Section 15301. A Notice of Exemption will be filed by Facilities Management (FM) staff with the County Clerk within five days of Board approval.

FM recommends the Board of Supervisors (Board) approve the RUHS-BH Indio Anti-Ligature Compliance Project, the preliminary project budget in the not to exceed amount of \$1,117,895 and approve the Professional Services Agreement for architectural services with Prest Vuksic in the not to exceed amount of \$176,100. FM will procure the most cost effective and efficient project delivery method and award in accordance with applicable Board policies. Upon approval of plans from HCAI, and associated final construction cost estimate, FM may return to the Board for additional funding.

Impact on Residents and Businesses

The RUHS-BH Indio Anti-Ligature Compliance Project will ensure patient safety and will bring the facility into compliance with current requirements.

Additional Fiscal Information

The approximate allocation of the preliminary project budget is as follows:

BUDGET LINE ITEMS	BUDGET AMOUNT
DESIGN PROFESSIONAL OF RECORD	176,100
SPECIALTY CONSULTANTS	25,500
REGULATORY PERMITTING	11,500
CONSTRUCTION	759,668
COUNTY ADMINISTRATION	43,500
PROJECT CONTINGENCY	101,627
PRELIMINARY PROJECT BUDGET	\$ 1,117,895

(Continued)

Additional Fiscal Information (Continued)

All costs associated with this Board action will be 100% funded with RUHS-Behavioral Health Department - General Fund 10000. Expenditures for FY 22/23 are estimated at \$447,158; and expenditures for FY 23/24 are estimated at \$670,737.

Attachment:

Professional Services Agreement with Prest Vuksic Greenwood Architects

RS:SP:AR:sc:tv; FM05410011587 MT Item #19992
G:\Project Management Office\FORM 11'S\Form 11's_In Process\19992_D7 - 011587 - Indio BH Anti-Ligature Compl - In-Princ,
Prelim Proj Budg & PSA - Vuksic 022823.doc

Kristine Bell-Valdez, Supervising Deputy County County 5/4/2023

Riverside County
Facilities Management
3450 14th Street, 2nd Floor, Riverside, CA 92501

FOR COUNTY CLERK USE ONLY

TM STAFF

TO FILE

NOTICE OF EXEMPTION

September 1, 2022

Project Name: Riverside University Health System (RUHS) Telecare In-Patient Treatment Facility (ITF) Anit-Ligature Compliance, Indio

Project Number: FM05410011587

Project Location: 47915 Oasis Street, north of Avenue 48, Indio, California 92201; Assessor's Parcel Number (APN): 614-150-033

Description of Project: The Indio RUHS Behavioral Health Telecare In-Patient Treatment Facility located at 47915 Oasis Street in Indio, California is an inpatient unit serving consumers on involuntary holds, who are actively a danger to themselves, others, and/or are unable to provide for themselves food, clothing, or shelter. The current property is not in compliance with the current standards for a involuntary hold crisis services facility.

In response to the Joint Commission's Final Report of November 19, 2018 for the In-Patient Treatment Facility in Riverside, Behavioral Health and its safety officer began an evaluation of similar facility components and systems to review updates and improvements for compliance with current anti-ligature regulations at its Indio Telecare In-Patient Treatment Facility. The provider, safety and county staff agreed to implement a phased approach to address anti-ligature safety concerns. Some of the first priority anti-ligature safety items were completed at that time.

The Indio Behavioral Health Telecare ITF Anti-Ligature Compliance Project will complete the final phase of the antiligature compliance work to bring the building up to meet the same anti-ligature standards as other County Emergency Treatment Services and In-Patient Treatment Facilities. Additionally, these improvements are also directed to address antiligature concerns of the voluntary Crisis Residential Treatment program, where consumers are placed in a behavioral health crisis facility in lieu of an involuntary hold.

The scope of the Project includes, but is not limited to, interior remodel of various patient occupied areas in three wings within the facility, plumbing fixtures and accessory replacement in shower and restrooms, mechanical and electrical engineering, and accessibility compliance within the remodeled areas. The anti-ligature compliance improvements at the Indio RUHS IFT is identified as the proposed project under the California Environmental Quality Act (CEQA). The operation of the facility will continue to provide public services and will not result in a change or expansion of existing use. No additional direct or indirect physical environmental impacts are anticipated.

Name of Public Agency Approving Project: Riverside County Facilities Management

Name of Person or Agency Carrying Out Project: Riverside County Facilities Management

Exempt Status: State California Environmental Quality Act (CEQA) Guidelines, Section 15301 Existing Facilities Exemption; 15061(b) (3), General Rule or "Common Sense" Exemption, Codified under Title 14, Articles 5 and 19, Sections 15061, and 15301.

Reasons Why Project is Exempt: The proposed project is categorically exempt from the provisions of CEQA specifically by the State CEQA Guidelines as identified below. The project will not result in any specific or general exceptions to the use of the categorical exemption as detailed under State CEQA Guidelines Section 15300.2. The project will not cause an impact to an environmental resource of hazardous or critical concern nor would the project include unusual circumstances which could have the possibility of having a significant effect on the environment. The project would not result in impacts to scenic highways, hazardous waste sites, historic resources, or other sensitive natural environments, or have a cumulative effect to the environment. No significant environmental impacts are anticipated to occur with the antiligature compliance improvements.

- Section 15301 (b)—Existing Facilities: This Class 1 categorical exemption includes the operation, repair, maintenance, leasing, or minor alteration of existing public or private structures or facilities, provided the exemption only involves negligible or no expansion of the previous site's use. The project, as proposed, is limited to anti-ligature compliance improvements at the existing Indio RUHS ITF to meet current health standards. The use of the facility would continue to provide public services and would not result in a significant increase in capacity or intensity of use. Therefore, the project is exempt as it meets the scope and intent of the Categorical Exemption identified in Section 15301, Article 19, Categorical Exemptions of the CEQA Guidelines.
- Section 15061 (b) (3) "Common Sense" Exemption: In accordance with CEQA, the use of the Common Sense Exemption is based on the "general rule that CEQA applies only to projects which have the potential for causing a significant effect on the environment." State CEQA Guidelines, Section 15061(b) (3). The use of this exemption is appropriate if "it can be seen with certainty that there is no possibility that the activity in question may have a significant effect on the environment." Ibid. This determination is an issue of fact and if sufficient evidence exists in the record that the activity cannot have a significant effect on the environment, then the exemption applies and no further evaluation under CEQA is required. See No Oil, Inc. v. City of Los Angeles (1974) 13 Cal. 3d 68. The ruling in this case stated that if a project falls within a category exempt by administrative regulation or 'it can be seen with certainty that the activity in question will not have a significant effect on the environment', no further agency evaluation is required. With certainty, there is no possibility that the project may have a significant effect on the environment. The proposed anti-ligature compliance improvements within the Indio RUHS will not result in any direct or indirect physical environmental impacts. The improvements would occur within existing facility, would not alter the footprint and are being completed to create a more efficient and functional health facility. The use of the facility for public health services would remain unchanged. Therefore, in no way, would the project as proposed have the potential to cause a significant environmental impact and the project is exempt from further CEQA analysis.

Based upon the identified exemptions above, the County of Riverside, Facilities Management hereby concludes that no physical environmental impacts are anticipated to occur and the project as proposed is exempt under CEQA. No further environmental analysis is warranted.

Signed: _	Mallelle	Date: 9-1-2022	

Mike Sullivan, Senior Environmental Planner County of Riverside, Facilities Management



STANDARD FORM OF PROFESSIONAL SERVICES AGREEMENT BETWEEN COUNTY AND ARCHITECT

by and between

PREST VUKSIK GREENWOOD ARCHITECTS, INC.

(the "Architect")

and

THE COUNTY OF RIVERSIDE

(the "County")

FOR:

RIVERSIDE UNIVERSITY HEALTH SYSTEM BEHAVIORAL HEALTH INDIO
INPATIENT AND CRISIS FACILITY ANTI-LIGATURE COMPLIANCE PROJECT
(RUHS-BH INDIO ANTI-LIGATURE COMPLIANCE PROJECT)

FM05410011587

47915 OASIS STREET, INDIO, CA 92201

MAY 23 2023 3.23

TABLE OF CONTENTS

		<u>.</u>	Page
ARTICI E 1 G	ENERAL	PROVISIONS	1
1.1	DEFIN	IITIONS	
	1.1.1	Acceptance	2
	1.1.2	Additional Insured	2
	1.1.3	Additional Services.	
	1.1.4	Additional Services Compensation.	2
	1.1.5	Additional Services Fees.	2
	1.1.6	Agreement	2
	1.1.7	Applicable Laws	2
	1.1.8	Application for Payment	
	1.1.9	Architect	
	1.1.10	Architect's Own Expense	2
	1.1.11	Architect's Project Schedule	2
	1.1.12	Architect's Representative	2
	1.1.13	Basic Services	2
	1.1.14	Basic Services Allowance	3
	1.1.15	Basic Services Compensation.	3
	1.1.16	Basic Services Fees.	3
	1.1.17	Bidding Phase	3
		Board of Supervisors.	
		Close-Out Completion	
		Close-Out Documents	
	1.1.21	Conflicting Interpretations.	3
		Construction Contract.	
		Construction Costs	
		Construction Documents	
		Construction Phase	
		Contract Documents.	
		Contractor.	
		County	
		County Consultant.	
		Day	
	1.1.31	Defective Work	4
		Delay.	
		Design Coordination Standards	
		Design Development Documents.	
		Design Development Phase.	
		Design Documents	
		Design Elements.	
		Director of Facilities Management.	
		Disability Laws.	
		Drawings.	
		Effective Date	
		Environmental Laws	
		Estimate of Construction Costs	
		Excluded Services.	
		Existing Improvements	
		Final Construction Documents.	
		Final Construction Documents Phase.	

	1.1.49		
	1.1.50		
	1.1.51		
	1.1.52		
	1.1.53	Good Faith Determination	6
	1.1.54	Governmental Authority.	7
	1.1.55	Hazardous Substance	7
	1.1.56	Hourly Rates.	7
	1.1.57	Indemnitees	
	1.1.58	Initial Program.	
	1.1.59	Intellectual Property Rights	
	1.1.60	Interest Rate	
	1.1.61	Invoice for Payment.	
	1.1.62	Key Personnel, Key Person.	7
	1.1.63		7
	1.1.64	Master Project Schedule	
	1.1.65	Maximum Hourly Fee.	
	1.1.66		
		Mold.	
	1.1.67	Notice of Completion	
	1.1.68	Payment Schedule	8
	1.1.69	Period of Inactivity	8
		Phase.	
	1.1.71	Planning/Programming Phase.	8
	1.1.72		
	1.1.73		
		Program.	
		Project	
		Project Team	
	1.1.77	Proprietary Information.	9
	1.1.78	Record Documents.	9
	1.1.79	Record Drawings, Record Specifications.	9
	1.1.80	Reference Documents.	9
	1.1.81	Reimbursable Expenses.	
	1.1.82	Restrictive Specifications.	
	1.1.83	Schematic Design Documents	
	1.1.84	Schematic Design Phase	
	1.1.85	Separate Contractor	
	1.1.86	Site.	
	1.1.87	Specifications.	
	1.1.88	Subconsultant	
	1.1.89		
		Submittal	_
	1.1.91	Substitution. 10	_
		Tier	
		Work	
1.2	INCOR	PORATION OF RECITALS10)
1.3	PERFO	RMANCE STANDARD10)
1.4	AUTHO	PRITY OF ARCHITECT10)
	ARCHI	TECT'S PERSONNEL10)
	1.5.1	Commitment, Cooperation	
	1.5.2	Key Personnel	
	1.5.3	Architect's Representative	1

1.5

	1.5.4	Sole Responsibility	11
1.6	SUBC	ONSULTANTS	12
	1.6.1 1.6.2 1.6.3 1.6.4 1.6.5 1.6.6 1.6.7	Retention	12 12 12 12
1.7	OWNE	ERSHIP OF DOCUMENTS	13
	1.7.1 1.7.2 1.7.3 1.7.4 1.7.5 1.7.6 1.7.7 1.7.8 1.7.9	Property of County. Assignment of Rights. County Use. No Infringement. Distribution. Inspection. Delivery. Indemnification. Disputes.	13 13 13 13 14 14
1.8	APPLI	CABLE LAWS	14
	1.8.1 1.8.2 1.8.3 1.8.4 1.8.5 1.8.6	Compliance by Architect Changes in Laws Disability Laws Conflicts in Interpretations Legal Notices Direct Communications	14 14 14 14
1.9		DF ESSENCE	
ARTICLE 2 B	ASIC SE	RVICES	15
2.1		SERVICES, ADDITIONAL SERVICES, EXCLUDED SERVICES	
	2.1.1 2.1.2 2.1.3 2.1.4	Basic Services Additional Services. Excluded Services. Phases.	. 15 . 15
2.2	PROG	RAM	. 15
	2.2.1 2.2.2 2.2.3 2.2.4 2.2.5	Initial Program. Final Program. Changes. Notice. Integrated Design.	. 15 . 15 . 15
2.3	FIXED	LIMIT	. 16
	2.3.1 2.3.2 2.3.3 2.3.4 2.3.5 2.3.6 2.3.7 2.3.8	Fixed Limit Amount. Architect's Obligation. Review of Estimates. Fixed Limit Adjustments. Notice of Overruns. Redesign Services. Architect's Own Expense. Non-Programmatic Revisions.	. 16 . 16 . 16 . 17 . 17

	2.3.9 2.3.10 2.3.11	Basic Services Continuous Performance No Limitation.	17
2.4		ON DOCUMENT SUBMISSIONS	
	2.4.1 2.4.2 2.4.3 2.4.4 2.4.5 2.4.6	Submission to County. County Approval. Governmental Authorities. Corrections. Restrictive Specifications. Legal Review.	18 18 18 18
2.5	CONS	TRUCTION DOCUMENTS	
2.6	DESIG	N COORDINATION STANDARDS	19
2.7	COMM	IUNICATIONS	19
2.8	CONS	TRUCTION MEANS, METHODS AND SAFETY	19
2.9	REJEC	CTION OF WORK	19
	2.9.1 2.9.2 2.9.3 2.9.4 2.9.5	Recommendations by Architect. County Approval. No Duty of County. Contractor Responsibility. No Authority to Stop Work.	19 20 20
2.10	PERMI	TS, APPROVALS	20
		Submission to Governmental Authorities. Off-Site Improvements. Notice to County. Expiration.	20 20
2.11	TESTIN	NG AND INSPECTIONS	20
	2.11.2 2.11.3 2.11.4	Recommendations by Architect. Review of Reports. Additional Inspection, Testing. No Duty of County. Defective Work.	20 21 21
2.12	COUNT	TY CONSULTANTS	. 21
	2.12.2	Retention	.21
2.13	MEETIN	NGS	. 22
	2.13.2 2.13.3 2.13.4 2.13.5	Basic Services Participants Conduct of Meetings Action Items Meeting Minutes Numerical Limitations	.22 .22 .22
2.14	SITE AN	ND EXISTING IMPROVEMENTS	. 23
2.15	ELECT	RONIC PROGRAM MANAGEMENT	. 23

	2.16	TIME A	AND SCHEDULE	24
		2.16.4 2.16.5	Architect's Project Schedule. Time Extensions. Schedule Updates. County Approvals. County Review. No Delay by Architect.	24
	2.17	SUSTA	NNABLE DESIGN REQUIREMENTS	. 25
			LEED Accreditation	
ARTIC	LE 3 AD	DITION	AL SERVICES	. 25
	3.1	DEFIN	TION	. 25
	3.2	ENUM	ERATION OF ADDITIONAL SERVICES	. 26
	3.3	AUTHO	PRIZATION BY COUNTY	. 26
	3.4	NOTIC	E TO COUNTY	. 27
	3.5	DISPU [*]	TES	. 27
ARTICL	E 4 CO	UNTY R	ESPONSIBILITIES	. 27
	4.1	ADMIN	ISTRATION BY COUNTY	. 27
		4.1.1 4.1.2 4.1.3 4.1.4	County Requirements. County Approvals. Director of Facilities Management. Board Approval.	. 27 . 27
	4.2	SURVE	YS, SERVICES AND REPORTS	27
		4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 4.2.6 4.2.7 4.2.8	Reference Documents Testing and Inspection. Land Survey. Geotechnical. County's Expense. No Warranty by County. Notice of Defects. County-Furnished Designs.	28 28 28 28 28 28
	4.3	ACCES	S TO SITE	29
	4.4	APPRO	VALS	29
ARTICL	E 5 ARG	CHITEC	T'S COMPENSATION	29
	5.1	BASIC	SERVICES COMPENSATION	29
		5.1.2 5.1.3 5.1.4	Total Compensation Basic Services Fees Guaranteed Amounts Basic Services Allowances Design Contingency Allowance	29 30 30

5.2	REIMBURSABLE EXPENSES	31
	5.2.1 Exclusive List	31
5.3	- ·	
	5.3.1 Additional Services Compensation. 5.3.2 Additional Services Fees. 5.3.3 Direct Engagement.	32
5.4	HOURLY RATES	32
5.5	RELEASE FOR PRIOR SERVICES	32
5.6	DISPUTES	32
5.7	NO WAIVER OR RELEASE OF RIGHTS	33
ARTICLE 6	PAYMENTS TO ARCHITECT	33
6.1	INVOICES FOR PAYMENT	33
	6.1.1 Invoice Submission.	33
6.2	PAYMENT SCHEDULE FOR BASIC SERVICES	34
6.3	ACCOMPANYING DOCUMENTATION	34
6.4	REVIEW AND PAYMENT	35
	6.4.1 Review by County	
6.5	PAYMENT DISPUTES	35
6.6	WITHHOLDING BY COUNTY	35
6.7	LIENS, STOP PAYMENT NOTICES, CLAIMS	35
ARTICLE 7	ACCOUNTING, RECORDS AND AUDIT	36
7.1	FINANCIAL MANAGEMENT	36
7.2	RECORD KEEPING	36
	7.2.1 Books and Records	
7.3	INSPECTION, PRODUCTION AND AUDITING	36
7.4	NONCOMPLIANCE BY ARCHITECT	37
	7.4.1 Cost of Audit	
	7.4.2 County Remedies	
	7.4.5 Willinolating	37

7.5	SUBC	ONSULTANTS	37
ARTICLE 8 D	EFAULT,	TERMINATION AND SUSPENSION	38
8.1	TERM	INATION BY COUNTY FOR CAUSE	38
	8.1.1 8.1.2 8.1.3 8.1.4 8.1.5 8.1.6 8.1.7	Default by Architect. Remedies Upon Default. Rights Cumulative. Disability, Insolvency. Architect Obligations. Completion by County. Payment to Architect.	38 38 38 38
8.2	TERM	INATION WITHOUT CAUSE	
	8.2.1 8.2.2	Termination for Convenience. Payment to Architect.	
8.3	SUSPE	ENSION BY COUNTY	40
8.4	TERMI	INATION BY ARCHITECT	40
	8.4.1 8.4.2	Architect's Remedies. Payment to Architect.	
ARTICLE 9 IN	IDEMNIF	ICATION	41
9.1	INDEM	INIFICATION BY ARCHITECT	41
	9.1.1 9.1.2 9.1.3 9.1.4 9.1.5 9.1.6	Basic Indemnity. Indemnity for Design Professional Services. Limitations on Indemnity Obligation. Subconsultant Indemnity Agreements. No Limitation by Insurance. Enforcement.	41 41 41
ARTICLE 10 II	NSURAN	CE	42
10.1	ARCHI	TECT'S INSURANCE	42
	10.1.2 10.1.3 10.1.4 10.1.5 10.1.6 10.1.7	Required Coverages. Notice of Cancellation. Additional Insureds. Self Insured Retentions. Certificates of Insurance. Waiver of Subrogation. Lapse in Coverage. Subconsultants.	43 43 43 44
ARTICLE 11 D		RESOLUTION	
11.1	RESOL	UTION OF DISPUTES	44
11.2	GOOD	FAITH DETERMINATIONS	44
11.3	ATTOR	NEY'S FEES	44
ARTICLE 12 R	OYALTIE	ES, PATENTS, COPYRIGHTS AND TRADE SECRETS	45
12.1	ROYAL	TIES	45
12.2	INFRIN	GEMENT	45

12.3	NOTICE BY ARCHITECT	45
ARTICLE 13 M	MISCELLANEOUS	45
13.1	GOVERNING LAW AND VENUE	45
13.2	HAZARDOUS SUBSTANCES	
	13.2.1 Introduction by Architect	
13.3	NO WAIVER	46
13.4	NO THIRD-PARTY RIGHTS	46
13.5	EXTENT OF AGREEMENT	46
13.6	SUCCESSORS AND ASSIGNS	46
13.7	CONFIDENTIALITY	47
13.8	INDEPENDENT CONTRACTOR	47
13.9	ARCHITECT'S REPRESENTATIONS	47
13.10	SURVIVAL	48
13.11	SEVERABILITY	48
13.12	INTERPRETATION	48
13.13	ADVERTISING	48
13.14	ELECTRONIC DOCUMENTS	48
13.15	EXECUTION	48
13.16	COUNTERPARTS	48
13.17	TITLES FOR CONVENIENCE	49
13.18	NONDISCRIMINATION	49
13.19	SERVICES PERFORMED BY ARCHITECT OR AFFILIATES	49
13.20	REBATES, KICKBACKS	49
ARTICLE 14 N	IOTICES	50
14.1	DELIVERY AND ADDRESSES	
	14.1.1 Delivery	
14.2	CHANGE OF ADDRESS	51
ARTICLE 15 E	XHIBITS	51

STANDARD FORM OF PROFESSIONAL SERVICES AGREEMENT BETWEEN COUNTY AND ARCHITECT



PREAMBLE

THIS STANDARD FORM OF PROFESSIONAL SERVICES AGREEMENT BETWEEN COUNTY AND ARCHITECT ("Agreement") is entered into on this ______ day of ______, 2023 by and between THE COUNTY OF RIVERSIDE, a political subdivision of the State of California, ("County") and PREST VUKSIC GREENWOOD ARCHITECTS, INC., a California Corporation, ("Architect"), for professional services.

RECITALS

- A. County is the legal owner of the parcel of property, described more particularly in the <u>Property Description Exhibit "A"</u> attached hereto, located at the following address or public road or highway intersections: **47915 Oasis Street, Indio, CA 92201**.
- **B.** Architect represents it has the background, knowledge, licensing, experience and expertise necessary to provide the professional services and other services and things required by this Agreement.
- **C.** County and Architect desire to enter into this Agreement for Architect to provide County, without limitation, professional services in connection with the design and construction of the Project generally described on the cover page to this Agreement.

TERMS AND CONDITIONS

NOW, THEREFORE, in consideration of the mutual promises and covenants contained herein and other valuable consideration, receipt of which is hereby acknowledged by their signatures below, it is mutually agreed by and between the undersigned as follows:

ARTICLE 1 GENERAL PROVISIONS

1.1 **DEFINITIONS**

Capitalized terms (and, if otherwise stated in this <u>Section 1.1</u>, non-capitalized terms) used in this Agreement shall have the meanings assigned to them in this <u>Section 1.1</u>. If not defined in this Section, they shall have the meanings assigned to them in the <u>General Conditions of the Standard Form of Construction Contract Between County and Contractor - Exhibit "K"</u> attached hereto. If not defined in this Agreement or the General Conditions, they shall have the meanings reasonably understood to apply to them by the context in which they are used. If a term is defined both in this <u>Section 1.1</u> and the General Conditions and the definitions are not identical, the definition set forth in this <u>Section 1.1</u> shall, unless otherwise stated in this Agreement, govern for purposes of interpreting Architect's obligations under this Agreement. Terms that are phrased in the singular shall be deemed to include the plural, and vice versa, where appropriate to their context.

1.1.1 Acceptance. "Acceptance" means the point at which the Project is formally accepted by the Board of Supervisors and a Notice of Completion is recorded by County.

- **1.1.2** Additional Insured. "Additional Insured" means each of the Indemnitees and other persons or entities that, under the terms of this Agreement and its exhibits, the Architect or its Subconsultants are required to name as an additional insured under their policies of insurance.
- **1.1.3** Additional Services. "Additional Services" means the services described or referenced in <u>Article</u> 3 of this Agreement.
- **1.1.4 Additional Services Compensation.** "Additional Services Compensation" means the compensation that is payable to Architect under this Agreement for its performance of authorized Additional Services performed in accordance with this Agreement, which compensation consists of two components: Additional Services Fees and Reimbursable Expenses.
- **1.1.5** Additional Services Fees. "Additional Services Fees" means those fees for services payable to Architect under this Agreement for authorized Additional Services performed in accordance with this Agreement.
- **1.1.6 Agreement.** "Agreement" means this Standard Form of Professional Services Agreement Between County and Architect.
- 1.1.7 Applicable Laws. "Applicable Laws" means all statutes, ordinances, regulations, policies and guidelines (including, without limitation, Environmental Laws and Disability Laws) enacted by Governmental Authorities, codes adopted or promulgated by Governmental Authorities (including, without limitation, building and health and safety codes), lawful orders of Governmental Authorities and common law, including, but not limited to, principles of equity applied by the courts, which are in effect at the time the services or other obligations that are required under or in connection with the performance of this Agreement are performed; provided, however, that the term Applicable Laws when used in this Agreement to refer to the Architect's obligation to comply with Applicable Laws is intended to be limited to those laws, statutes, ordinances, policies, guidelines, regulations, codes, orders and principles that are relevant and applicable to the scope of services that are required of Architect or the Subconsultants under this Agreement.
- **1.1.8 Application for Payment.** "Application for Payment" means the Contractor's or a Separate Contractor's itemized application for payment for Work.
- **1.1.9 Architect.** "Architect" means the person or entity identified as "Architect" in the preamble to this Agreement.
- **1.1.10 Architect's Own Expense.** "Architect's Own Expense" means that Architect agrees to assume sole responsibility to pay and be responsible for any resulting or associated Loss and Delay, without any form of compensation or reimbursement, of any kind, by County.
- **1.1.11 Architect's Project Schedule.** "Architect's Project Schedule" means a detailed schedule prepared by Architect depicting Architect's proposed time plan for performance of this Agreement, as adjusted for extensions of time permitted by this Agreement.
- **1.1.12** Architect's Representative. "Architect's Representative" means the individual acting on behalf of Architect as its representative with the authority set forth in Paragraph 1.5.3 of this Agreement.
- **1.1.13 Basic Services**. "Basic Services" means the services required by <u>Article 2</u> of this Agreement to be performed by Architect or described elsewhere in this Agreement or its exhibits as services that are part of Basic Services, including, without limitation, those services marked by a "yes" designation in the <u>Description of Basic, Additional and Excluded Services Exhibit "B" attached hereto.</u>

- **1.1.14 Basic Services Allowance.** "Basic Services Allowance" means a dollar amount reserved in the Basic Services Compensation for performance of Basic Services with respect to which the County and Architect have agreed, in lieu of a Fixed Fee or Maximum Hourly Fee, to a budgeted amount that is not guaranteed by Architect and that may, with prior approval by County, be exceeded by Architect.
- **1.1.15 Basic Services Compensation.** "Basic Services Compensation" means the compensation that is payable to Architect under this Agreement for its performance of Basic Services in accordance with this Agreement, which compensation consists of two components: Basic Services Fees and Reimbursable Expenses.
- **1.1.16 Basic Services Fees.** "Basic Services Fees" means those fees for services, as distinguished from reimbursement of costs, that are payable to Architect under this Agreement as part of the Basic Services Compensation, which may be in the form of a Fixed Fee, a Maximum Hourly Fee or a Basic Services Allowance.
- **1.1.17 Bidding Phase.** "Bidding Phase" means the Phase of Architect's Basic Services described in Paragraph 2.1.6 of the Description of Basic, Additional and Excluded Services Exhibit "B" attached hereto.
- **1.1.18** Board of Supervisors. "Board of Supervisors" means the Board of Supervisors for the County of Riverside.
- **1.1.19 Close-Out Completion**. "Close-Out Completion" means the point at which: (1) all conditions set forth in the Contract Documents for Substantial Completion and Final Completion of the entirety of Work to be performed by the Contractor or a Separate Contractor have been, and continue to be, fully satisfied; and (2) all Close-Out Documents have been received by County.
- **1.1.20 Close-Out Documents.** "Close-Out Documents" means all documents (including, without limitation, paper and electronic versions) and other things that are required under the terms of the Contract Documents to be submitted by Contractor or a Separate Contractor after Final Completion and as a condition of Final Payment to the Contractor or a Separate Contractor, including, without limitation, Record Documents, warranties, guarantees, technical and product information, product samples, operations and maintenance manuals and excess, replacement and attic stock parts and materials.
- **1.1.21 Conflicting Interpretations.** "Conflicting Interpretations" means an express and specific interpretation by a Governmental Authority, or an official thereof acting in his or her official capacity, of a requirement of an Applicable Law as applied to a specific and discrete portion of the Design Documents or a Submittal and that either (1) conflicts with an interpretation of the same Applicable Law by Architect that is made in accordance with the standard of care set forth in <u>Section 1.3</u>, below, or (2) is consistent with an interpretation of the same Applicable Law by Architect that is made in accordance with the standard of care set forth in <u>Section 1.3</u>, below, but conflicts with an interpretation of the same Applicable Law by another Governmental Authority or an official of the same or a different Governmental Authority acting in his or her official capacity. For purposes of this definition, issuance by a Governmental Authority of a general approval or permitting does not constitute an express and specific interpretation by an official of a Governmental Authority and shall not be considered sufficient to form the basis for a Conflicting Interpretation.
- **1.1.22 Construction Contract.** "Construction Contract" means the written contract executed between County and the Contractor or a Separate Contractor for construction of all or a portion of the Work.
- 1.1.23 Construction Costs. "Construction Costs" means the total costs, whether estimated or actual, to construct those elements of the Project designed or specified by Architect or its Subconsultants, inclusive of overhead and profit to the Contractor and Separate Contractors performing the Work, but exclusive of the following: (1) reserves established for use by County; (2) the fees and costs of professional services of Architect, Subconsultants or County Consultants; (3) land acquisition costs; (4) finance costs; (5) County's

administrative costs; and (6) legal fees and costs.

- **1.1.24 Construction Documents.** "Construction Documents" means progressive and final versions of those Design Documents delineated in <u>Description of Basic, Additional and Excluded Services Exhibit "B"</u> and <u>Construction Document Deliverables Exhibit "O"</u> attached hereto that are required to be provided by Architect as part of Basic Services performed during the Design Development Phase or Final Construction Documents Phase.
- **1.1.25 Construction Phase.** "Construction Phase" means the Phase of Architect's Basic Services described in <u>Paragraph 2.1.7</u> of the <u>Description of Basic, Additional and Excluded Services Exhibit "B"</u> attached hereto.
- 1.1.26 Contract Documents. "Contract Documents" means the following collection of documents as they may pertain to the Work to be performed by Contractor or a Separate Contractor under a Construction Contract: (1) the Construction Contract; (2) Addenda; (3) General Conditions; (4) Specifications; (5) Plans and Drawings; (6) Modifications; (7) Reference Documents (as defined in the General Conditions); (8) Change Orders; (9) Unilateral Change Orders; (10) Construction Change Directives; and (11) other documents that comprise exhibits, attachments or riders to the documents listed in preceding Clauses (1) through (10).
- **1.1.27 Contractor.** "Contractor" means the individual or firm under a Construction Contract with County to serve as the principal, supervising general contractor for construction of the Project.
- 1.1.28 County. "County" means the County of Riverside, a political subdivision of the State of California.
- **1.1.29 County Consultant.** "County Consultant" means a professional, of any Tier, retained by County to provide professional services, other than those provided by Architect or its Subconsultants, including, without limitation, those County Consultants listed in the <u>Subconsultants and County Consultants List Exhibit "F"</u> attached hereto.
- **1.1.30** Day. "Day" means, whether capitalized or not, calendar day, including weekends and legal holidays, unless otherwise specifically stated to be a working or business day.
- **1.1.31 Defective Work.** "Defective Work" means Work by the Contractor or a Separate Contractor that is: (1) faulty, defective or deficient; or (2) does not conform to Applicable Laws, the Contract Documents, the directives of County or Architect issued in accordance with the Contract Documents or the requirements of any inspection, reference standard, test, code or approval specified in the Contract Documents.
- 1.1.32 Delay. "Delay" means any circumstances involving delay, disruption, hindrance or interference.
- 1.1.33 Design Coordination Standards. "Design Coordination Standards" means the following standards for coordination that must be met in the preparation of Design Documents submitted by Architect to County: (1) proper coordination and relationship among the Design Elements; (b) proper coordination between the Design Elements and the reasonably foreseeable requirements for utilization of the Site during construction by County, Contractor and Separate Contractors; (c) proper coordination among the Design Elements, the conditions at the Site, and Existing Improvements that are reasonably observable or that are disclosed by information reasonably accessible to Architect; and (d) a system of references and cross-references that is thorough, accurate, and consistent.
- **1.1.34 Design Development Documents.** "Design Development Documents" means the progressive iterations of the Construction Documents developed by the Architect and the Subconsultants as part of the Design Development Phase.

- **1.1.35 Design Development Phase**. "Design Development Phase" means the Phase of Architect's Basic Services described in <u>Paragraph 2.1.4</u> of the <u>Description of Basic</u>, <u>Additional and Excluded Services</u> Exhibit "B" attached hereto.
- **1.1.36 Design Documents.** "Design Documents" means all originals, copies and drafts (whether paper or electronic) of plans, drawings, tracings, specifications, programs, reports, calculations, presentation materials, samples, models and other material containing designs, specifications, engineering or other information prepared by Architect or its Subconsultants for the Project, including all building and other design contained therein.
- **1.1.37 Design Elements.** "Design Elements" means engineered and non-engineered designs, narratives, backgrounds, area plans, floor plans, details, sections, elevations, material and equipment schedules, products, quantities, dimensions, measurements, points of reference, sizes, capacities, performance characteristics, industry, professional and manufacturer standards, general and specific notes and legends.
- **1.1.38 Director of Facilities Management.** "Director of Facilities Management" means the Director of the County of Riverside Facilities Management Department or his/her designee.
- 1.1.39 Disability Laws. "Disability Laws" means applicable federal, state, local or municipal laws, rules, orders, regulations, statutes, ordinances, codes, decrees, or requirements of any Governmental Authority, which regulate, relate to or impose liability or standards of conduct with respect to, or accessibility for, persons with disabilities, including, without limitation, the Americans with Disabilities Act (42 USCA §§ 12101 et seq.) and the Fair Housing Amendments Act of 1988 (42 USCA §§ 3604 et seq.).
- **1.1.40 Drawings.** "Drawings" means the graphic and pictorial portions of the Design Documents showing the design, location and dimensions of the Work, including plans, elevations, details, schedules and diagrams. The term "Drawings" is used interchangeably with "Plans."
- **1.1.41** Effective Date. "Effective Date" means the date set forth in the Preamble to this Agreement, or, if none is set forth, the date that this Agreement is signed by the County.
- 1.1.42 Environmental Laws. "Environmental Laws" means applicable federal, state, local or municipal laws, rules, orders, regulations, statutes, ordinances, codes, decrees, or requirements of any Governmental Authority, which regulate, relate to, or impose liability or standards of conduct concerning any Hazardous Substance (including, without limitation, the use, handling, transportation, production, disposal, discharge or storage thereof), occupational or environmental conditions on, under, or about the Site or Existing Improvements (including, without limitation, soil, groundwater, and indoor and ambient air conditions), environmental protection (natural or manmade resources), or occupational health or industrial hygiene (but only to the extent related to Hazardous Substances on, under, or about the Site or Existing Improvements). as now or may at any later time be in effect, including, without limitation: the Comprehensive Environmental Response, Compensation and Liability Act of 1980 [42 U.S.C.A. §§ 9601 et seq.]; the Resource Conservation and Recovery Act of 1976 [42 U.S.C.A. §§ 6901 et seq.]; the Clean Water Act (also known as the Federal Water Pollution Control Act) [33 U.S.C.A. §§ 1251 et seq.]; the Toxic Substances Control Act [15 U.S.C.A. §§ 2601 et seq.]; the Hazardous Materials Transportation Act [49 U.S.C.A. §§ 1801 et seq.]; the Insecticide, Fungicide, Rodenticide Act [7 U.S.C.A. §§ 136 et seq.]; the Superfund Amendments and Reauthorization Act [42 U.S.C.A. §§ 6901 et seq.]; the Clean Air Act [42 U.S.C.A. §§ 7401 et seq.]; the Safe Drinking Water Act [42 U.S.C.A. §§ 300f et seq.]; the Solid Waste Disposal Act [42 U.S.C.A. §§ 6901 et seq.]; the Surface Mining Control and Reclamation Act [30 U.S.C.A. §§ 1201 et seq.]; the Emergency Planning and Community Right to Know Act [42 U.S.C.A. §§ 11001 et seq.]; the Occupational Safety and Health Act [29 U.S.C.A. §§ 655 and 657]; the Residential Lead-Based Paint Exposure Act (Title X of the Housing and Community Development Act of 1992) [15 U.S.C.A. §§ 2681 et seq.]; the Lead-Based Paint Poisoning Prevention Act [42 U.S.C.A. §§4821 et seq.], the Federal Endangered Species Act, the California Endangered Species Act, the Migratory Bird Treaty Act, the National Environmental Policy Act, the

California Environmental Quality Act, Porter Cologne Water Quality Act (California Water Code §§ 13000 et seq), and all similar federal, state or local laws, rules, orders, regulations, statutes, ordinances, codes, decrees, or requirements.

- **1.1.43 Estimate of Construction Costs.** "Estimate of Construction Costs" means a written estimate of the reasonable, anticipated Construction Costs for construction of the Project as depicted or described in the Design Documents prepared by Architect and its Subconsultants.
- **1.1.44 Excluded Services.** "Excluded Services" means those design and engineering services so designated and listed in the <u>Description of Basic</u>, <u>Additional and Excluded Services Exhibit "B"</u> attached hereto, that are excluded from the scope of Basic Services, and which shall be performed only if, when and on such terms as may be hereafter mutually agreed in writing between the County and the Architect.
- **1.1.45** Existing Improvements. "Existing Improvements" means improvements located on the Site as of the Effective Date, whether above or below the surface of the ground, including, but not limited to, existing buildings, utilities, infrastructure improvements and other facilities.
- 1.1.46 Final Construction Documents. "Final Construction Documents" means the final version of the Construction Documents at the point that: (1) they are sufficiently detailed so as to be capable of definitive and final pricing with a minimum of further clarifications or detailing by Architect or the Subconsultants; and (2) all corrections to the Construction Documents required by County or by Governmental Authorities for permitting or approval have been completed and the Construction Documents have been unconditionally approved for use in construction by the County and Governmental Authorities.
- **1.1.47 Final Construction Documents Phase.** "Final Construction Documents Phase" means the Phase of Architect's Basic Services described in <u>Paragraph 2.1.5</u> of the <u>Description of Basic, Additional and Excluded Services Exhibit "B"</u> attached hereto.
- **1.1.48 Final Program**. "Final Program" means the Program for the Project approved by County at the conclusion of the Planning/Programming Phase, including any modifications thereto that are approved by County in the manner required by this Agreement.
- **1.1.49** Fixed Fee. "Fixed Fee" means an agreed, fixed and lump sum amount for Basic Services Fees or Additional Services Fees for all or an agreed portion of Basic Services or Additional Services, where such Basic Services Fees or Additional Services Fees compensation is based on a lump sum, fixed fee amount, rather than Maximum Hourly Fee.
- 1.1.50 Fixed Limit. "Fixed Limit" means the dollar amount set forth in Paragraph 2.3.1, below.
- 1.1.51 Formal Project Meeting. "Formal Project Meeting" means (1) a face-to-face meeting, other than a meeting with Governmental Authorities, community organizations or bidders for Construction Contracts, (2) requested by County or Architect in advance of its occurrence; (3) attended by a representative of County and Architect; and (4) at which the Architect, unless excused by County, is continuously present for at least one hour.
- **1.1.52 General Conditions.** "General Conditions" means that portion of the Contract Documents between County and Contractor or Separate Contractor, titled "General Conditions", setting forth the general terms and conditions for construction of the Work and substantially conforming to the <u>General Conditions of the Standard Form Construction Contract Between County and Contractor Exhibit "K"</u> attached hereto.
- **1.1.53** Good Faith Determination. "Good Faith Determination" means a determination made by the Director of Facilities Management, which he/she believes in good faith to be a proper exercise of County's rights and to have a reasonable basis in fact, whether or not such determination is in fact proper, reasonable

or correct or adjudged to be so.

- 1.1.54 Governmental Authority. "Governmental Authority" means the United States, the State of California, the County of Riverside (acting in its regulatory, rather than proprietary, capacity), the City in which the Project is located and any other local, regional, state or federal political subdivision, authority, agency, department, commission, board, bureau, court, judicial or quasi-judicial body, and any legislative or quasi-legislative body, or instrumentality of any of them, which has jurisdiction over the Project, Work or Site, including, without limitation, any such authority having jurisdiction to review and approve or reject the Design Documents or the Work based on compliance or non-compliance with Applicable Laws.
- 1.1.55 Hazardous Substance. "Hazardous Substance" means the following: (1) any chemical, material or other substance defined as or included within the definition of "hazardous substances," "hazardous wastes," "extremely hazardous substances," "toxic substances," "toxic material," "restricted hazardous waste," "special waste", "contamination" or words of similar import under any Environmental Law, including, without limitation, the following: petroleum (including crude oil or any fraction thereof), asbestos, asbestos-containing materials, polychlorinated biphenyls ("PCBs") and PCB-containing materials, whether or not occurring naturally; or (2) any substance that because of its quantity, concentration or physical or chemical characteristics poses a significant present or potential hazard to human health and safety or to the environment, and which has been determined by any Governmental Authority to be a hazardous waste or hazardous substance.
- **1.1.56 Hourly Rates**. "Hourly Rates" means the hourly rates for services that are compensable under this Agreement on an hourly basis and that are set forth in either: (1) the <u>Hourly Rates Schedule Exhibit</u> "H" attached hereto; or (2) a contract between Architect and a Subconsultant that has been submitted to and approved by County in the manner required by Section 1.6 of this Agreement.
- **1.1.57 Indemnitees.** "Indemnitees" means those persons and entities identified as the "Indemnitees" in Paragraph 9.1.1, below.
- **1.1.58 Initial Program.** "Initial Program" means the County's initial statement, set forth in the <u>Initial Program Exhibit "C"</u> attached hereto, of its design objectives for the Project.
- **1.1.59 Intellectual Property Rights.** "Intellectual Property Rights" means all intellectual property rights, including, without limitation, patent, trademark, trade dress, copyright, industrial design rights, priority rights, and trade secrets.
- **1.1.60** Interest Rate. "Interest Rate" means the lesser of either: (1) ten percent (10%) per annum; or (2) the maximum legal rate of interest allowed by Applicable Laws.
- **1.1.61 Invoice for Payment.** "Invoice for Payment" means an itemized invoice requesting payment that is prepared and submitted by Architect in accordance with this Agreement.
- **1.1.62 Key Personnel, Key Person.** "Key Personnel" and "Key Person" mean those individuals employed by Architect and listed in the <u>Key Personnel List Exhibit "E"</u> attached hereto, and any additions or replacements thereto approved by County, whose personal performance is deemed of the essence to this Agreement.
- **1.1.63** Loss, Losses. "Loss" and "Losses" mean any and all economic and non-economic losses, costs, liabilities, claims, damages, actions, judgments, settlements and expenses, including, without limitation, full and actual attorney's fees (including, without limitation, attorney's fees for trial and on appeal), expert and non-expert witness fees, arbitrator and arbitration fees and mediator and mediation fees.
- 1.1.64 Master Project Schedule. "Master Project Schedule" means the Master Project Schedule Exhibit

- "D" attached hereto, which sets forth the mutually agreed dates and/or time periods for achieving key milestones related to the design and construction of the Project.
- **1.1.65 Maximum Hourly Fee.** "Maximum Hourly Fee" means the agreed, not-to-exceed amount applicable to Basic Services Fees or Additional Services Fees for all or an agreed portion of Basic Services or Additional Services, where such Basic Services Fees or Additional Services Fees compensation is based on an hourly compensation not to exceed an agreed maximum, rather than a Fixed Fee.
- **1.1.66 Mold.** "Mold" means mold, mildew, spores or other microorganisms of any type, nature or description or any by-product thereof, the presence of which poses an actual or potential threat to human health, including, without limitation, any species of organisms of the kingdom of fungi, or mycota, including yeasts, smuts, ruts, mildews, molds and mushrooms or any microbial contamination, either airborne or surficial, which arises out of or is related to the presence of fungi or spores (including, without limitation, aspergilius, cladosporium, penicillium and stachybortrys chartarum).
- **1.1.67 Notice of Completion.** "Notice of Completion" means a "notice of completion" as defined in California Civil Code § 8182.
- **1.1.68 Payment Schedule.** "Payment Schedule" means the <u>Payment Schedule Exhibit "I"</u> attached hereto that allocates by percentage the total or maximum amount of compensation that County agrees to pay to Architect as Basic Services Compensation for each Phase of Basic Services.
- **1.1.69 Period of Inactivity.** "Period of Inactivity" means a period of time during which the County has directed in writing that no services are to be performed by Architect or the Subconsultants.
- **1.1.70 Phase.** "Phase" means a phase of Basic Services set forth in the <u>Description of Basic, Additional and Excluded Services Exhibit "B" attached hereto.</u>
- **1.1.71 Planning/Programming Phase.** "Planning/Programming Phase" means the Phase of Architect's Basic Services described in <u>Paragraph 2.1.2</u> of the <u>Description of Basic, Additional and Excluded Services</u> Exhibit "B" attached hereto.
- **1.1.72 Plans.** "Plans" means the graphic and pictorial portions of the Design Documents showing the design, location and dimensions of the Work, including, without limitation, plans, elevations, details, schedules and diagrams. The term "Plans" is used interchangeably with "Drawings".
- **1.1.73 Post-Completion Phase.** "Post-Completion Phase" means the Phase of Architect's Basic Services described in <u>Paragraph 2.1.8</u> of the <u>Description of Basic, Additional and Excluded Services Exhibit "B" attached hereto.</u>
- **1.1.74 Program**. "Program" means the Initial Program or Final Program and any revisions thereto authorized in writing by County.
- **1.1.75 Project.** "Project" means the work of improvement generally described on the cover page to this Agreement, with respect to which the improvements designed by Architect and Subconsultants, whether constituting a whole or a part of such work of improvement, are necessary or appurtenant to the County's use or occupancy thereof.
- **1.1.76 Project Team.** "Project Team" means County, County Consultants, Architect, Subconsultants, Contractor, Separate Contractors, Subcontractors and other firms or individuals retained by County, or retained by others with County's approval, participating in the planning, programming, design or construction of the Project.

- **1.1.77 Proprietary Information.** "Proprietary Information" means the confidential information described in <u>Section 12.7</u>, below.
- **1.1.78 Record Documents.** "Record Documents" means the collection of documents assembled and prepared by Contractor (including, without limitation, the Record Drawings and Record Specifications) showing the condition of the Work as actually built.
- **1.1.79 Record Drawings, Record Specifications.** "Record Drawings" and "Record Specifications" mean the Drawings and Specifications marked by Contractor to show the condition, location and placement of the Work as actually built, including, without limitation, the locations of mechanical, electrical, plumbing or similar portions of the Work that are depicted diagrammatically in the Drawings.
- **1.1.80 Reference Documents.** "Reference Documents" means the documents listed in <u>Reference Documents List Exhibit "J "</u>attached hereto, consisting of those reports, studies, surveys and other information provided by County for Architect's review and consideration in preparing the Design Documents.
- **1.1.81** Reimbursable Expenses. "Reimbursable Expenses" means the cost reimbursement component of Basic Services Compensation and Additional Services Compensation that involves reimbursement of out-of-pocket expenses incurred and paid in connection with the performance of Basic Services or Additional Services.
- **1.1.82 Restrictive Specifications.** "Restrictive Specifications" means a portion of the Design Documents prepared by Architect or a Subconsultant that is restricted as to brand, trade name or manufacturer in the manner defined in <u>Paragraph 2.4.5</u>, below.
- **1.1.83 Schematic Design Documents.** "Schematic Design Documents" means the Design Documents prepared by Architect and submitted by Architect for approval by County as part of the Architect's performance during the Schematic Design Phase.
- **1.1.84 Schematic Design Phase.** "Schematic Design Phase" means the Phase of Architect's Basic Services described in <u>Paragraph 2.1.3</u> of the <u>Description of Basic, Additional and Excluded Services</u> Exhibit "B" attached hereto.
- **1.1.85 Separate Contractor.** "Separate Contractor" means a person or firm, other than the Contractor, under separate contract with County to perform or supply work, materials or equipment to the Project.
- **1.1.86 Site.** "Site" means: (1) the parcel of land identified in the <u>Property Description Exhibit "A"</u> attached hereto and such additional parcels as may be purchased by County for the Project after execution of this Agreement; (2) all areas adjacent to such parcels that may be used by Contractor or a Separate Contractor or their Subcontractors for staging, storage, parking or temporary offices; and (3) all land areas, both private and public, adjacent to such parcels on which Work is required to be performed under the Contract Documents, Applicable Laws or permits relating to the Project.
- **1.1.87 Specifications.** "Specifications" means the portion of the Design Documents consisting of the written requirements for materials, equipment, standards and workmanship for the Work and performance of related services.
- **1.1.88 Subconsultant.** "Subconsultant" means a person or firm that has a contract with Architect to provide professional services to the Project.
- **1.1.89 Subcontractor.** "Subcontractor" means a person or firm that has a contract to perform a portion of Work of Contractor or a Separate Contractor, including, without limitation, subcontractors, sub-

subcontractors, suppliers, equipment operators, manufacturers and vendors, of every Tier.

- **1.1.90 Submittal.** "Submittal" means shop drawings, detailed designs, samples, exemplars, product data, fabrication plans, installation drawings, lists, graphs, operating instructions, and other similar documents required to be submitted by the Contractor, a Separate Contractor or a Subcontractor for review and approval by Architect under the terms of the Contract Documents.
- **1.1.91 Substitution.** "Substitution" means a material, product or item of equipment proposed by a Bidder in place of that specified in the Bidding Documents.
- **1.1.92 Tier.** "Tier" means the contractual level of a Subconsultant with respect to Architect, a Subcontractor with respect to the Contractor or a Separate Contractor or a County Consultant with respect to County. For example, a "first-Tier" Subcontractor is under contract with the Contractor. A subsubcontractor under contract with a first-tier Subcontractor is in the "second Tier," and so on.
- **1.1.93** Work. "Work" means all labor, materials, equipment, services, permits, licenses and taxes and all other actions and things necessary for Contractor or a Separate Contractor to fully perform its obligations under the Contract Documents (including, without limitation, any changes, additions or deletions requested by County).

1.2 INCORPORATION OF RECITALS

The Recitals set forth above are hereby incorporated as part of this Agreement.

1.3 **PERFORMANCE STANDARD**

Without limitation to the Architect's other express and implied obligations under this Agreement, all services performed in connection with this Agreement shall be performed in a manner consistent with the professional standard of care applicable to those who provide services of the type called for by this Agreement for projects of a scope and complexity that is comparable to the Project.

1.4 AUTHORITY OF ARCHITECT

Architect's authority is limited to its scope of authority set forth in this Agreement and the General Conditions. Notwithstanding anything else stated in this Agreement or any of the Contract Documents, Architect does not have the express or implied authority to obligate County to any expenditure of money or extension of contractual time periods, including, without limitation, any adjustment to the price or time of performance of any contract between County and the Contractor, a Separate Contractor, a County Consultant or any other third person or entity.

1.5 ARCHITECT'S PERSONNEL

1.5.1 Commitment, Cooperation. Recognizing the necessity of a close working relationship with County, Architect's principals and employees shall: (1) furnish their professional skill, efforts and judgment to the fullest extent in the performance of their duties and responsibilities under this Agreement; (2) provide their knowledge, ideas, experience and abilities for the efficient and cost effective design and construction of the Project; and (3) cooperate fully with all members of the Project Team.

1.5.2 Key Personnel.

.1 Of Essence. The furnishing of services by Key Persons is of the essence to this Agreement.

.2 Key Personnel List. Those persons who have been identified by County and Architect at the time of execution of this Agreement to perform services as Key Persons for the Project are listed in the Key Personnel List - Exhibit "E" attached hereto. Architect represents that it has performed a thorough background check of each of the Key Persons, and that each such check disclosed no felony conviction or other matter which casts any reasonable doubt on the competency, reliability, or honesty of such person.

.3 Additions, Removals, Replacements.

- (1) Additions. It is contemplated that from time to time, as appropriate and necessary to the stage of planning, programming, design, and construction, the need may arise for persons to be added to the Key Personnel List Exhibit "E" attached hereto to perform the functions of one or more Key Persons. Architect shall anticipate the need for such additions by submitting to County, no later than seven (7) Days prior to the need therefor, a written request for any proposed additions and the reasons therefor. County shall promptly review and respond to the Architect's request, including in such response its reasons for any disapproval. Architect shall neither allow any person who is not a Key Person approved by County to perform the functions of a Key Person previously approved by County without the advance written approval of County, which approval may be withheld if the County, acting in good faith, objects thereto.
- (2) Removal. Architect shall not, for so long as any person is employed by Architect as a Key Person, remove, replace or transfer the responsibilities of such person without County's prior written approval, which may be granted or withheld in County's sole and absolute discretion. If County is for any reason dissatisfied with the services rendered by any Key Person, Architect shall promptly recommend a substitute person as a replacement pursuant to Subparagraph 1.5.2.3, (3), below.
- (3) Replacements. In the event that Architect learns that any Key Person will be leaving the employ of Architect, Architect shall promptly notify County. In such case, or if a Key Person is requested to be removed pursuant to <u>Subparagraph 1.5.2.3</u>, (2), above, Architect shall promptly recommend for approval by County a proposed replacement person of at least equal qualifications to perform the functions of the removed Key Person, which approval may be granted or denied in County's sole and absolute discretion. Architect shall bear, at Architect's Own Expense, all Loss associated with replacing, for any reason, any Key Person, including, without limitation, all additional costs and expenses associated with familiarizing the Key Person's replacement with the particular facts, circumstances and history of the Project.
- (4) No County Liability. Neither County's request for removal, nor County's approval or disapproval, of a Key Person shall be interpreted as (a) creating any liability or responsibility on the part of County for the acts or omissions of such Key Person; (b) waiving any of County's rights under this Agreement or Applicable Laws; or (c) relieving Architect of its sole responsibility for the acts and omissions of all persons employed by Architect who perform services for the Project, including, without limitation, all Key Persons and their replacements.
- 1.5.3 Architect's Representative. The Architect's Representative is <u>John Greenwood</u>. The Architect's Representative is deemed to be a Key Person. The Architect's Representative has the authority to act on behalf of Architect in respect to all matters that are the subject of this Agreement, including, without limitation, the power and authority to contractually bind Architect to agreements and modifications of agreements. The Architect's Representative shall be qualified and authorized to make interpretations of the Design Documents and Contract Documents and shall be available at all times during all Phases to consult with County on matters pertinent to the Project.
- **1.5.4 Sole Responsibility.** All persons employed by Architect shall be the employees of Architect and not of County. Architect and its Subconsultants each shall be solely responsible for any workers'

compensation obligations, withholding taxes, unemployment insurance and any other employer obligations with respect to their respective employees.

1.6 SUBCONSULTANTS

- **1.6.1 Retention.** Architect may, with prior written approval by County granted or withheld in the sole and absolute discretion of County, retain Subconsultants to perform portions of the services required by this Agreement.
- 1.6.2 Approval. Architect's request for approval to retain a Subconsultant shall be submitted in a writing that describes the name of the proposed Subconsultant and the full contractual terms of the Subconsultant's retention, including, without limitation, the scope of services, total or maximum price and/or hourly rates, terms of reimbursement (including any markups or multipliers) and insurance. A copy of the entire proposed contract to be executed by Architect for the retention of the Subconsultant shall be provided to County if requested by County. County shall have sole and absolute discretion in approving or disapproving any Subconsultant. County shall use its best efforts to approve or disapprove of a proposed Subconsultant within seven (7) Days of Architect's request therefor and receipt by County of information requested by County pursuant to this Paragraph 1.6.2. Basic Services or Additional Services charged by Subconsultants who have not been approved by, or whose contractual terms of retention have not been submitted to, County pursuant to this Paragraph 1.6.2 will be deemed performed at Architect's Own Expense.
- **1.6.3 Pre-Approval.** County has pre-approved the Subconsultants listed in <u>Subconsultants and County Consultants List Exhibit "F"</u> attached hereto. Such pre-approval shall be deemed to be approval by County in accordance with <u>Paragraph 1.6.2</u>, above.
- 1.6.4 Written Contracts. Subconsultants shall be retained by written contract with Architect. Every contract entered into between Architect and a Subconsultant and between a Subconsultant and lower-Tier Subconsultant shall contain appropriate language whereby each Subconsultant, of every Tier, accepts and agrees, without thereby creating any contractual obligation on the part of County to the Subconsultant or any other Subconsultant, of any Tier, to be bound by all of the obligations of this Agreement, including, without limitation, those obligations pertaining to indemnification, insurance, records retention, audit, dispute resolution and ownership of Design Documents and Intellectual Property Rights. Architect further agrees to include in its contracts with its first-Tier Subconsultants the following provisions: (1) a contingent assignment of the contract to County or its designee, contingent only upon written acceptance by County or its designee; and (2) the optional right of County to directly contract with the Subconsultant for the performance of services related to the Project that are not within the scope of Architect's Basic Services under this Agreement.
- **1.6.5 Supervision.** All Basic Services that are within the field of professional practice of a Subconsultant approved by County and retained by Architect shall be directly performed or supervised by such Subconsultant. Notwithstanding the foregoing, Architect shall remain solely responsible, as between Architect, on the one hand, and County or any other person or entity to whom County may be liable, on the other hand, for the adequacy of the Subconsultant's performance and its compliance with the requirements of this Agreement.
- **1.6.6 Termination.** Architect may, upon advance written notice to County, terminate and replace the services of any County-approved Subconsultant, subject in all cases to the prior written approval of County, which approval shall not be unreasonably withheld, conditioned or delayed.
- **1.6.7 No County Responsibility.** Neither County's approval of the Architect's retaining a Subconsultant nor County's review or approval of a Subconsultant's contractual terms of retention, even if those terms conflict with this Agreement, shall give rise to any liability or responsibility on the part of County for the acts or omissions of the Subconsultant, waive any of County's rights, or relieve Architect of any of its obligations.

under this Agreement. Architect shall remain solely responsible to County, notwithstanding County's approval of any Subconsultant or its contractual terms of retention, for the quality and performance of such Subconsultant's services, and for the content, enforceability, and enforcement of all contractual terms relating to such Subconsultant's performance of services for the Project.

1.7 OWNERSHIP OF DOCUMENTS

- **1.7.1 Property of County.** Subject to the provisions of this <u>Section 1.7</u>, all Design Documents, including any designs, building designs or other depictions underlying or shown in them, shall be deemed the sole and exclusive property of County and ownership thereof is irrevocably vested in County, whether the Project is constructed or not.
- 1.7.2 Assignment of Rights. Architect shall, without further request or consideration from County, obtain and cause to be transferred to County, in writing, any and all Intellectual Property Rights in the Design Documents, including any designs, building designs or other depictions underlying or shown in them, free and clear of any liens or other encumbrances, claims or rights of third parties, and cooperate with County in securing and registering such rights so that County shall own all Intellectual Property Rights and any other tangible and/or intangible property rights in or associated with the Design Documents. Such transfer and assignment will be effective for the entire duration of the Intellectual Property Rights therein and include, but are not be limited to, all rights in related plans, specifications, documentation, derivative works and moral rights.
- 1.7.3 County Use. Without limitation to the other provisions of this Section 1.7, County shall have the right to use the Design Documents for: (1) the construction, use, occupancy or maintenance of the Project, including, without limitation, future additions, alterations, corrections or repairs to the Project; and (2) for the design, construction, use, occupancy or maintenance of any future building projects by County; provided, however, that if the County uses the Design Documents for any of the purposes set forth in Clause (2) of this Paragraph 1.7.3, then County will, without limitation to its other obligations set forth in Paragraph 1.7.8, below, indemnify and hold harmless Architect and its Subconsultants as set forth in Clause (3) of Paragraph 1.7.8, below.
- **1.7.4 No Infringement.** Architect represents and warrants that the Design Documents, whether prepared by Architect or a Subconsultant, and the use of the Design Documents in the ordinary course are free of any claim of infringement or any other violation of any Intellectual Property Right or other right of any third party.
- 1.7.5 Distribution. Except as otherwise stated in this Paragraph 1.7.5, Architect shall not copy, offer to sell, display, prepare derivative works of, distribute, make, or otherwise commercialize, any of the Design Documents, or any substantially or confusingly similar likeness thereof, for any purpose, without the prior written consent of County, which consent may be granted or denied in the sole and absolute discretion of County. Notwithstanding the foregoing, nothing herein shall be interpreted as limiting the right of Architect to copy, offer to sell, display, prepare derivative works of, distribute, make, or otherwise commercialize any standard or pre-existing details or designs, including details or designs that are part of the public domain, that are generally known or in use by other design professionals or which were developed or created by Architect or a Subconsultant prior to or independent of the services performed under this Agreement. County hereby grants to Architect and its Subconsultants a license during the term of Architect's performance of this Agreement and prior to final payment to Architect, revocable at will by County in the event of a termination of this Agreement, to use and copy the Design Documents and the designs depicted in or underlying them for the purpose of performing the services required under this Agreement.
- **1.7.6 Inspection.** County shall have the right at any time or times, upon prior written request by County, to review the status and condition of the Design Documents while in development or during construction and to request that copies thereof be provided to County.

- 1.7.7 **Delivery.** Architect shall, at any time upon request by County and without request by County upon or after termination or full performance of this Agreement, promptly deliver to County the originals and copies (including paper and electronic versions) of all Design Documents, whether prepared by Architect or the Subconsultants. Electronic versions shall be submitted using AutoCAD, Adobe Acrobat or other software satisfactory to County and shall be in a form that is indexed and editable. Architect shall be permitted to retain copies, including reproducible copies, of the Design Documents for its files, information and reference.
- 1.7.8 Indemnification. County shall indemnify and hold harmless Architect and its Subconsultants against Loss resulting from: (1) the use of the Design Documents by County in a manner not permitted by Paragraph 1.7.3, above; (2) a modification by or at the request of County of the Design Documents, without the prior written consent of the Architect (in the case of Design Documents prepared by Architect) or the Subconsultant responsible for preparation of the Design Documents (in the case of Design Documents prepared by a Subconsultant); or (3) the use of the Design Documents by County in the manner permitted by Clause (3) of Paragraph 1.7.3, above.
- **1.7.9 Disputes.** Payment shall not be interpreted as a condition to, nor shall disputes between County and Architect diminish or in any way limit, the rights of County under this <u>Section 1.7</u>.

1.8 APPLICABLE LAWS

- **1.8.1 Compliance by Architect.** Subject to the other provisions and limitations of this Section 1.8 and without intending to limit Architect's right to Additional Services Compensation pursuant to Article 3, below, for changes in Applicable Laws, Architect shall, at all times in its performance under this Agreement, comply with Applicable Laws. Subject to the provisions of Paragraph 1.8.4, below, Architect shall not be relieved from its obligation under this Paragraph 1.8.1 by the errors or oversights by Governmental Authorities in their issuance of general approvals or permitting of Design Documents for the Project.
- **1.8.2 Changes in Laws.** Architect is obligated, exercising the standard of professional care set forth in Section 1.3, above, to keep informed and advise County of changes in Applicable Laws that affect the Project and promptly inform County of such changes in advance of their becoming effective.
- 1.8.3 Disability Laws. The Architect's duty with respect to conforming its Design Documents to Disability Laws shall be limited to the application and incorporation of the technical requirements of either or both (as applicable by the terms of the Disability Laws) the Americans With Disabilities Act (ADA) and the Fair Housing Administrations Act (FHAA), as well as the technical requirements of other Disability Laws. The County acknowledges and agrees that such technical requirements may be subject to different interpretations, legal or otherwise, and that the Architect cannot, and does not, guaranty or warrant compliance with all interpretations of Disability Law requirements; provided, however, that the Architect remains at all times obligated to interpret applicable technical requirements of Disability Laws exercising the standard of professional care set forth in Section 1.3, above.
- **1.8.4** Conflicts in Interpretations. Nothing stated in this Agreement shall be interpreted as creating or implying any responsibility or liability on the part of Architect for a Conflicting Interpretation of an Applicable Law.
- **1.8.5 Legal Notices.** Architect shall provide such legal notices or certifications as may be required by Applicable Laws to be provided by architects providing design services for comparable projects, including, without limitation, notice that Work is being, or is about to be, carried out at the Site, information concerning the identity of the Contractor and Separate Contractors and other required information concerning the condition or performance of the Work. Copies thereof shall be submitted to County for review and approval prior to their being distributed, posted or published.

1.8.6 Direct Communications. Architect shall not communicate directly with any Governmental Authority without County's prior approval, which shall not be unreasonably delayed, conditioned or withheld.

1.9 TIME OF ESSENCE

All time limits set forth in this Agreement pertaining to Architect's performance of any obligation or act for the benefit of County or the Project are deemed to be of the essence to this Agreement.

ARTICLE 2 BASIC SERVICES

2.1 BASIC SERVICES, ADDITIONAL SERVICES, EXCLUDED SERVICES

- **2.1.1 Basic Services.** Architect's Basic Services consist of the professional design and engineering services, whether provided by Architect or its Subconsultants, described in this <u>Article 2</u>, including, without limitation, those specific Basic Services listed and checked "Yes" in the <u>Description of Basic, Additional and Excluded Services Exhibit "B"</u> attached hereto, as well as any other services, within the scope of Architect's or its Subconsultants' fields of professional practice, that are reasonably inferable as being necessary, or that are customarily furnished by other providers of professional services of the type and nature provided for in this Agreement, to accomplish the Basic Services expressly required by this Agreement. Basic Services do not include Additional Services or Excluded Services.
- **2.1.2** Additional Services. Additional Services are those services so defined as Additional Services in Article 3, below. Additional Services shall be performed only if authorized in writing by County in the manner required by Article 3, below.
- **2.1.3 Excluded Services.** Excluded Services are services so designated as "Excluded Services" in the Description of Basic, Additional and Excluded Services Exhibit "B" attached hereto. Excluded Services shall be not be performed by Architect unless mutually agreed upon between County and Architect.
- **2.1.4 Phases.** References in this Agreement or its exhibits to performance of Basic Services during a particular Phase is for convenience only and not as a limitation, with the understanding that Basic Services described in this Agreement as being part of a particular Phase shall be performed by Architect as and whenever reasonable and necessary for the proper and complete design of the Project.

2.2 PROGRAM

- **2.2.1 Initial Program.** Architect shall, as part of its Basic Services performed during the Planning/Programming Phase, review and revise the Initial Program as necessary to meet the County's expressed criteria and requirements for the Project.
- **2.2.2 Final Program.** A Final Program shall be approved by County in writing upon completion of the Planning/Programming Phase. Architect shall thereafter proceed to complete the design of the Project in accordance with the Final Program.
- **2.2.3 Changes.** Subject to Architect's rights to Additional Services Compensation for Additional Services authorized in accordance with <u>Article 3</u>, below, County reserves the right, exercised in its sole and absolute discretion, to modify the Initial Program and Final Program at any time. No other changes to the Initial Program or Final Program are permitted.
- 2.2.4 Notice. If Architect learns of any circumstances (including, without limitation, any directive or

instruction by County or any other Project Team member) which involve, or if implemented may result in, a deviation from the Final Program, it shall promptly, prior to performance of any services or further services related thereto, notify the County in writing and request approval of such deviation. Such written notice shall include a statement of the impact that such deviation, if approved and implemented, would have upon the Architect's Project Schedule and the Fixed Limit and an estimate of the cost to County of any related Additional Services Compensation permitted under Article 3, below. Upon written approval by County, Architect shall proceed to revise the Design Documents to conform to such approved deviations. Deviations from the Final Program are the sole responsibility of the Architect and any services required by Architect or its Subconsultants to modify the Design Documents as a result thereof shall be performed at the Architect's Own Expense, unless the Architect has specifically informed the County in writing of such deviation and the County has given specific written approval thereof in accordance with this Paragraph 2.2.4.

2.2.5 Integrated Design. The requirements of the Final Program shall be deemed to include the requirement that Architect and its Subconsultants adopt an integrated approach to the design of the Project that makes optimal use of opportunities for the synergies of different design elements into an integrated whole that is energy efficient and economical to construct, maintain and operate.

2.3 FIXED LIMIT

- **2.3.1 Fixed Limit Amount.** The Fixed Limit for the Project is <u>Seven Hundred Fifty-Nine Thousand, Six Hundred Sixty Eight Dollars</u> (\$759,668). The Fixed Limit shall not be reduced except upon mutual agreement of County and Architect. The County shall have the right, exercised in its sole and absolute discretion, to increase the Fixed Limit at any time or for any reason.
- **2.3.2 Architect's Obligation.** It is the obligation of the Architect and its Subconsultants to design the Project for a total sum of Construction Costs that does not exceed the Fixed Limit by more than ten percent (10%) of the Fixed Limit as adjusted in accordance with <u>Paragraph 2.3.4</u>, below.
- **2.3.3 Review of Estimates.** Architect shall carefully review any Estimates of Construction Costs provided to it by County or a County Consultant and exercising the standard of care set forth in <u>Section 1.3</u>, above, advise County if any portions thereof that are excessive, inadequate or otherwise irregular. Architect shall provide such information as County or County Consultants may reasonably require in order to assist them in preparing or revising such Estimates of Construction Costs.
- **2.3.4 Fixed Limit Adjustments.** The Architect is not responsible for, and the Fixed Limit shall be increased on account of, any of the following:
- .1 an adjustment to the Fixed Limit provided for by <u>Paragraph 2.3.8</u>, below, due to County's rejection of a non-programmatic design revision recommended by Architect;
- .2 a change to the Final Program approved by County that is requested in writing by County and that materially increases the Construction Costs for the Project;
- .3 a change to the Design Documents that is ordered or approved in writing by County after the start of construction and that is not the result of information in the Design Documents that constitutes an error, omission, deficiency, conflict or violation of Applicable Laws or that is the result of the negligence, willful misconduct or violation of an Applicable Law by Architect or a Subconsultant or a breach of this Agreement by Architect; or
- .4 demonstrated and quantifiable escalations in costs of construction materials and/or labor as reported by the Engineering News Record or other recognized industry publication on construction prices, for construction costs in the classification area that includes the County of Riverside, or if more than one classification area includes the County of Riverside that classification area that is most

precisely reflective of the prices in the County of Riverside; subject to the further conditions that an adjustment to the Fixed Limit pursuant to this <u>Subparagraph 2.3.4.4</u> for escalation shall only be permitted: (1) if bids for construction have not been opened by County within three (3) months after approval by County of the Final Construction Documents and (2) such escalation adjustment shall be limited to the escalation that occurs during the period of time starting with the expiration of such 3-month period and ending when bids for the Project are actually opened by the County.

- **2.3.5 Notice of Overruns.** In addition to and without limitation upon the Architect's obligations under Paragraph 2.3.6, below, Architect shall, exercising the standard of care set forth in Section 1.3, above, advise County within five (5) Days of learning of any circumstances indicating that Construction Costs for the Project are reasonably likely to exceed the Fixed Limit.
- 2.3.6 Redesign Services. Without limitation to Architect's obligation under Paragraph 2.3.2, above, if at any time the Fixed Limit as adjusted in accordance with Paragraph 2.3.4, above, is exceeded by more than ten percent (10%) of such adjusted Fixed Limit by either: (1) the sum of the Construction Costs to construct the Project that are set forth in an Estimate of Construction Costs prepared by Architect, a Subconsultant or a County Consultant; or (2) the "lowest responsive bid" obtained by the County from a responsible contractor pursuant to a competitive bidding process, then Architect shall, if so directed by County in writing, propose and, if further directed by County in writing, implement revisions to the Design Documents (including, without limitation, multiple redesign alternatives) for the purpose of reducing the sum of Construction Costs to an amount that is within the sum of the Fixed Limit as adjusted in accordance with Paragraph 2.3.4, above, plus ten percent (10%) of such adjusted Fixed Limit. The Architect shall clearly identify, in writing, at the time its redesign recommendations are submitted whether its recommendations are consistent or inconsistent with the Final Program. Nothing stated in this Paragraph 2.3.6 or elsewhere in this Agreement shall be interpreted as obligating County to accept, or to pay additional compensation to Architect for, any redesign recommendation, or any redesign services to implement a redesign recommendation, that is inconsistent with the Final Program.
- **2.3.7 Architect's Own Expense.** Services by Architect or a Subconsultant to redesign pursuant to Paragraph 2.3.6, above, shall be performed at Architect's Own Expense.
- **2.3.8 Non-Programmatic Revisions.** The Fixed Limit shall be adjusted in accordance with <u>Paragraph 2.3.4</u>, above, if and to the extent that the total of the Construction Costs for the Project exceeds such Fixed Limit by more than ten percent (10%) of such Fixed Limit due to the County's refusal or failure to accept a redesign recommendation that is consistent with the Final Program and that is submitted by Architect in accordance with the requirements of <u>Paragraph 2.3.6</u>, above. The amount of such adjustment to the Fixed Limit shall be based on the reasonable, estimated reduction in Construction Costs that would have been achieved if the County had accepted the Architect's recommendation.
- **2.3.9** Basic Services. All services required to be performed by Architect or its Subconsultants in compliance with the Architect's obligations under this <u>Section 2.3</u> shall be furnished as part of Basic Services.
- **2.3.10 Continuous Performance.** In the event of a dispute over the amount of total Construction Costs estimated or incurred to construct the Project as designed by Architect and its Subconsultants, Architect shall, without interruption or delay to its Basic Services and without waiving Architect's right to an appropriate adjustment to the Fixed Limit, perform as directed by County.
- **2.3.11 No Limitation.** Nothing stated herein shall be interpreted as a limitation on any of County's rights or remedies that may exist under Applicable Laws, including, without limitation, the right to terminate this Agreement for default (with or without extending to Architect the opportunity to redesign) in the event of a failure by Architect to perform any of its obligations under this <u>Section 2.3</u>.

2.4 DESIGN DOCUMENT SUBMISSIONS

- **2.4.1 Submission to County.** Architect shall make such formal submittals of Design Documents to County as and when required by the terms of the <u>Description of Basic</u>, <u>Additional and Excluded Services Exhibit "B"</u> attached hereto. All submissions by Architect of Drawings or Specifications for review by County shall be accompanied by a transmittal listing the issuance and revision dates of each document submitted, including, without limitation, a complete list by sheet numbers of all Drawings comprising the submission and a comprehensive narrative explaining all significant changes and additions made since the last formal submission of Drawings to County. All such submissions shall be in multiple sets and such format as required in the <u>Description of Basic</u>, <u>Additional and Excluded Services Exhibit "B"</u> attached hereto. Electronic versions shall be submitted using AutoCAD, Adobe Acrobat or other software satisfactory to County.
- **2.4.2 County Approval.** County shall promptly approve or disapprove, with explanation of the reasons for disapproval, of all formal submittals of Design Documents that are required by the terms of the Description of Basic, Additional and Excluded Services Exhibit "B" attached hereto. County's review is not for the purpose of discovering errors, omissions, conflicts, lack or coordination, ambiguities or violations of Applicable Laws in the Design Documents. Architect shall, notwithstanding any review or approval of Design Documents by the County, remain solely responsible for the content, completeness, accuracy and sufficiency of all Design Documents prepared by Architect or its Subconsultants and, except for deviations approved by County in accordance with Paragraph 2.2.4, above, for their compliance with the Final Program and Fixed Limit.
- **2.4.3 Governmental Authorities.** Unless otherwise directed in writing by County, submission of Design Documents to Governmental Authorities for their review or approval shall be sufficiently complete at the time of submission so as to require either no, or only minimal and minor, corrections and supplementation.
- **2.4.4 Corrections.** Services and related costs required to make corrections that are the result of information in the Design Documents constituting errors, omissions, conflicts, lack of coordination or violations of Applicable Laws shall under no circumstances constitute grounds for Additional Services Compensation. Moreover, if the Basic Services Compensation is based upon hourly rates and if such corrections are, in the reasonable opinion of County, excessive then any services and associated costs and expenses to make those corrections that are so judged by County to be excessive shall be performed and incurred at the Architect's Own Expense. All corrections shall be incorporated into the next Design Documents submission and highlighted by notation or clouding or if not incorporated then such submission shall be accompanied by a written statement to County, not in the form of notes or clouding on the Design Documents, explaining why such corrections were not incorporated.
- **2.4.5** Restrictive Specifications. Architect shall not, without the prior written approval of County, include or permit there to be included a Restrictive Specification in the Design Documents. For purposes of this Paragraph 2.4.5, a Restrictive Specification is any portion of the Design Documents that: (1) expressly or impliedly calls for the supply, furnishing or installation of any material, product, equipment, or other thing by a single producer of a particular brand, trade name, or manufacturer to the exclusion of other producers; (2) does not expressly permit the submission for consideration and possible approval by County in accordance with the requirements of California Public Contract Code §3400 of other "or equal" materials, products, equipment or things produced under or by another brand, trade name or manufacturer; or (3) fails to list a brand, trade name or manufacturer that is known to Architect or its Subconsultants to be a brand, trade name or manufacturer that produces a material, product, equipment, or other thing in California that is equal to the material, product, equipment or other thing specified in the Design Documents.
- **2.4.6** Legal Review. Architect understands and acknowledges that County has developed standard forms of legal terms and conditions governing bidding and performance of the Work. Unless approved in writing by County, Architect shall not include provisions in its Final Construction Documents that modify or

conflict with the terms of the County's standard forms. In order to assist Architect is this regard, Architect shall, if it wishes County to review its Design Documents for such possible modifications or conflicts, submit drafts to the County requesting the County's legal review. Such submission shall afford the County at least thirty (30) Days to perform such review without causing Delay to the Project. All communications between Architect and County's legal advisors with respect to the matters addressed by this Paragraph 2.4.6 constitute confidential attorney-client communications, protected attorney work product and contain other confidential information that is not be disclosed to third parties. Architect shall maintain such confidentiality, privileges and protections against any intentional or inadvertent disclosure or other action within the control of Architect or its Subconsultants that might cause such privileges or protections to be lost or diminished.

2.5 **CONSTRUCTION DOCUMENTS**

Without limitation to the other provisions this Agreement requiring the production, preparation or delivery of documents or things by Architect, the Construction Documents delivered to County shall include those Design Documents described in Construction Document Deliverables – Exhibit "O " attached hereto in a condition that is appropriate to each of the descriptions of the Phases of Basic Services set forth in the Description of Basic, Additional and Excluded Services - Exhibit "B" attached hereto.

2.6 **DESIGN COORDINATION STANDARDS**

All Drawings and Specifications submitted by Architect to County for its review shall conform to the Design Coordination Standards. Notwithstanding the division or delegation by Architect of responsibility for preparation of some portions of the Design Documents to separate Subconsultants, Architect shall remain responsible for the complete coordination, in every respect and without limitation, of the information contained in the Design Documents, whether prepared by Architect or its Subconsultants.

2.7 COMMUNICATIONS

Architect shall comply with all written procedures issued by County for the conduct of communications relating to the Project or among the Project Team members. All communications with County shall be directed or copied to the attention of the Director of Facilities Management. County will endeavor to furnish Architect with copies of written communications from County to its Contractor, Separate Contractors and County Consultants that pertain to the Architect's services under this Agreement.

2.8 CONSTRUCTION MEANS, METHODS AND SAFETY

Unless the Architect has breached this Agreement or Architect or a Subconsultant has violated an Applicable Law, acted with willful misconduct or violated the standard of care set forth in Section 1.3, and then only to the extent of such breach, willful act or violation, Architect shall have no responsibility or liability with regard to, and Contractor and the Separate Contractors shall be solely responsible for, all selections (other than those selections expressly dictated by Architect or the Subconsultants in the Design Documents) of, and all supervision, implementation and enforcement relating to, construction means, methods, sequence, techniques, procedures or related matters involving health and safety of persons or protection of property at the Site during construction.

2.9 REJECTION OF WORK

- **2.9.1** Recommendations by Architect. Architect shall recommend in writing to County the rejection of Work that Architect determines or believes does not conform to the Contract Documents, including, without limitation, Work that has not been inspected or tested in accordance with the requirements of the Contract Documents.
- 2.9.2 County Approval. The decision whether to reject any portion of the Work recommended for

rejection by Architect shall be made only after consultation with and upon written approval by County.

- **2.9.3 No Duty of County.** Neither County's authority under this <u>Section 2.9</u> nor a decision made in good faith by County in accordance with the provisions of this <u>Section 2.9</u> either to reject or not reject Work shall give rise to a duty or liability of County to Architect, Contractor, or any other person or entity to reject Work, whether or not such rejection is recommended by Architect.
- **2.9.4 Contractor Responsibility.** No determination by Architect, that is exercised in good faith and in accordance with the standard of care set forth in <u>Section 1.3</u>, above, to recommend or not recommend rejection of Work shall give rise to any liability on the part of Architect for Defective Work provided by Contractor or a Separate Contractor. No recommendation by Architect to reject or not reject Work shall be interpreted as relieving Contractor or any Separate Contractor of its responsibility for failing to comply with the Contract Documents.
- **2.9.5 No Authority to Stop Work.** Architect shall immediately recommend to County the stopping of the Work if circumstances come to Architect's attention that reasonably require the stopping of the Work in order prevent Loss to County or injury or damage to persons or property. Nothing stated herein or elsewhere in this Agreement or its exhibits shall be interpreted as giving Architect or the Subconsultants the right or authority, under any circumstances, to direct any Contractor or Separate Contractor to stop performance of the Work.

2.10 PERMITS, APPROVALS

- **2.10.1 Submission to Governmental Authorities.** Architect shall promptly submit all necessary applications and other documents required by Applicable Laws for obtaining building permit(s) and approvals of Governmental Authorities for the construction of the Project that are not customarily submitted and secured by the Contractor or Separate Contractors.
- **2.10.2 Off-Site Improvements.** Architect recognizes that it is of critical importance to County that permits for off-Site improvements be obtained prior to the Bidding Phase and Architect will plan and schedule the performance of its services accordingly.
- **2.10.3 Notice to County.** Architect shall notify the County in writing upon receipt of those permits and approvals that are within the scope of Architect's responsibility under this <u>Section 2.10</u>, specifying in detail the scope of Work covered thereby. If Architect learns that any required permit or approval that it believes County is obligated to obtain directly has not been obtained, Architect shall notify County immediately in writing specifying the permit or approval required and the time frame within in which it must be obtained in order to not cause Delay to the Project.
- **2.10.4 Expiration.** Architect shall notify County in writing at least ninety (90) Days prior to the date that any permit or approval obtained by Architect will expire and shall notify the County immediately in writing if it learns that any other permit or approval will expire in fewer than ninety (90) Days.

2.11 TESTING AND INSPECTIONS

- **2.11.1 Recommendations by Architect.** Architect shall recommend in writing and include in the provisions of its Specifications, any special inspection or testing of the Work if, in Architect's or any Subconsultant's professional judgment, such inspection or testing is required by Applicable Laws or is necessary or advisable.
- **2.11.2** Review of Reports. Architect and its Subconsultants shall review all inspection reports, laboratory reports, and test data generated from the conduct of special inspections or testing recommended by Architect pursuant to Paragraph 2.11.1, above, in order to determine whether such data conforms with the

requirements of the Contract Documents and Submittals approved by Architect.

- **2.11.3 Additional Inspection, Testing.** Architect shall recommend additional inspection or testing not required by the Contract Documents that Architect, in the exercise of the standard of care set forth in <u>Section 1.3</u>, above, determines is necessary, at any point in time that Work is being fabricated, installed or completed, in order to confirm that the Work conforms to the Contract Documents.
- **2.11.4** No Duty of County. Neither the authority of County, County Consultants or Inspectors of Record to decide whether inspection or testing is needed, nor a decision made in good faith by any of them to order or not order inspection or testing, shall give rise to a duty or liability of County to Architect, Contractor, or any other person or entity.
- **2.11.5 Defective Work.** No determination that is made by Architect in accordance with the standard of care set forth in <u>Section 1.3</u>, above, to recommend or not recommend testing or inspection shall give rise to any liability on the part of architect for Defective Work performed by a Contractor or Separate Contractor or be interpreted as relieving any Contractor or Separate Contractor of its responsibility for Defective Work.

2.12 COUNTY CONSULTANTS

- **2.12.1 Retention.** Architect shall advise County on the appropriate time for retention of County Consultants, not identified in <u>Subconsultants and County Consultants List Exhibit "F"</u>, whose services are necessary to the implementation of the Project.
- **2.12.2 Prequalification.** Architect shall provide advice on retention of County Consultants sufficiently in advance to allow time for prequalification and selection of County Consultants in accordance with Applicable Laws and the guidelines, practices and procedures of County. At points in time appropriate to the stage and status of a Project, Architect shall assist County with prequalification and selection of County Consultants, including: (1) preparation of prequalification criteria; (2) preparation of requests for qualifications; (3) conduct of prequalification conferences and responses to questions by proposers; (4) evaluation of proposers; (5) establishment of a list of prequalified professionals; and (6) preparation of a definitive scope of services.
- **2.12.3 Coordination and Review.** Architect shall efficiently and expeditiously coordinate its performance under this Agreement with the services provided by the County Consultants including, without limitation, the following:
- .1 making provision in the Design Documents prepared by Architect and its Subconsultants to adapt them to the requirements, constraints and limitations of the work product provided by County Consultants to Architect;
- .2 furnishing County Consultants with information, as soon as it is available to Architect or its Subconsultants, that the Architect should reasonably expect, in the exercise of the standard of care set forth in Section 1.3, above, the County Consultants may require in order to adapt their work product to the requirements, constraints and limitations of the Site, Existing Improvements and the Design Documents prepared by Architect or its Subconsultants;
- .3 evaluating and interpreting the work product of County Consultants to determine if it is consistent with (1) the express, written requirements of Governmental Authorities, including, without limitation, permits and approvals that have been issued or will be required for the Project, and (2) the observable or known conditions at the Site and in Existing Improvements;
- .4 evaluating the work product of the County Consultants to determine it is consistent with the Architect's design intent;

- .5 assuming primary and lead responsibility to initiate and maintain a regular and continuous exchange of information among the Architect, Subconsultants, and County Consultants so that the coordination required by this <u>Paragraph 2.12.3</u> is accomplished in a manner that avoids inconsistencies, conflicts, omissions, unnecessary revisions and duplication of effort; and
- .6 promptly advising County in writing if Architect or a Subconsultant learns of any defects or deficiencies in the work product prepared by a County Consultant; provided, however, that nothing stated in this Paragraph 2.12.3 shall be interpreted as relieving the County Consultants of liability or responsibility for their work product nor as creating or implying an assumption of any duty or responsibility on the part of the Architect or any Subconsultant for matters that are outside the scope of their respective fields of professional practice.

2.13 MEETINGS

- **2.13.1 Basic Services.** Architect shall, throughout all Phases of performance of its Basic Services, attend all meetings with County and other Project Team members that are either requested by County or that are reasonably necessary to maintain progress of the Project and shall promptly address and resolve matters in question. Except as otherwise provided in Paragraph 2.13.6, below, attendance at all such meetings, regardless of the number, frequency, timing or duration of meetings requested or required, are part of Basic Services.
- **2.13.2 Participants.** Architect shall arrange for and ensure attendance at meetings by those persons employed by Architect and its Subconsultants who are knowledgeable in the matters to be discussed. If any such person is not available, another person reasonably familiar with such matter shall attend in his/her place.
- **2.13.3 Conduct of Meetings.** Architect shall come to all meetings prepared to discuss the status of the Project (including both the status of design and construction) and to address any matters in question brought to Architect's attention by other Project Team members. Unless excused by County, Architect's representative attending the meeting shall remain present for the duration of the meeting.
- **2.13.4 Action Items.** Architect shall respond promptly with respect to matters assigned to Architect or its Subconsultants at meetings for action or resolution.
- **2.13.5 Meeting Minutes.** If requested by County, Architect shall prepare and distribute minutes of meetings. If Architect prepares the minutes, Architect shall promptly review and implement requests for corrections to meeting minutes within five (5) Days after receipt thereof and re-distribute copies of corrected minutes.
- **2.13.6 Numerical Limitations.** If under the terms of <u>Description of Basic, Additional and Excluded Services Exhibit "B"</u> attached hereto or elsewhere in this Agreement or its exhibits, County and Architect have agreed to limit the number of meetings required to be attended by Architect or its Subconsultants as part of Basic Services, then the following provisions shall apply:
- .1 Formal Project Meetings. In order for a meeting to be considered included in and subject to an agreed numerical limitation applicable to meetings attended as part of Basic Services, the meeting must be a Formal Project Meeting. With respect to each Formal Project Meeting, Architect's Basic Services shall include, for all persons attending the meeting on behalf of Architect and its Subconsultants, all time expended by Architect and its Subconsultants: (1) in preparation for, travel (including subsistence) to and from, and attendance at the Project Meeting; (2) in the preparation, review and correction of Project Meeting minutes; and (3) in performing services to follow-up on matters identified in the Project Meeting for further action by a Project Team member, unless and except such follow-up action is based on circumstances constituting independent and authorized grounds for performance of Additional

Services and payment of Additional Services Compensation

- .2 Other Basic Services Meetings. Except for meetings that are required as part of an Additional Service, participation in meetings, conferences and conversations, whether face-to-face or otherwise that do not constitute a Formal Project Meeting, regardless of their frequency, number, duration or timing, shall all be considered part of Basic Services and are not subject to any agreed limitation on frequency, number, duration or timing.
- Additional Services. Formal Project Meetings that exceed the total number of Project Meetings agreed to be included as Basic Services shall, subject to the other terms and conditions applicable to Additional Services set forth in Article 3, below, constitute an Additional Service provided that Architect notifies County in writing, as far in advance of the Formal Project Meeting as reasonably practicable, that participation in the Formal Project Meeting is believed by Architect to constitute an Additional Service. Such notice, if timely and properly given, shall be deemed to satisfy the requirement for written notice required under Section 3.4, below, and no separate or additional written notice shall be required pursuant to Section 3.4, below. Failure to give such notice shall result in such meeting being deemed to not be a Formal Project Meeting, in which case such meeting will be attended by Architect and its pertinent Subconsultants as part of Basic Services pursuant to Subparagraph 2.13.6.2, above.
- Compensation for time spent by Architect or its Subconsultants in connection with Formal Project Meetings on the basis that the number of meetings occurring in a Phase exceeds the number of meetings allocated to that Phase unless the total aggregate number of Formal Project Meetings (whether attended by Architect or a Subconsultant) occurring during all Phases exceeds the total aggregate number of meetings agreed to for all Phases (including meetings to be attended by Architect or a Subconsultant). Further, if the number of such Formal Project Meetings occurring during a Phase is fewer than the agreed number of meetings (including meetings to be attended by Architect or a Subconsultant) allocated to that Phase, the shortfall shall be "banked" and available for use by County on any other Phase in which the number of such Formal Project Meetings exceeds the agreed number of meetings (including meetings to be attended by Architect or a Subconsultant) allocated to that Phase. Architect shall not be required to reduce its Basic Services Compensation in the event that the total aggregate number of Formal Project Meetings (whether attended by Architect or a Subconsultant) occurring in all Phases is fewer than the agreed aggregate total of meetings for all Phases.

2.14 SITE AND EXISTING IMPROVEMENTS

Architect shall (1) carefully review all reports, studies, surveys, data and other documents concerning the conditions, both visible and concealed, at the Site and in Existing Improvements that are: (a) provided to Architect by the County (including, but not limited to, the Reference Documents listed in the Reference Documents List - Exhibit "J" attached hereto); (b) recorded with the County's Office of the County Recorder; (c) on file with, and available for review (without necessity of a formal public records request) from, the County's departments responsible for planning, zoning and construction within the County; and (2) exercising the standard of care set forth in Section 1.3, above, identify and review any other reports, studies, surveys, data and other documents that are reasonably available from other sources concerning the conditions, both visible and concealed, at the Site and in Existing Improvements. Architect shall coordinate its Design Documents with the information disclosed in said reports, studies, surveys, data and other documents.

2.15 ELECTRONIC PROGRAM MANAGEMENT

If County has established an electronic program management system for the furnishing, storage, exchange and transmission of electronic documentation and communications relating to the Project, then Architect shall, as part of its Basic Services comply with the requirements of such system, including, without limitation,

the following: (1) using e-mail for communications, wherever possible; (2) using electronic versions of Design Documents for distribution for bidding or other purposes; (3) scanning of documents; and (4) cooperating in maintaining a common file and electronic archive. All costs of setting up the electronic program management system will be paid directly by County. Costs incurred by Architect to use the system (including, without limitation, the purchase of any software) shall be at Architect's Own Expense. County shall have the right, exercised in its sole and absolute discretion, to cease operation of the electronic program management system at any time upon written notice to Architect if County determines that continued operation is not in the best interests of the County.

2.16 TIME AND SCHEDULE

- **2.16.1 Master Project Schedule.** Architect shall meet the requirements of the <u>Master Project Schedule Exhibit "D"</u> attached hereto, including, without limitation, the dates set forth therein for completion of each Phase of Architect's Basic Services, subject to such extensions as are permitted by <u>Paragraph 2.16.3</u>, below.
- **2.16.2** Architect's Project Schedule. Architect shall, within ten (10) Days after execution of this Agreement by County and Architect, prepare and submit for County's approval an Architect's Project Schedule for the Project that: (1) conforms to and incorporates the deadlines and time periods applicable to each Phase as set forth in the Master Project Schedule; (2) coordinates and integrates the planning, programming, design and other activities of Project Team members in appropriate detail to County's satisfaction; (3) shows estimated commencement, duration, responsible parties and sequence for planning, programming and design; (4) shows the interdependencies of the planning and design activities; and (5) shows deadlines and estimates of time for review and receipt of all approvals, decisions and other information to be provided by County and Governmental Authorities, allowing a reasonable time for such review, approvals and decisions.
- 2.16.3 Time Extensions. Architect shall be entitled to an adjustment of the Master Project Schedule and Architect's Project Schedule for, and shall not be considered in default of this Agreement because of, unavoidable and unforeseeable Delays that are beyond Architect's and its Subconsultants' reasonable control and beyond Architect's responsibility under this Agreement (such as, but not limited to, strikes, lockouts, work slowdowns or stoppages, accidents, acts of God, failure of any Governmental Authority to act in a reasonably timely manner, failure by County to timely provide information or approvals or Delays caused by the fault of the Contractor or Separate Contractors) for which Architect has given written notice to County of the circumstances of such Delay within three (3) Days after first learning of such circumstances. Failure to provide such written notice shall result in a waiver by Architect of any right to an adjustment to the Architect's Project Schedule on account of such circumstances. Without limitation to the foregoing, neither the bankruptcy, insolvency nor financial inability of Architect or any Subconsultant, nor any failure by a Sububconsultant to perform any obligation imposed by contract or Applicable Laws, shall constitute a grounds for extension under the terms of this Paragraph 2.16.3.
- **2.16.4 Schedule Updates.** Architect shall, no less frequently than monthly, update and expand the level of detail in the Architect's Project Schedule as the Project progresses, indicating current status of scheduled activities and projections of potential completion of major tasks. If significant variance from planned activities occurs, Architect shall recommend recovery plans to County and, upon obtaining County's approval thereof, modify the Architect's Project Schedule to incorporate such recovery plans.
- **2.16.5** County Approvals. Failure by County, a County Consultant or any other Project Team member to provide an approval, decision or other information needed by Architect shall not constitute grounds for extension to the Architect's Project Schedule unless such approval, decision or other information is not provided, without reasonable justification, within seven (7) Days after written request therefor by Architect delivered to the Project Team member from whom such approval, decision or other information is required, with a copy to County, stating that failure to provide such approval, decision or other information within

- seven (7) Days may or will result in Architect seeking an extension of the Architect's Project Schedule on the grounds of Delay.
- **2.16.6 County Review.** Architect is solely responsible, notwithstanding County's review or approval thereof, for the completeness, accuracy and suitability of the Architect's Project Schedule and all updates thereof.
- **2.16.7 No Delay by Architect.** Architect shall not delay its interpretations, decisions, reviews or other functions pursuant to this Agreement or otherwise cause or contribute to a Delay to the progress of design or construction of the Project.
- **2.16.8 Delay Costs.** Architect's sole and exclusive right and remedy for recovery or compensation for Losses related to Delay, of any kind, are: (1) its right to Additional Services to the extent permitted by Paragraph 3.2.6, below; and (2) its right to adjustment of the Hourly Rates, if any, provided for by this Agreement. All other rights and claims by the Architect, on its own behalf and on behalf of its Subconsultants, for Losses relating to Delay, from any cause whatsoever, are hereby waived.

2.17 SUSTAINABLE DESIGN REQUIREMENTS

- **2.17.1 LEED Accreditation.** Architect and each of its Subconsultants in the fields of mechanical design and electrical design, each shall assign at least one (1) individual who has received an accreditation as a "LEED-Accredited Professional" by the U.S. Green Building Council ("USGBC") and demonstrated experience in designing projects to the standards of USGBC to participate in the preparation of the Construction Documents. Such individual shall be considered a Key Person under this Agreement.
- **2.17.2** Approvals and Design Credits. Architect shall include all of the necessary designs and specifications in its Construction Documents, and provide all of the other supporting documentation required by USGBC, to enable County to obtain all full approval and certification by the USGBC that the relevant portions of the Construction Documents meet the requirements of the LEED NC Reference Guide (First Edition, 2005) and so that County can receive the USGBC "design credits" listed in LEED's Design Credits Exhibit "Q", attached hereto. Submittals by Architect and Subconsultants for USGBC approval shall, as a part of Basic Services, be corrected or revised, as deemed necessary by the USGBC for issuance of such approvals and design credits. In the event that, due to an error or omission by Architect or a Subconsultant in the preparation of its submittal to USGBC, an appeal is necessary, Architect shall provide as part of Basic Services the services, including revisions to its submittals, that are necessary to the prosecution of the appeal to a final resolution.

ARTICLE 3 ADDITIONAL SERVICES

3.1 **DEFINITION**

Additional Services are services, which if authorized by and performed in accordance with this Agreement, are paid for by County in the form of Additional Services Compensation. Additional Services consist solely and exclusively of those services listed in Section 3.2, below, and those other services marked by a "No" designation in the Description of Basic, Additional and Exhibit "B" attached hereto. Additional Services, whether or not listed in this Article 3 or the Description of Basic, Additional and Excluded Services - Exhibit "B" attached hereto, do not include any service that arises, in whole or in part, from the breach of this Agreement by Architect, an act or omission of Architect or a Subconsultant constituting negligence (ordinary or professional), willful misconduct or violation of Applicable Laws or information contained in the Design Documents constituting errors, omissions, conflicts, ambiguities or violations of Applicable Laws.

3.2 ENUMERATION OF ADDITIONAL SERVICES

In addition to those services listed in the <u>Description of Basic</u>, <u>Additional and Excluded Services</u> - <u>Exhibit</u> "B"" that are marked "No" (indicating that they are not part of Basic Services), the following services constitute Additional Services:

- **3.2.1** preparing revisions to the Design Development Documents or Construction Documents that are: (1) required to implement changes, additions or deletions to the Final Program that are material in scope and are requested and approved by County in writing after County's approval of the completed Schematic Design Documents, or (2) the result of issuance by County of written directives or instructions that materially conflict with prior written directives or instructions by County; or (3) required by reasonably unforeseeable enactments or revisions of Applicable Laws enacted subsequent to the preparation of the portion of the Design Development Documents or Construction Documents affected thereby:
- **3.2.2** providing services as necessary to correct Defective Work furnished by Contractor or a Separate Contractor;
- **3.2.3** providing services in connection with a mediation, arbitration, or legal proceeding, except where a party to such proceeding has alleged in good faith the occurrence of any of the following: (1) breach of this Agreement by Architect; (2) an act or omission of Architect or a Subconsultant constituting negligence, willful misconduct, or violation of Applicable Laws; or (3) that the Design Documents prepared by Architect or a Subconsultant contain information constituting errors, omissions, conflicts, ambiguities, or violations of Applicable Laws;
- **3.2.4** providing services made necessary by termination of the Contractor or a Separate Contractor, but only to the extent such services exceed the level of services that would have been provided in the absence of such termination:
- **3.2.5** providing services to make revisions to the Design Development Documents or Construction Documents due to a substantial inaccuracy in any surveys, test data or other information provided by County pursuant to <u>Section 4.2</u> of this Agreement under circumstances in which the inaccuracy was not known or reasonably discoverable by Architect or a Subconsultant at the time the portion of the Design Documents being revised was originally prepared by Architect or such Subconsultant;
- **3.2.6** providing Basic Services for the Construction Phase of a Project that, solely due to circumstances giving rise to a right of Architect to an extension of time under <u>Paragraph 2.16.3</u>, above, are required to be performed following the sixtieth (60th) Day after (1) the original date for Final Completion established by the Construction Contract at the time the Construction Contract is executed by County and Contractor; plus (2) the number of Days of any Periods of Inactivity under this Agreement; provided, however, that nothing herein shall be interpreted as entitling Architect to be paid duplicative compensation (both as Additional Services and Basic Services) for services covered under and compensated by its Basic Services Compensation; or
- **3.2.7** subject to the provisions of <u>Paragraph 2.13.6</u>, above, if the scope of Basic Services set forth in <u>Description of Basic</u>, <u>Additional and Excluded Services Exhibit "B"</u> attached hereto expressly limits Basic Services to a specific number of hours, instances or occasions, providing services that are requested by County that exceed the number of hours, instances or occasions that are agreed to be provided as part of Basic Services.

3.3 AUTHORIZATION BY COUNTY

Additional Services shall be performed only if authorized and directed in writing by County in advance and in accordance with this <u>Article 3</u>.

3.4 NOTICE TO COUNTY

Architect shall notify County in writing within five (5) Days after learning of any circumstance (including, without limitation, any direction or request by County or other Project Team member) that Architect believes may give rise to performance of Additional Services. Unless otherwise provided in <u>Section 3.5</u>, below, Architect waives the right to compensation for Additional Services performed without prior written approval by the Director of Facilities Management expressly acknowledging that the service is an Additional Service.

3.5 **DISPUTES**

If a good faith dispute arises as to whether a particular service performed or to be performed is a Basic Service or an Additional Service, Architect will, if requested to do so by County in writing, nevertheless promptly perform such service and pay any expenses associated with such performance, pending resolution of such dispute. Neither County's request, Architect's performance or County's acceptance of such performance nor County's payment therefor will constitute a waiver on the part of County or Architect of their respective rights or defenses with respect to the appropriate classification of such service or expense, which rights and defenses shall remain subject to determination in accordance with Article 11, below.

ARTICLE 4 COUNTY RESPONSIBILITIES

4.1 ADMINISTRATION BY COUNTY

- **4.1.1 County Requirements.** County shall provide information regarding its objectives, schedule, constraints, criteria, space requirements and relationships, flexibility, expandability, special equipment and Site requirements.
- **4.1.2 County Approvals.** County shall promptly respond to Architect's submittals and requests for decisions, approvals or information; provided, however, that no failure by County to respond shall entitle Architect to an adjustment of the Master Project Schedule or Architect's Project Schedule except as permitted by Section 2.16, above. County's review, approval or evaluation of Design Documents shall neither be interpreted as an assumption by County of any responsibility or liability for the technical accuracy or completeness of the Design Documents nor as relieving Architect of any liability or responsibility under this Agreement.
- **4.1.3 Director of Facilities Management.** Subject in all cases to approval by the Board of Supervisors as required by Applicable Laws, the Director of Facilities Management is the sole representative of County with authority on behalf of County to: (1) approve or revise the Final Program or Fixed Limit; (2) approve of any extension to the Master Project Schedule or Architect's Project Schedule; (3) authorize the performance of Additional Services; or (4) authorize the incurring of Reimbursable Expense in amounts that exceed any applicable monetary limitations agreed to in writing between County and Architect.
- **4.1.4 Board Approval.** Notwithstanding anything stated in this Agreement to the contrary, only the Board of Supervisors has authority on behalf of County to commit or bind County to an obligation to pay any sum of money or additional compensation that exceeds or is beyond the limits of the agreed Basic Services Compensation.

4.2 SURVEYS, SERVICES AND REPORTS

4.2.1 Reference Documents. County has provided or made available to Architect for its review the Reference Documents listed in the Reference Documents List - Exhibit "J" attached hereto. Execution of

this Agreement shall be deemed acknowledgement by Architect that it has received and read the Reference Documents and in entering into this Agreement has taken into consideration the matters disclosed thereby. County makes no representation that the Reference Documents so listed constitute all of the documents in its custody that may be pertinent to the matters addressed in said Reference Documents or pertinent to Architect's preparation of its Design Documents for the Project.

- **4.2.2 Testing and Inspection.** County shall directly contract for and pay the costs of such special testing and inspection that, based upon the recommendations of Architect pursuant to <u>Section 2.11</u>, above, are specified by Architect in the Final Construction Documents.
- **4.2.3 Land Survey.** If requested by Architect and required for the performance of this Agreement by Architect, County shall furnish a land survey of the Site and the location on the Site of Existing Improvements, prepared by a licensed surveyor, including grades and lines of streets, alleys, pavements and adjoining property and locations, dimensions and elevations of trees and Existing Improvements above the surface of the ground.
- **4.2.4 Geotechnical.** County shall, based upon the recommendations of Architect, furnish geotechnical data and reports, or employ County Consultants to provide such data or reports, when reasonably deemed necessary by Architect, including test logs, soil classifications, soil bearing values and other data and information reasonably appropriate to identifying and evaluating subsoil conditions at the Site.
- **4.2.5 County's Expense.** The services, information, surveys and reports to be furnished by County as required under this <u>Section 4.2</u> shall, except as otherwise provided in this Agreement, be furnished at County's expense.
- **4.2.6 No Warranty by County.** In performing its obligations under this Agreement, Architect shall be entitled to rely upon the accuracy and sufficiency of surveys, data, reports or other information furnished by County pursuant to this Article 4 (including, without limitation, the Reference Documents and any drawings, design or technical specifications furnished by County as referenced in Paragraph 4.2.8, below); however, County does not impliedly or expressly warrant the accuracy or suitability thereof and County shall have no liability to Architect (other than as permitted by Article 3, above, for authorized Additional Services Compensation) in the event that such surveys, data, reports or other information are found to be inaccurate, incomplete or insufficient.
- **4.2.7 Notice of Defects.** County and Architect shall each provide prompt written notice to the other party if either becomes aware of any defect or deficiency in the Design Documents.
- **4.2.8 County-Furnished Designs.** Drawings, designs or technical specifications provided by the County to the Architect for the Architect's use in connection with its performance under this Agreement (including, without limitation, drawings, designs or technical specifications prepared by Architect for other projects previously designed by Architect for County or prepared by other architects, engineers or design professionals, and that County directs be replicated, in whole or in part, by Architect in its Design Documents) are provided for the convenience of the Architect with the understanding that Architect will, based upon a careful and thorough review of them, make any changes or revisions that Architect deems appropriate or necessary before using or incorporating any portion thereof into the Design Documents for the Project. Any portion thereof that is used or incorporated by Architect or any Subconsultant in the Design Documents shall be the sole risk of Architect who shall be solely responsible for the accuracy, completeness and sufficiency thereof to the same extent as if such incorporated drawings, designs or technical specifications were originally prepared by Architect solely for use on the Project.

4.3 ACCESS TO SITE

Architect shall at all times during performance of this Agreement have access to the Site and to the Work, at whatever stage the Work is in its preparation or progress, to facilitate Architect's performance of its obligations under this Agreement. Architect is responsible to see to it that employees of Architect and its Subconsultants, at all times while present on the Site of the Project, comply with the safety requirements applicable to the Site, including, but not limited to the requirements of the safety programs of Contractor and Separate Contractors performing Work on the Site.

4.4 APPROVALS

Approvals by or on behalf of County of Design Documents prepared by Architect, or of any design or engineering services performed under this Agreement, shall not, under any circumstances, be interpreted as relieving Architect from its sole responsibility to produce and prepare such documents and perform such services in a manner that strictly complies with this Agreement, Applicable Laws and the standards of performance applicable to Architect's performance under this Agreement.

ARTICLE 5 ARCHITECT'S COMPENSATION

5.1 BASIC SERVICES COMPENSATION

- **5.1.1 Total Compensation.** Architect shall be paid a total Basic Services Compensation for performance of Basic Services (including, without limitation, Basic Services performed by Subconsultants) comprised exclusively of (1) Basic Services Fees plus (2) authorized Reimbursable Expenses. Basic Services Compensation constitutes the Architect's sole, exclusive and complete compensation for performance of Basic Services, including, without limitation, all costs and expenses, of any kind, incurred by Architect or its Subconsultants in performance of Basic Services.
- **5.1.2** Basic Services Fees. Subject to the provisions of <u>Paragraph 5.1.4</u>, below, Architect's Basic Services Fees for performance of Basic Services shall be as follows: [Check appropriate box(es)]:

.1 a Fixed Fee for a	II Basic Services of \$N/A;
Subconsultants' professional personnel in per	(1) the actual hours expended by Architect's and its formance of Basic Services multiplied by (2) the applicable of which shall not exceed a Maximum Hourly Fee of
.3 a Basic Services F both Fixed Fees and Maximum Hourly Fees	Fee based on a combination of compensation comprised of for each of the following categories of Basic Services:
Basic Services Description:	Basic Services Fees: [Check appropriate box]
(1) Architectural and	
Consulting Engineering Svcs.:	☑ Fixed Fee: <u>\$175,300.00</u>
	☐ Maximum Hourly Fee: \$_N/A;
(2) Reimbursable Expenses	
	☑ Fixed Fee: <u>\$800.00</u>
	☐ Maximum Hourly Fee: \$_N/A;

29 of 53

(3)N/A	
	☐ Fixed Fee: \$_N/A
	☐ Maximum Hourly Fee: \$N/A

- **5.1.3 Guaranteed Amounts.** An amount agreed to by County and Architect pursuant to <u>Paragraph 5.1.2</u>, above, as a Fixed Fee or Maximum Hourly Fee represents the County's maximum liability to Architect for the complete performance by Architect and its Subconsultants of the Basic Services or portion of Basic Services covered by such Basic Services Fees. Subject only to Architect's rights under <u>Section 5.2</u> below, any fees, costs or expenses, of any kind, incurred by Architect or a Subconsultant, for performance of Basic Services or a portion of Basic Services for which a Fixed Fee or Maximum Hourly Fee has been agreed to in <u>Paragraph 5.1.2</u>, above, that if charged to County would exceed the amount of such Fixed Fee or Maximum Hourly Fee shall be deemed incurred at Architect's Own Expense.
- **5.1.4** Basic Services Allowances. Basic Services Allowances represent estimated Basic Services Fees for a category of Basic Services for which the County and Architect have not agreed upon a Fixed Fee or Maximum Hourly Fee. Architect does not represent, agree or guarantee that the Basic Services Fees that may be required to be paid by County for complete performance of Basic Services that are within the scope of the description of a Basic Services Allowance will not exceed the agreed amount assigned to such Basic Services Allowance. Notwithstanding the foregoing, Basic Services Fees payable by County for Basic Services that are covered by a Basic Services Allowance shall not exceed the amount of the Basic Services Allowance set forth below without the prior written approval of County and absent such approval are deemed performed at Architect's Own Expense. In the event Basic Services Fees for Basic Services that are covered by a Basic Services Allowance are less than the amount of the Basic Services Allowance, set forth below, County shall only pay and be liable for the actual amount of such Basic Services Fees earned by Architect.

Basic Services Allowance:	Amount:	
(1)N/A		
	\$N/A	
(2) <u>N/A</u>		
	\$N/A	

- **5.1.5 Design Contingency Allowance.** Design Contingency Allowance shall provide for designated Consulting activities in the Work specified in this Agreement at a fixed amount exclusive of the total compensation amounts agreed to by County and Architect pursuant to <u>Paragraph 5.1.2</u> and <u>Paragraph 5.1.4</u>, above, for the complete performance by Architect and/or its Subconsultants, and described in the requirements:
 - .1 the Design Contingency Allowance is used only as directed by the County.
 - .2 the Design Contingency Allowance is used exclusively for the County's purposes and for the defined Scope of Work.
 - .3 the Architect will prepare a detailed breakdown of all costs associated with the work defined for the Design Contingency Allowance.
 - at project closeout, unused Design Contingency Allowance shall be credited to the

County.

Design Contingency Allowance :	Amount:
(1) <u>N/A</u>	\$

5.2 REIMBURSABLE EXPENSES

- **5.2.1** Exclusive List. Reimbursable Expenses include, and are limited to, the a reasonable amount for the following costs and expenses if and to the extent they are incurred and paid by Architect in the performance of Basic Services or Additional Services and not as a result of the negligence, willful misconduct or violation of an Applicable Law by Architect or its Subconsultants or the failure by Architect to comply with the requirements of this Agreement:
- .1 if approved in advance by County, mileage for vehicle travel (at the rates set forth in the Reimbursable Expenses Schedule Exhibit "G" attached hereto), air travel (coach fare only) and related subsistence (meals and lodging at standard business accommodation rates) for travel from Architect's or a Subconsultant's place of business (whether located within or outside the County of Riverside) to a point of destination outside the County or Riverside, but excluding the following: (1) travel and related subsistence to or from the County's offices or the Site for purposes of conducting inspections, observations or attending meetings that are part of Basic Services; (2) travel to and from residences to the Architect's or a Subconsultant's place of business; and (3) travel to or from Architect's or a Subconsultant's place of business located outside the County of Riverside to a location within the County of Riverside.
- .2 printing and reproduction (paper and electronic) of Design Documents, at the agreed rates set forth in the Reimbursable Expenses Schedule Exhibit "G" attached hereto, that under the terms of this Agreement are required to be delivered to County or that County requests or approves be provided to another member of the Project Team (by way of example, without limitation, costs of printing or reproduction for internal uses by, or for copies transmitted between or among, Architect and/or its Subconsultants are not reimbursable);
- .3 fees for permits or approvals of Governmental Authorities paid for by Architect on behalf of County as requested by County;
- .4 costs of renderings or mockups requested by County that are in addition to those required as part of Basic Services; and
- .5 costs listed in <u>Subparagraphs 5.2.1.1 through 5.2.1.4</u>, above, incurred and paid by Subconsultants in the performance of Basic Services or Additional Services; provided that (1) such costs are due and payable by Architect pursuant to terms of a contract approved by County pursuant to <u>Section 1.6</u>, above; (2) such costs are not included in or covered by any fixed fee agreed to by the Subconsultant under the terms of the Subconsultant's contract; and (3) such costs are not in excess of any not-to-exceed amount applicable thereto under the terms of the Subconsultant's contract.
- **5.2.2** Approval Limitations. Reimbursable Expenses shall not exceed, either individually or in the aggregate, the limits set forth in the <u>Reimbursable Expenses Schedule Exhibit " G "</u> attached hereto without the prior written approval of County. Reimbursable Expenses incurred without such approval shall be deemed incurred at Architect's Own Expense.
- **5.2.3** Mark Ups. Neither the Architect nor any Subconsultant shall include or charge any markup or multiplier upon any Reimbursable Expense, save and except for such markups or multipliers as may be permitted, if at all, by the terms of the Reimbursable Expenses Schedule Exhibit "G" attached hereto.

5.2.4 Expense Records. In addition to Architect's obligations under <u>Section 6.3</u>, below, accurate and detailed records of Reimbursable Expenses shall be maintained in an orderly manner on the basis of generally accepted accounting practices and shall be available at Architect's office (or at County's request, shall be brought by Architect to County's offices) for inspection, auditing and/or copying by County and its representatives pursuant to Article 7, below.

5.3 ADDITIONAL SERVICES COMPENSATION

- **5.3.1 Additional Services Compensation.** Architect shall be paid a total Additional Services Compensation for performance of Additional Services comprised exclusively of Additional Services Fees plus authorized Reimbursable Expenses. Additional Services Compensation constitutes the Architect's sole, exclusive and complete compensation for Additional Services, including, without limitation, all costs and expenses, of any kind, incurred in connection with Architect's and its Subconsultants' performance of Additional Services.
- **5.3.2** Additional Services Fees. Prior to performance of an Additional Service, Architect and County shall attempt in good faith to negotiate terms for Additional Services Fees on the basis of either: (1) a lump sum price; or (2) actual hours expended multiplied times the Hourly Rates for the personnel involved in providing such Additional Service as set forth in the Hourly Rates Schedule Exhibit "H" attached hereto, not-to-exceed an agreed maximum amount. In addition to County's rights under Section 5.6, below, if the parties are unable to agree, then the County shall have the right, without limitation, to direct in writing that Architect perform the Additional Services based on actual hours expended at the agreed Hourly Rates, without a not-to-exceed amount, plus reimbursement of authorized Reimbursable Expenses. Additional Services performed without prior written authorization pursuant to this Paragraph 5.3.2 or written direction pursuant to Section 5.6, below, shall be deemed performed at Architect's Own Expense.
- **5.3.3 Direct Engagement.** County reserves the right, without thereby being considered in breach of this Agreement, to contract for the performance of Additional Services by others.

5.4 **HOURLY RATES**

Hourly Rates for Basic Services and Additional Services performed on an hourly basis are set forth in the <u>Hourly Rates Schedule - Exhibit "H"</u> attached hereto. Hourly Rates shall remain fixed for the duration of Architect's performance of this Agreement.

5.5 RELEASE FOR PRIOR SERVICES

Architect waives and releases County from any obligation or liability for payment of money or compensation for services, of any kind, performed and for costs or expenses, of any kind, incurred, prior to execution of this Agreement by Architect.

5.6 **DISPUTES**

If a good faith dispute arises as to whether a service is Basic Services or Additional Services or whether an expense is reimbursable as a Reimbursable Expense, Architect will nevertheless promptly perform such service and pay such expense, if requested to do so by County in writing, pending resolution of such dispute. Neither County's request, Architect's performance nor County's payment therefor or thereof will constitute a waiver on the part of County or Architect of their respective rights or defenses with respect to the appropriate classification of such service or expense, which rights and defenses shall remain subject to determination in accordance with Article 11, below.

5.7 NO WAIVER OR RELEASE OF RIGHTS

Neither authorization nor payment by County of any amount for Basic Services, Additional Services or Reimbursable Expenses shall be interpreted as a waiver, release or settlement of any rights or claims that County may have: (1) for Losses resulting from the fault, negligence or willful misconduct of the Architect or its Subconsultants or the breach by Architect of an obligation under this Agreement; or (2) to recoup and recover from Architect amounts paid by County that were not in fact due and owing to Architect under the terms of this Agreement at the time they were paid.

ARTICLE 6 PAYMENTS TO ARCHITECT

6.1 INVOICES FOR PAYMENT

- **6.1.1 Invoice Submission.** On the 1st day of each month, Architect shall submit to County an accurate and complete Invoice for Payment, using the <u>Invoice for Payment Form Exhibit "N"</u> attached hereto, signed by Architect and requesting payment for the preceding thirty (30) Day period as follows:
- .1 Basic Services Fees. Amounts included in an Invoice for payment for Basic Services Fees shall be computed in accordance with the following provisions, as applicable:
- (1) Fixed and Maximum Fees. Amounts included by Architect in its Invoices for Payments for Basic Services Fees on account of Basic Services or any portion of Basic Services for which a Fixed Fee or a Maximum Hourly Fee has been agreed to in Section 5.1, above, shall not exceed a pro-rated portion of the agreed Basic Services Fees based on the product of (1) the percentage of completion of such Basic Services that has been actually achieved by Architect, multiplied times (2) the agreed Fixed Fee or Maximum Hourly Fee applicable to such Basic Services; and provided further, that where such Basic Services or portion of Basic Services are to be performed in Phases, such pro-rated portion shall be proportionate to and shall not exceed for any Phase of such Basic Services or portion of Basic Services, the percentage of such Basic Services Fees that is assigned to such Phase in the Payment Schedule Exhibit "I " attached hereto.
- (2) Basic Services Allowances. Amounts included by Architect in its Invoices for Payments for Basic Services Fee on account of Basic Services covered by a Basic Services Allowance shall not, prior to the time that the Basic Services Fees incurred equal the Basic Services Allowance amount, exceed a pro-rated portion of the agreed Basic Services Allowance amount based on the product of (1) County's Good Faith Determination of the percentage of completion of such Basic Services Allowance that has been actually achieved by Architect, multiplied times (2) the agreed Basic Services Allowance amount. If and when the Basic Services Fees incurred for Basic Services covered by a Basic Services Allowance equal the Basic Services Allowance amount, the Architect shall, if County has previously approved in writing of further Basic Services Fees being incurred in excess of the amount of the Basic Services Allowance, thereafter include in its Invoices for Payment the actual amount of such Basic Services Fees incurred up to, but not exceeding, any additional maximum amount authorized by County as part of such approval.
- .2 Additional Services Fees. Architect's Invoice for Payment shall include amounts for Additional Services Fees earned for the proper performance of Additional Services authorized pursuant to Article 3, above. Each item of Additional Services shall be separately itemized, in accordance with the following methods of calculation, as applicable:
- (1) Lump Sum: If the agreed Additional Services Fees are based on a lump sum price, by taking the County's Good Faith Determination of the percentage of the Additional Services properly completed and multiplying that percentage times the agreed lump sum price for such Additional

Services and subtracting therefrom payments previously made on account thereof.

(2) Hourly/Not-to-Exceed: If the Additional Services Fees are based on an hourly compensation, by taking the number of hours of Additional Services performed during the thirty (30) Day period covered by the Invoice for Payment and multiplying those hours times the applicable Hourly Rates for such personnel; provided, however, that if the parties have agreed to a not-to-exceed amount for such Additional Services Fees, then under no circumstances shall the total of the amounts paid and payable by County for such Additional Services Fees at any time exceed a pro rata share of the agreed not-to-exceed amount for such Additional Services based on County's Good Faith Determination of the percentage of such Additional Services properly completed in accordance with this Agreement multiplied times the agreed not-to-exceed amount.

.3 Reimbursable Expenses. Architect's Invoice for Payment shall include amounts for authorized Reimbursable Expenses incurred and paid by Architect during the thirty (30) Day period covered by the Invoice for Payment that have not been previously reimbursed by County. Reimbursable Expenses associated with Basic Services and Additional Services shall be separately itemized. Reimbursable Expenses for Additional Services shall be further separately itemized to correspond to the Additional Service for which they were incurred and paid.

6.2 PAYMENT SCHEDULE FOR BASIC SERVICES

The County's obligation for payment of Basic Services Fees for any Phase of Basic Services shall under no circumstances exceed a pro rata share of either the lump sum amount or Maximum Hourly Fee (as applicable pursuant to Section 5.1, above) that County is obligated to pay for Basic Services Fees. Such pro rated share shall be calculated based on the percentages assigned to each Phase of Basic Services in the Payment Schedule - Exhibit "I "attached hereto. In cases where only a portion of a Phase is completed, the amount payable shall not exceed County's Good Faith Determination of the percentage of Basic Services completed within that Phase expressed as a separate percentage of the percentage of overall Basic Services allocated in the Payment Schedule to that Phase.

6.3 ACCOMPANYING DOCUMENTATION

Each Invoice for Payment shall be accompanied by the following:

- **6.3.1** in the case of Basic Services and Additional Services performed and compensated on an hourly basis, detailed time summaries for Basic Services and Additional Services performed during the period of time covered by the Invoice for Payment that are broken down by time keeper, task and time expended (block billings are not permitted) and copies of all time sheets prepared by any time keeper who performed any part of the Basic Services and Additional Services which are the subject of the Invoice for Payment and which reflect or record such Basic Services and Additional Services;
- **6.3.2** copies each of the invoices, receipts and other documentation verifying the amounts of Reimbursable Expenses for which reimbursement is sought in the Invoice for Payment, along with a tally of all Reimbursable Expenses requested in the Invoice for Payment the sum of which totals the total amount of Reimbursable Expenses for which reimbursement is sought by Architect in the Invoice for Payment;
- **6.3.3** conditional waivers and releases of stop payment notice rights executed by Architect and its Subconsultants, of every Tier, using the <u>Release Forms Exhibit "M"</u> attached hereto, conditionally releasing to the fullest extent allowable by Applicable Laws all stop payment notice rights for all services performed and costs incurred during the period of time covered by the then-current Invoice for Payment;
- **6.3.4** unconditional waivers and releases of stop payment notice and bond rights executed by Architect and its Subconsultants, of every Tier, using the Release Forms Exhibit "M" attached hereto

unconditionally releasing to the fullest extent allowable by Applicable Laws all stop payment notice and bond rights for all services performed and costs incurred during the period of time covered by the Invoice for Payment immediately preceding the current, pending Invoice for Payment; and

6.3.5 such other documentation substantiating Architect's or its Subconsultants' charges or time as may be reasonably requested by County.

6.4 REVIEW AND PAYMENT

- **6.4.1** Review by County. County shall, within fourteen (14) Days after receipt of an Invoice for Payment prepared and submitted in accordance with this Agreement, notify Architect if the Invoice for Payment is approved or rejected, in whole or in part, along with an explanation of the reason(s) for any disapproval.
- **6.4.2** Payment by County. Payment of undisputed amounts included in an Invoice for Payment prepared and submitted in accordance with this Agreement shall be made by County monthly within thirty (30) Days after receipt by County of the Invoice for Payment requesting payment that is prepared and submitted in accordance with this Agreement.

6.5 **PAYMENT DISPUTES**

Without limitation to County's rights under <u>Section 6.6</u>, below, in the event there is a good faith dispute over a request for payment included in an Invoice for Payment, County shall have the right to either: (1) make all or part of such disputed payment to Architect without prejudice to County's right to contest the amount so paid; or (2) withhold only the amount of such payment as to which County makes a Good Faith Determination that there is a dispute and provide to Architect written notice of the reason(s) for such withholding. County and Architect shall use their good faith efforts to attempt to resolve their dispute as quickly as practicable under the circumstances. Architect shall not be entitled to terminate this Agreement or suspend performance of its services hereunder on account of such nonpayment provided that County makes payment of all undisputed sums. If County withholds payment under Clause (2) of this <u>Section 6.5</u> and if it is determined subsequently that County's withholding was wrongful, County shall pay such amount to Architect plus any penalties that may be due pursuant to California Civil Code Section 3320; no additional amounts shall be payable to Architect for interest on such unpaid amounts. If County chooses to proceed under Clause (1) of this <u>Section 6.5</u> and it is subsequently determined that County overpaid Architect, Architect shall refund to County the amount of such payment plus accrued interest computed at the Interest Rate from the date of such overpayment until refunded.

6.6 WITHHOLDING BY COUNTY

Without limitation to County's rights under California Civil Code Section 3320, County shall have the right, after written notice to Architect, to withhold from payment to Architect 150% of the amount of any Loss resulting from, or threatened as a result of, the negligence of Architect or a Subconsultant or a failure by Architect to perform an obligation under this Agreement. Such withholding shall not constitute a final determination or waiver of any rights or liabilities of County or Architect with respect to responsibility for such Loss, which rights and liabilities shall remain subject to determination in accordance with Article 11 of this Agreement.

6.7 LIENS, STOP PAYMENT NOTICES, CLAIMS

Except as otherwise provided herein, Architect shall not permit to be created or to remain undischarged any lien, encumbrance, stop payment notice, claim or charge (collectively, "lien") which arises out of, or relates to, the provision by Architect or its Subconsultants of any services or things under this Agreement upon the property of County, the construction fund of County, or the income from any such property or construction fund, or any part thereof, or suffer any other matter or thing whereby the estate, rights and

interest of County in the Project property or construction fund, or any part thereof, might be impaired. If any such lien is filed, then within thirty (30) Days after notice of filing thereof Architect shall cause the same to be fully discharged of record, released and removed by any lawful means available, such as, but not limited to, payment, deposit, bond, order of a court of competent jurisdiction or otherwise. If Architect shall fail to cause such lien to be so discharged within the period aforesaid, then, in addition to any other right or remedy, County may, but shall not be obligated to, discharge the lien by any means, including, but not limited to, withholding amounts pursuant to Section 6.6, above, paying the amounts claimed to be due (including, without limitation, interest and attorney's fees claimed due), bonding or any other means that County determines, in its sole and absolute discretion, appropriate. Any Loss incurred by County as a result of Architect's failure to comply with its obligations under this Section 6.7 shall be paid by Architect to County on demand. Architect shall be excused from its obligations under this Section 6.7 with respect to, but only to the extent of, amounts included in a lien that are unpaid to the claimant upon the lien as the direct result of County's breach of its payment obligations to Architect under this Agreement.

ARTICLE 7 ACCOUNTING, RECORDS AND AUDIT

7.1 FINANCIAL MANAGEMENT

Architect shall set up and exercise accounting and control systems for the proper financial management of its performance under this Agreement that are satisfactory to County, comply with the prevailing custom and practice for similar projects and afford County the ability to verify all charges and duplicate all calculations made by the Architect and Subconsultants.

7.2 RECORD KEEPING

- **7.2.1** Books and Records. Architect shall keep full and detailed books and records concerning the Project, including, without limitation, all documents (including, all hard copies and computer readable data, if it exists) that comprise or relate or refer to any of the following: (1) agreements, contracts, proposals, commitments, invoices, billings, statements, receipts, checks, certificates, releases, waivers, plans, specifications, notes, schedules, reports, studies, test data, approvals, permits, applications, diaries, logs, photographs, videos, shop drawings, samples, product data, job reports, change orders, field orders, directives, orders, bulletins, transmittals, requests for information, addenda, receipts, vouchers, correspondence, memoranda, messages, minutes, accounting records, job files, settlement agreements, and general ledgers; (2) any charge, cost or expense for which Architect seeks reimbursement or payment by County as part of any Invoice for Payment, Claim or other demand; and (3) any other documents that County, in its reasonable judgment, deems relevant to the Project.
- **7.2.2 Maintenance and Retention.** Architect shall at all times maintain such books and records in an organized and systematic form that allows for reasonably easy access and review and shall retain and preserve such books and records for a period of ten (10) years after the later of either final payment to Architect under this Agreement or Final Completion of the Project, or for such longer period as may be required by Applicable Laws.

7.3 INSPECTION, PRODUCTION AND AUDITING

Architect shall allow County and the auditor for the State of California (and their respective authorized representatives, auditors, and attorneys), not later than the third business day after written notice to Architect, full access at Architect's offices nearest to the Project to inspect, audit and copy any or all of Architect's books and records as described in Section 7.2, above. Architect shall, at Architect's Own Expense, furnish facilities and staff assistance for, and cooperate fully with, such inspection or audit. Audits by the County and the auditor for the State of California may be conducted jointly or separately. Upon request, Architect shall provide reproducible copies of such books and records for reproduction by or on

behalf of the person conducting the audit. Except as otherwise provided in <u>Section 7.4</u>, below, such reproduction shall be at the expense of the entity conducting the audit. The audit rights provided for under this <u>Section 7.3</u> may be exercised at any time, and as often, before or after Final Completion, as County or the auditor for the State of California deems, in its sole and absolute discretion, necessary.

7.4 NONCOMPLIANCE BY ARCHITECT

- **7.4.1** Cost of Audit. If an inspection or audit pursuant to Section 7.3, above, discloses that any amount (other than amounts permitted under the terms of this Agreement to be charged by Architect as lump or fixed fee charges) cannot be verified due to a failure by Architect or any Subconsultant to comply with this Article 7, has been improperly, inaccurately or excessively charged to County by Architect or any Subconsultant or has been overpaid by County, and if the total of such amounts for any calendar year audited is five percent (5%) or more of the total amount (exclusive of amounts permitted under the terms of this Agreement to be charged by Architect as lump or fixed fee charges) invoiced to County during such year, then Architect shall pay, at Architect's Own Expense, 100% of the actual cost to County and/or the State of California of such inspection or audit and any resulting report. If such inspection, audit or report is by County using in-house staff, then such actual cost to County shall be computed on the basis of two (2) times the direct payroll of the staff completing such inspection, audit or report.
- **7.4.2 County Remedies.** Without limitation to any of County's rights or remedies for recovery or withholding of any amounts from Architect as may be permitted by Applicable Laws or elsewhere in this Section 7.4 or this Agreement, if an inspection or audit pursuant to Section 7.3, above, discloses that an amount has been overpaid by County, then County shall have the right to withhold such amount from any payments due to Architect or if no payments are due Architect shall immediately reimburse such amount to County Amounts overpaid by County shall earn interest at the Interest Rate from the date of overpayment until the date reimbursed by Architect to County.
- **7.4.3 Withholding.** In addition, and without limitation upon any of the other provisions for withholding of payment that are set forth in this <u>Section 7.4</u> or elsewhere in this Agreement, County shall have the right to withhold from any payment to Architect an additional sum of up to ten percent (10%) of any amount claimed due by Architect until (other than amounts permitted to be charged by Architect as lump or fixed fee charges) has fully complied with any outstanding and unsatisfied request for performance by Architect of any obligation under this <u>Article 7</u>. Upon Architect's full compliance, such sum withheld under this <u>Paragraph 7.4.3</u> shall be released to Architect.
- **7.4.4 Legal Proceedings.** Architect's compliance with the requirements of this <u>Article 7</u> shall be a condition precedent to maintenance by Architect of any legal action or arbitration against County relating to Architect's or County's performance under or related to this Agreement.

7.5 **SUBCONSULTANTS**

Architect shall ensure that the provisions of this <u>Article 7</u> are included in all contracts entered into by Subconsultants, of every Tier, who perform services for the Project; provided, however, that Architect shall have the right to limit the scope of a Architect's obligation to allow for inspection or audit of books and records concerning actual costs of performance to costs that are related to: (1) costs of Subconsultant's administering its performance under its contract with Architect for the Project; (2) services that are performed on an hourly or cost reimbursement basis; (2) Additional Services; (3) cost or expenses that are payable on a reimbursement basis; and (4) Claims.

ARTICLE 8 DEFAULT, TERMINATION AND SUSPENSION

8.1 TERMINATION BY COUNTY FOR CAUSE

- **8.1.1 Default by Architect.** Architect shall cure any default in performance of its obligations under this Agreement within two (2) Days after receipt of written notice from County; provided, however, that if the breach cannot reasonably be cured within such time, then Architect will commence to cure the breach within two (2) Days and will diligently and continuously prosecute such cure to completion within a reasonable time, which shall in no event be later than ten (10) Days after receipt of such written notice. Nothing herein shall be interpreted as obligating County to give an opportunity to cure in the case of an emergency or if the default is of the type that County determines, in good faith, cannot be cured, or cannot be fully cured, within the time periods set forth in this Section 8.1.
- **8.1.2** Remedies Upon Default. In the event of any default by Architect, including, without limitation, a default that Architect fails to cure within the time periods set forth in Paragraph 8.1.1, above, then County may by written notice to Architect, effective upon Architect's receipt of such notice or upon such later date as may be set forth in such notice, pursue any remedies available under Applicable Laws, including, without limitation, the following:
- .1 Take-Over. County may, without terminating this Agreement, terminate or discontinue the Architect's performance and delete, take over or arrange for performance by others of some or all of the Basic Services and Additional Services, reserving to itself all rights to recover all Losses, including, without limitation, any Losses related thereto.
- .2 **Termination.** County may terminate this Agreement upon written notice, reserving to itself all rights to recover all Losses, including, without limitation, all Losses related thereto.
- **8.1.3** Rights Cumulative. All of County's rights and remedies under this Agreement are cumulative and shall be in addition to those rights and remedies available under Applicable Laws. No termination or other action taken by County after exercise of its rights under this <u>Article 8</u> shall prejudice any other rights or remedies of County provided by Applicable Laws or by this Agreement.
- **8.1.4 Disability, Insolvency.** In addition to the other rights granted to County under this Agreement or Applicable Laws, County shall have the right to terminate this Agreement for default by giving seven (7) days written notice to Architect, if: (1) Architect is an individual and should die or be adjudged incompetent; (2) Architect attempts to assign this Agreement; (3) a petition of bankruptcy is filed by Architect or Architect is adjudicated or admitted to be a bankrupt in connection with an involuntary petition of bankruptcy filed against Architect; (4) Architect should make a general assignment for the benefit of creditors; or (5) a receiver should be appointed on account of Architect's insolvency.
- **8.1.5** Architect Obligations. Upon Architect's receipt from County of notice of County's exercise of any of its rights under Paragraph 8.1.2, above, Architect shall, unless the notice directs otherwise, do the following:
- .1 immediately discontinue the performance of Basic Services and Additional Services to the extent specified in the notice;
- .2 provide to County a description, in writing, no later than seven (7) Days after receipt of the notice of termination, of all contracts with Subconsultants that are outstanding, including, without limitation, with respect to each such contract separately, the terms of the original price, payments made to date, the balance owing, the status of the services performed and any outstanding withholding of funds or default, and a copy of the contract and any written changes, amendments or modifications thereto, together

with such other information as County may determine necessary in order to decide whether it is in County's best interests to accept assignment of, or request Architect to terminate, the contract; and

- .3 thereafter only perform such Basic Services and Additional Services as may be necessary to complete the portion of the Basic Services and Additional Services not terminated, taken over or discontinued.
- **8.1.6** Completion by County. In the event County exercises its rights under Paragraph 8.1.2, above, County shall have the further right, without releasing Architect from liability for failure to fulfill this Agreement, to proceed to complete the Basic Services and Additional Services by any means that County determines is expedient and withhold all or a portion of the monies, if any, owing to Architect until County has completed such Basic Services and Additional Services.

8.1.7 Payment to Architect.

- Services that are terminated, discontinued or taken over by County pursuant to an exercise by County of its rights under Paragraph 8.1.2, above, and without limitation to County's other rights under this Agreement or Applicable Laws: (1) if the Losses to County, whether incurred or threatened, arising out of any default by Architect (whether or not such default was the subject of the County's notice of default) or County's exercise of its remedies for default by Architect, exceed the amount of Basic Services Compensation and Additional Services Compensation calculated pursuant to Sections 6.1 and 6.2, above, that was earned by Architect for such Basic Services and Additional Services performed up to, and not beyond, the effective date of such termination, discontinuance or take over by County, then Architect shall be liable to County for the difference and shall promptly remit same to County; or (2) if the sum of such Losses is less than the amount of such Basic Services Compensation and Additional Services Compensation, then County shall pay the difference to Architect within forty-five (45) Days after receipt by County of an Invoice for Payment prepared in accordance with this Section 8.1 and Sections 6.1 through 6.3, above, requesting payment of such Basic Services Compensation and Additional Services Compensation.
- **Continuing Services.** In the event of a partial termination, discontinuance or take over by County pursuant to an exercise by County of its rights under Paragraph 8.1.2, above, and without limitation to County's other rights under this Agreement or Applicable Laws: (1) with respect to that portion of Basic Services or Additional Services that is not terminated, discontinued or taken over by County, County shall make a Good Faith Determination of an adjustment in the Basic Services Compensation and Additional Services Compensation under this Agreement to reflect the reduction in the scope of Basic Services and Additional Services remaining to be performed by Architect; and (2) Architect shall continue performance of such Basic Services and Additional Services and shall be paid by County therefor in accordance with the terms of this Agreement.
- ,3 Conversion. In the event a termination, discontinuance or take over by County for cause pursuant to this <u>Section 8.1</u> is determined to be wrongful, Architect's right to payment or recovery shall be governed by the provisions of <u>Subparagraphs 8.2.2.1 through 8.2.2.3</u>, below, in lieu of any other rights, remedies or recovery provided for by Applicable Laws.

8.2 TERMINATION WITHOUT CAUSE

8.2.1 Termination for Convenience. Upon at least three (3) Days' written notice to Architect prior to the effective date of an exercise of a right under this <u>Section 8.2</u>, County shall have the right, in its sole and absolute discretion and without cause and for its convenience, to terminate, discontinue or take over all or any portion of this Agreement or Architect's performance under this Agreement. Upon receiving such notice, Architect shall, unless the notice directs otherwise, take the actions required by <u>Paragraph 8.1.5</u>, above.

8.2.2 Payment to Architect.

- .1 Terminated Services. With respect to any or all Basic Services and Additional Services that are terminated, discontinued or taken over by County pursuant to Paragraph 8.2.1, above, Architect shall, within seven (7) Days after exercise by County of a right to terminate, discontinue or take over pursuant to Paragraph 8.2.1, above, submit to County an Invoice for Payment prepared in accordance with Sections 6.1 through 6.3, above, for the amount of Basic Services Compensation and Additional Services Compensation that was earned by Architect for such Basic Services and Additional Services performed up to, and not beyond, the effective date of such termination, discontinuance or take over by County. Without limitation to County's rights under Sections 6.5 and 6.6, above, within forty-five (45) Days after receipt by County of an Invoice for Payment prepared in accordance with this Section 8.2, County shall pay to Architect the amount, if any, owing to Architect under this Paragraph 8.2.2.
- **Continuing Services.** In the event of a partial termination, discontinuance or take over by County pursuant to an exercise by County of its rights under Paragraph 8.2.1, above, and without limitation to County's other rights under this Agreement or Applicable Laws: (1) with respect to that portion of Basic Services or Additional Services that is not terminated, discontinued or taken over by County, County shall make a Good Faith Determination of an adjustment in the Basic Services Compensation and Additional Services Compensation under this Agreement to reflect the reduction in the scope of Basic Services and Additional Services remaining to be performed by Architect; and (2) Architect shall continue performance of such Basic Services and Additional Services and shall be paid by County therefor in accordance with the terms of this Agreement.
- .3 **Exclusive Remedy.** Architect agrees to accept the payments provided for under this Paragraph 8.2.2 as its sole and exclusive right and remedy in lieu of all other rights and claims that Architect may have under this Agreement or Applicable Laws for recovery of Losses caused or claimed to be caused by County's termination, discontinuance or takeover of this Agreement, including, without limitation, Losses associated with lost profits, lost opportunity, and other consequential damages.

8.3 SUSPENSION BY COUNTY

County shall have the right to order, in writing, a suspension of performance of all services by Architect without cause and for County's convenience. If services are entirely suspended by written order of County for a continuous period of more than sixty (60) consecutive Days, and such suspension is not due to a breach of this Agreement by Architect or the negligence, willful misconduct or violation of an Applicable Law by Architect or a Subconsultant, and if County thereafter requests in writing that Architect resume performance following such suspension, then Architect shall be entitled to payment as additional compensation of any unavoidable direct, out-of-pocket costs payable by Architect or Subconsultants to third-party vendors of supplies as a result of such suspension. No other adjustment to Architect's compensation and no other recovery by Architect or any Subconsultant of Losses associated with such suspension shall be permitted.

8.4 TERMINATION BY ARCHITECT

- **8.4.1** Architect's Remedies. If County fails within the applicable time period for payment provided for in Article 6, above, to make payment of sums that are not in good faith disputed by County and fails to cure such failure within thirty (30) Days after receipt of written notice of nonpayment from Architect, then, upon an additional ten (10) Days' written notice to County of intent to terminate, Architect may terminate this Agreement. The foregoing constitutes the Architect's sole and exclusive right to terminate this Agreement for any reason, including, but not limited to, any breach by County.
- **8.4.2** Payment to Architect. In the event of a termination by Architect pursuant to this <u>Section 8.4</u>, Architect's right to further payment or recovery shall be governed by the provisions of <u>Subparagraphs</u>

<u>8.2.2.1 through 8.2.2.3</u>, above, in lieu of any other rights, remedies or recovery provided for by Applicable Laws.

ARTICLE 9 INDEMNIFICATION

9.1 INDEMNIFICATION BY ARCHITECT

- 9.1.1 Basic Indemnity. To the fullest extent permitted by Applicable Laws, Architect agrees to defend (through legal counsel reasonably acceptable to County), indemnify, and hold harmless County of Riverside, its Agencies, Districts, Departments and Special Districts, Board of Supervisors, elected and appointed officials, and each of their respective directors, members, officers, employees, agents, representatives, and volunteers ("Indemnitee(s)"), and each of them, from any and all Losses that arise out of or relate to any act or omission constituting ordinary and not professional negligence (including, without limitation, negligent breach of contract), recklessness, or willful misconduct on the part of Architect or its Subconsultants, or their respective employees, agents, representatives, or independent contractors. Architect further agrees to and shall indemnify and hold harmless the Indemnitees from all liability arising from suits, claims, demands, actions, or proceedings made by agents, employees or subcontractors of Architect for salary, wages, compensation, health benefits, insurance, retirement or any other benefit not explicitly set forth in this contract and arising out of work performed for County pursuant to this Agreement. The Indemnitees shall be entitled to the defense and indemnification provided for hereunder regardless of whether the Loss is in part caused or contributed to by the acts or omissions of an Indemnitee or any other person or entity; provided, however, that nothing contained herein shall be construed as obligating Architect to indemnify and hold harmless any Indemnitee to the extent not required under the provisions of Paragraph 9.1.3, below.
- 9.1.2 Indemnity for Design Professional Services. To the fullest extent permitted by Applicable Law, Architect agrees to defend (through legal counsel reasonably acceptable to County), indemnify and hold harmless the Indemnitees, and each of them, against any and all Losses that arise out of, pertain to, or relate to, any negligence, recklessness or willful misconduct constituting professional negligence on the part of Architect or its Subconsultants, or their respective employees, agents, representatives, or independent contractors. The Indemnitees shall be entitled to the defense, and indemnification provided for hereunder regardless of whether the Loss is, in part, caused or contributed to by the acts or omissions of an Indemnitee or any other person or entity; provided, however, that nothing contained herein shall be construed as obligating Architect to indemnify and hold harmless any Indemnitee to the extent not required under the provisions of Paragraph 9.1.3, below. Architect shall defend and pay, all costs and fees, including but not limited to attorney fees, cost of investigation, and defense, in any loss, suits, claims, demands, actions, or proceedings to the extent and in proportion to the percentage, such costs and fees arise out of pertain to, or relate to the negligence, recklessness or willful misconduct of Architect arising out of or from the performance of professional design services under this Agreement. The duty to defend applies to any alleged or actual negligence, recklessness, willful misconduct of Architect. The cost for defense shall apply whether or not Architect is a party to the lawsuit and shall apply whether or not Architect is directly liable to the plaintiffs in the lawsuit. The duty to defend even applies even if Indemnitees are alleged or found to be actively negligent, but only in proportion to the percentage of fault or negligence of Architect.
- **9.1.3 Limitations on Indemnity Obligation.** Without affecting the rights of County under any other provision of this Agreement, Architect shall not be required to indemnify or hold harmless or provide defense or defense costs to an Indemnitee for a Loss due to that Indemnitee's negligence, recklessness or willful misconduct; provided, however, that such negligence, recklessness or willful misconduct has been determined by agreement of Architect and Indemnitee or has been adjudged by the findings of a court of competent jurisdiction.
- 9.1.4 Subconsultant Indemnity Agreements. Architect agrees to obtain or cause to be obtained

executed defense and indemnity agreements with provisions identical to those set forth in this <u>Section 9.1</u> from each and every Subconsultant, of every Tier.

- **9.1.5 No Limitation by Insurance**. Architect's indemnification obligations under this Agreement shall not be limited by the amount or type of damages, compensation or benefits payable under any policy of insurance, workers' compensation acts, disability benefit acts or other employee benefit acts.
- **9.1.6** Enforcement. The Indemnitees shall be entitled to recover their attorneys' fees, costs and expert and consultant costs in pursuing or enforcing their right to defense and/or indemnification under this Agreement.

ARTICLE 10 INSURANCE

10.1 ARCHITECT'S INSURANCE

- 10.1.1 Required Coverages. Prior to the commencement of any services, Architect shall, at its own expense, purchase from, and maintain with, a company or companies lawfully authorized and approved by Governmental Authorities to do business in the jurisdiction in which the Project is located and having an A.M. Best Company rating of no less than A-8, the insurance coverages set forth in this Section 10.1, which coverages shall remain in force throughout Architect's performance of this Agreement and for such longer periods as may be required by this Agreement, unless such requirements are waived, in writing, by the County's Risk Manager. If the County's Risk Manager waives a requirement for a particular insurer, such waiver is only valid for that specific insurer and only for one policy term after which full compliance with this Section 10.1 shall be required. Except as otherwise expressly provided in this Section 10.1, such policies and coverages shall, without limitation, protect Architect from claims which may arise out of, or result from, the Architect's performance of this Agreement, whether such performance be by itself or by any Subconsultant, or by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable, and shall comply with the following requirements:
- Services Office ("ISO") Commercial General Liability Coverage "occurrence" form CG20101185 (Form B) or ISO Comprehensive General Liability "occurrence" form acceptable to the County with the Broad Form Comprehensive General Liability Endorsement GLO404 (with no Property Damage Liability exclusions pertaining to loss by explosion, collapse or underground damage), including, without limitation, coverage for bodily injury, sickness, disease, or death of any person, injury to, or destruction of tangible property, including loss of use resulting therefrom, blanket contractual liability coverage (including, without limitation, coverage for the Architect's indemnification obligations set forth in Article 9, above), and including an endorsement amending the aggregate limits to apply on a per location or per project basis, with limits of liability coverages of no less than the following amounts:

\$1,000,000	General Aggregate (Other Than Products-Completed Operations)
\$1,000,000	Products-Completed Operations Aggregate Limit for a period of five (5) years
	following Final Completion and Acceptance of the Project
\$1,000,000	Personal and Advertising Injury Limit
\$1,000,000	Per Occurrence Limit

.2 Professional Liability insurance, issued on a "claims made" basis, with limits of liability coverage in the amounts of no less than the following: (1) if the Fixed Limit is \$5 million or less: \$1,000,000 per claim and \$1,000,000 in the annual aggregate; (2) if the Fixed Limit is over \$5 million and \$10 million or less: \$2,000,000 per claim and \$2,000,000 in the annual aggregate; and (3) if the Fixed Limit is over \$10 million: \$5,000,000 per claim and \$5,000,000 in the annual aggregate. Such policy shall provide coverage (including, without limitation, all costs and expenses resulting from the investigation and defense

of any claim) for damages from claims for bodily injury or property damage to County or to any third party (including, without limitation, loss of use of damaged and non-damaged property) due to any breach of duty in the performance of professional services. Professional liability coverage shall have an inception date or a retroactive date coinciding with, or prior to, the date of execution of this Agreement or the date of first performance of any services under this Agreement, whichever date is earlier, and coverage shall continue uninterrupted until five (5) years after Final Completion and Acceptance of the entire Project. Coverage for such post-completion period may be provided by renewal or replacement of the policy for each of five (5) years or by a five-year extended reporting period endorsement that reinstates the aggregate limit for the extended reporting period. Renewal or replacement policies shall not allow for any advancement of the retroactive date. Any deductible or self-insured retention under the foregoing professional liability policy shall not, except with the approval of County granted or withheld in the County's sole and absolute discretion, exceed \$100,000.

- .3 Motor Vehicle Liability insurance issued on an ISO Business Auto Coverage form, including Symbol 1, acceptable to the County with limits of liability coverage of not less than \$1,000,000 per occurrence combined single limit for bodily injury and property damage for all owned, hired, and non-owned vehicles.
- .4 Workers' Compensation insurance (Coverage A) as prescribed by the laws of the State of California. The Policy shall include Employers' Liability (Coverage B) including Occupational Disease with limits of not less than \$1,000,000 per person per accident and shall provide a Borrowed Servant/Alternate Employer Endorsement.
- **10.1.2 Notice of Cancellation.** Each policy of insurance shall: (1) be in a form, and with insurers, satisfactory to County; (2) incorporate such endorsements as County may reasonably request; and (3) provide for thirty (30) Days' advance notice to County of non-renewal, material change, cancellation, or potential exhaustion of aggregate limits.
- 10.1.3 Additional Insureds. Architect shall have the following named as Additional Insureds by means of endorsement to its General Liability, Excess (or Umbrella) Liability, and Motor Vehicle Liability policies: (1) the Indemnitees; (2) the persons or entities listed in the Additional Insureds List Exhibit "L" attached hereto; and (3) all subsidiary companies, corporations, entities, joint ventures, LLC's, or partnerships that are owned, managed or controlled by the entities listed in Clauses (1) or (2) of this Paragraph 10.1.3. Such coverage shall contain no special limitations on the scope of protection afforded to the Additional Insureds. The "Insured" clause covering Additional Insureds shall: (a) be no more restrictive than the coverage afforded by ISO 2010 11/85 edition; (b) state that the coverage provided to the Additional Insureds is primary and non-contributing with any other insurance available to the Additional Insureds; and (c) require a waiver of subrogation in favor of all Additional Insureds.
- **10.1.4 Self-Insured Retentions.** Policies of insurance for the coverages described in <u>Paragraph 10.1.1</u>, above, with the sole exception of professional liability insurance, shall not have self-insured retentions which exceed \$10,000 per occurrence. All deductibles and self-insured retentions on insurance required to be obtained by Architect under this Agreement shall be borne by Architect at its sole expense and without reimbursement by County.
- 10.1.5 Certificates of Insurance. Prior to the commencement of any services under this Agreement, and at any time thereafter upon County's request during the term of this Agreement, Architect shall provide County with written evidence of the required coverages in the form of certificates of insurance with the applicable endorsements (including, without limitation, an endorsement confirming coverage for the Additional Insureds) attached or copies of the policies. County reserves the right to require complete, certified copies of all required insurance policies at any time, including endorsements providing the coverages required by this Agreement.

- **10.1.6 Waiver of Subrogation**. For Commercial General Liability and Workers' Compensation insurance, the insurer shall agree to waive all rights of subrogation against the Additional Insureds for Losses arising from activities and operations of an insured in the performance of services under this Agreement.
- **10.1.7 Lapse in Coverage.** If Architect or any Subconsultant, for any reason, fails to maintain any insurance coverage which is required pursuant to this Agreement, the same shall be deemed a material breach of this Agreement. County, at its sole option, may thereupon terminate this Agreement and obtain damages from Architect resulting from said breach. Alternatively, County may purchase such coverage (but has no obligation to do so) and, without further notice to Architect, may deduct from sums due to Architect any premium costs advanced by County for such insurance.
- **10.1.8 Subconsultants.** Except as otherwise stated in <u>Subconsultant Insurance Requirements Exhibit</u> "P" attached hereto, Subconsultants shall be required to maintain insurance on the same terms and with the same coverages as required of Architect under this Agreement.

ARTICLE 11 DISPUTE RESOLUTION

11.1 RESOLUTION OF DISPUTES

Disputes between County and Architect shall be resolved by way of an action filed in the Superior Court of the State of California, in and for the County of Riverside.

11.2 GOOD FAITH DETERMINATIONS

Wherever in this Agreement it is provided that the County may or shall make a determination or decision in the exercise of good faith (including, without limitation, provisions in this Agreement calling for a Good Faith Determination), any such determination or decision that the person exercising such right on behalf of County believes in good faith to be a proper exercise of County's rights and to have a reasonable basis in fact, whether or not such determination is in fact proper, reasonable or correct or adjudged to be so, shall be complied with by Architect without Delay to Architect's performance under this Agreement. However, unless this Agreement expressly provides otherwise, neither such good faith determination or decision nor Architect's compliance therewith shall be interpreted as precluding the Architect from exercising its rights to seek adjudication of its rights in the manner permitted by this Agreement or Applicable Laws.

11.3 ATTORNEY'S FEES

If any legal action, arbitration, or other legal proceeding is brought in connection with, or related to, the interpretation, performance, or enforcement of this Agreement, including, but not limited to, an action to rescind this Agreement, the prevailing party therein shall be entitled to recover from the other party the prevailing party's actual costs, expenses, and attorneys' fees at arbitration, mediation, trial, and on appeal, including, without limitation, a sum for time expended by in-house attorneys and paralegals. The determination of the "prevailing party" shall be based upon the party who prevails upon the matters actually litigated or arbitrated and shall not be determined solely based on which party receives a net monetary recovery.

ARTICLE 12 ROYALTIES, PATENTS, COPYRIGHTS AND TRADE SECRETS

12.1 ROYALTIES

Architect shall pay all royalties and license fees in connection with its performance of this Agreement. Compensation for such royalties and fees is included in Architect's Basic Services Compensation and shall not be separately reimbursed.

12.2 **INFRINGEMENT**

Architect shall not infringe any United States patent, copyright, trade secret, or other proprietary right for or in any work of authorship, material, product, or any other form of intellectual property, or any part thereof (including, without limitation, software, hardware, service, design or equipment), used or furnished in connection with this Agreement.

12.3 NOTICE BY ARCHITECT

In those instances where Architect has reason to believe that a particular design, process, or product of one or more manufacturers that Architect is directed to use by County would infringe upon any of the rights listed in <u>Section 12.2</u>, above, Architect shall immediately notify County of its belief and the reasons therefor in writing.

ARTICLE 13 MISCELLANEOUS

13.1 GOVERNING LAW AND VENUE

This Agreement shall, without regard to the law of conflicts of laws that may otherwise call for application of the laws of a different jurisdiction, be governed by the laws of the State of California. The Superior Court for the County of Riverside shall have exclusive jurisdiction over any litigation arising out of or relating to this Agreement.

13.2 HAZARDOUS SUBSTANCES

13.2.1 Introduction by Architect. Architect and its Subconsultants shall not cause or knowingly permit. or include in its Design Documents any provision allowing for, any Hazardous Substances to be deposited, stored, disposed, placed, generated, manufactured, buried, refined, transported, treated, discharged, handled, or located on the Site or in Existing Improvements, except as may be specifically authorized in writing by County; provided, however, that Hazardous Substances may be specified for temporary use or storage where reasonably required for, and in quantities appropriate to, the performance of the Work and where the use and storage of such Hazardous Substances is permitted by, and specified to be performed in conformity with, Applicable Laws. Should Architect or a Subconsultant violate the foregoing obligation, Architect shall at its own expense and without limitation to County's other rights or remedies for default immediately: (1) inform County in writing of such event; (2) advise County with respect to any release reporting or notification requirement that may apply as a result of such event; (3) assist County in complying with any such reporting or notification requirement as determined by County; and (4) perform any investigation, remediation, removal, or other response that is necessary or desirable in order to abate or clean up the condition resulting from such event, to the full satisfaction of County and any applicable Governmental Authority. Such Hazardous Substances shall be removed and properly disposed of as soon as they can be accepted at an appropriate disposal facility, and in no event later than sixty (60) Days after such waste is generated, unless a longer time is approved by County.

13.2.2 Existing Hazardous Substances. Architect recognizes that Hazardous Substances may exist at or beneath the ground at the Site and that certain waste materials, such as, but not limited to, drill cuttings and drilling fluids, must be handled as if contaminated until a determination as to whether they are Hazardous Substances is made. If the Architect's Basic Services do not include the investigation or assessment of environmental conditions or Hazardous Substances, then in the event Architect or its Subconsultants encounter materials existing or otherwise present at the Site that are reasonably believed to be Hazardous Substances that have not been rendered harmless, Architect and/or Subconsultant shall report the condition to County in writing and County shall be solely responsible for arranging for and paying the costs lawfully to transport, store, treat, recycle, dispose of, or otherwise handle the Hazardous Substances present at the Site. If the Architect's Basic Services include the investigation or assessment of environmental conditions or Hazardous Substances, then Architect shall: (1) promptly make a determination whether the materials encountered are Hazardous Substances; (2) promptly advise County of the options and costs for handling, storing and disposing of such materials (whether they are Hazardous Substances or not); (3) appropriately handle, contain and label such materials as are Hazardous Substances in accordance with Applicable Laws; (4) promptly inform County that such handling, containerization and labeling have been performed; and (5) leave the containers on Site in an appropriate designated location for lawful storage and disposal by County. County shall be solely responsible for arranging for and paying the costs to lawfully transport, store, treat, recycle, dispose of or otherwise handle Hazardous Substances generated by Architect's proper performance of its professional services. Should the proper and lawful transportation and disposal of any such materials be required, Architect's responsibilities shall be limited to preparing manifests or related documents for execution by County. In this regard, County shall sign all manifests and bills of lading, and approve similar documents, including subcontracts for disposal activities, that identify County as the generator/owner of any hazardous or contaminated material that is removed from the Site. County shall be solely responsible for notifying all appropriate federal, state, local or other governmental agencies of the existence of any Hazardous Substances on or about the Site or discovered during performance of this Agreement; no such notice shall be given by Architect without prior discussion and approval by County.

13.3 NO WAIVER

A waiver, by either party to this Agreement, of any breach of any term, covenant, or condition contained herein shall not be deemed to be a waiver of any subsequent breach of the same or any other term, covenant, or condition contained herein, whether of the same or a different character. County's approval, acceptance or use of, or payment for, all or any part of Architect's services shall not in any way alter Architect's obligations, or waive any of County's rights, under this Agreement.

13.4 NO THIRD-PARTY RIGHTS

Nothing contained in this Agreement is intended to make any person or entity who is not a signatory to this Agreement a third-party beneficiary of any right or obligation created by this Agreement or by operation of Applicable Laws.

13.5 **EXTENT OF AGREEMENT**

This Agreement represents the entire Agreement between County and Architect for the furnishing of services to the Project, and supersedes all prior negotiations, representations or agreements, either written or oral, and may be amended only by written instrument signed by both County and Architect.

13.6 SUCCESSORS AND ASSIGNS

This Agreement shall be binding upon County and Architect and their respective successors and assigns. Neither the performance of this Agreement nor any part thereof, nor any monies due or to become due hereunder, nor any claim hereunder, may be assigned by Architect without the prior written consent and

approval of County, which may be granted or withheld in County's sole and absolute discretion. This Agreement and all of County's rights in and to the Design Documents may be assigned by County upon written notice to Architect. County shall have no liability or responsibility to Architect for payment for any services performed or cost incurred after the date of such assignment and notice thereof by County to Architect.

13.7 **CONFIDENTIALITY**

The Architect acknowledges that, in the course of the Architect's employment and performance under this Agreement, the County may make available to the Architect, and the Architect may utilize and may participate in the creation of, proprietary and confidential information, including, without limitation, plans, specifications, projected and actual budgets, construction and development schedules, operating procedures, pricing data, transaction terms, Site-related information, studies (including survey, soil, environmental, structural, topographic and seismic) and other Project information (hereinafter collectively, "Proprietary Information"). The Architect agrees on behalf of itself and its employees, officers, board members and Subconsultants that all Proprietary Information shall be kept strictly confidential. Such confidentiality obligation includes, without limitation, the following covenants by Architect: Proprietary Information shall not be disclosed, either verbally or in writing, to a person or entity that is not related to the Project; (2) the Proprietary Information shall not be disclosed to any person or entity related to the Project other than County unless such disclosure is essential to the Architect's performance of this Agreement or to the performance by a Project-related person or entity of its Project-related work, services or obligations; (3) Architect shall not publicly reveal any Proprietary Information unless such disclosure is essential to the Architect's performance of this Agreement or to the performance by a Project-related person or entity of its Project-related work, services or obligations; (4) subject to the provisions of Paragraph 1.7.7, above, Architect shall return all Proprietary Information (including all copies made thereof) to County upon request and in any event within sixty (60) Days after termination or full performance of this Agreement: (5) Architect shall not be deemed the author of any of the Proprietary Information and retains no Intellectual Property Rights in the Proprietary Information; (6) to the extent the Architect provides any Proprietary Information to a Subconsultant, the Architect shall be responsible for obtaining and enforcing a written agreement from each such Subconsultant pursuant to which such Subconsultant agrees to be bound by the terms of this Section 13.7; and (7) in the event that the Architect or any Subconsultant is required, or becomes legally compelled, to disclose any of the Proprietary Information or take any other action prohibited hereby, the Architect will provide County with prompt written notice so that the County may seek a protective order or other appropriate remedy and/or waive in writing compliance with the provisions of this Section 13.7. County shall have full recourse under Applicable Laws in enforcing this Section 13.7, including without limitation the right to seek specific performance and injunctive relief and to recover all damages resulting from a violation hereof. Architect shall instruct all of its employees working on the Project of the foregoing confidentiality obligation.

13.8 INDEPENDENT CONTRACTOR

Architect is and shall at all times remain, as to County, a wholly independent contractor, both in respect to its design and construction administration obligations, as well as all other acts or omissions that occur in connection with its performance of this Agreement. Neither County nor any of its agents shall have control over the conduct of Architect or any of Architect's officers, agents or employees, except as otherwise herein set forth. Architect shall not, at any time, or in any manner, represent that it or any of its agents or employees are in any manner agents or employees of County.

13.9 ARCHITECT'S REPRESENTATIONS

Without limitation to any other covenants, agreements, or representations contained in this Agreement, Architect warrants and represents that: (1) it is financially solvent, able to pay its debts as they mature and possessed of sufficient working capital to complete the services and perform the obligations required by

this Agreement; (2) it is authorized to do business in the State of California; (3) it is duly licensed in accordance with Applicable Laws to enter into this Agreement for performance of the services to be provided by this Agreement; and (4) all services required to be performed by this Agreement or performed under this Agreement shall be performed by persons duly licensed in accordance with Applicable Laws to perform such services. Architect shall require that the foregoing warranties be provided, in writing, by each of its Subconsultants as part of its contracts with its Subconsultant.

13.10 SURVIVAL

The provisions of this Agreement which by their nature survive, or involve a right that is to be or may be exercised by or afforded to a party, or an act or obligation that is to be assumed or performed by a party, after the point in time that full performance or termination of this Agreement has occurred, including, without limitation, all obligations of indemnification, insurance, audit, dispute resolution, confidentiality, ownership of documents and records retention, shall remain in full force and effect after full performance or termination of this Agreement.

13.11 **SEVERABILITY**

In the event a provision of this Agreement, or portion thereof, is held to be invalid, illegal, or unenforceable, the validity, legality, and enforceability of the remaining provisions or portions shall not be affected, and such remaining provisions or portions shall be enforceable to the fullest extent allowable by Applicable Laws in order to give maximum legal force and effect to those provisions or portions that are not invalid, illegal or unenforceable.

13.12 INTERPRETATION

Architect and County acknowledge that the terms of this Agreement have been mutually negotiated and, accordingly, shall not be interpreted against either County or Architect on the basis that either party was solely responsible for or in control of the drafting of this Agreement.

13.13 ADVERTISING

Architect may not use County's name or refer to County or the Project, directly or indirectly, in any promotional materials, advertisement, news release or release to any professional or trade publication without County's prior written approval, which may be granted or withheld in its sole and absolute discretion.

13.14 ELECTRONIC DOCUMENTS

In the event of any conflict between a document contained in an electronic file and the hard copy of such document maintained in the files of County or Architect, the hard copy shall control.

13.15 EXECUTION

Execution by means of signature of a party hereto on a facsimile copy or electronically transmitted copy shall be binding to the same extent as execution of an original.

13.16 COUNTERPARTS; DIGITAL SIGNATURES:

This Agreement may be executed in any number of counterparts, each of which will be an original, but all of which together will constitute one instrument. Each party to this Agreement agrees to the use of electronic signatures, such as digital signatures that meet the requirements of the California Uniform Electronic Transactions Act (("CUETA") Cal. Civ. Code §§ 1633.1 to 1633.17), for executing this Agreement. The parties further agree that the electronic signatures of the parties included in this

Agreement are intended to authenticate this writing and to have the same force and effect as manual signatures. Electronic signature means an electronic sound, symbol, or process attached to or logically associated with an electronic record and executed or adopted by a person with the intent to sign the electronic record pursuant to the CUETA as amended from time to time. The CUETA authorizes use of an electronic signature for transactions and contracts among parties in California, including a government agency. Digital signature means an electronic identifier, created by computer, intended by the party using it to have the same force and effect as the use of a manual signature, and shall be reasonably relied upon by the parties. For purposes of this section, a digital signature is a type of "electronic signature" as defined in subdivision (i) of Section 1633.2 of the Civil Code.

13.17 TITLES FOR CONVENIENCE

The table of contents and the headings of articles and paragraphs are for convenience only and shall not modify rights and obligations created by this Agreement.

13.18 NONDISCRIMINATION

Architect shall comply, and cause its Subconsultants, of every Tier, to comply, with all requirements of Applicable Laws pertaining to equal opportunity employment and nondiscrimination, including, without limitation, those requirements prohibiting discrimination against or segregation of any person or group of persons on account of age, ancestry or national origin, color, creed, disability, gender, marital status, race, religion or sexual orientation, nor shall Architect permit any such practice prohibited by such requirements to take place in connection with the selection, location or number of consultants or vendors employed. Architect shall include the provisions of this Section 13.18 in all contracts entered into with Subconsultants for performance of services provided for under this Agreement.

13.19 SERVICES PERFORMED BY ARCHITECT OR AFFILIATES

The Architect shall not enter into any subcontract, contract, agreement, purchase order, or other arrangement ("Arrangement") for the furnishing of any portion of the services provided for in this Agreement with any party or entity if such party or entity is an Affiliated Entity (as defined below), unless such Arrangement has been approved in writing by the County after full disclosure in writing by the Architect to the County of such affiliation or relationship and all details relating to the proposed Arrangement. The term "Affiliated Entity" as used in this Section 13.19 means any entity related to or affiliated with the Architect or with respect to which the Architect has direct or indirect ownership or control, including, without limitation: (1) any entity owned in whole or part by the Architect; (2) if the Architect is a corporation, any holder of more than 10% of the issued and outstanding shares of the Architect; (3) if Architect is not a corporation, any holder of an ownership interest in Architect; or (4) any entity in which any officer, director, employee, partner, or shareholder (or member of the family of any of the foregoing persons) of the Architect, or any entity owned by the Architect, has a direct or indirect interest which interest includes, but is not limited to, that of a partner, employee, agent, or shareholder.

13.20 REBATES, KICKBACKS

Architect represents and warrants that it has neither paid or agreed to pay, nor will it pay, any sums or any other consideration to any member of the Board of Supervisors or any other director, officer, employee, agent or other representative of County in connection with this Agreement or any services hereunder, nor has any such payment or agreement for payment been requested or solicited by any such member, director, officer, employee, agent or representative. Architect hereby acknowledges that it understands that this representation and warrant constitute a material inducement upon which County is relying in entering into and performing this Agreement.

ARTICLE 14 NOTICES

14.1 DELIVERY AND ADDRESSES

- **14.1.1 Delivery.** Any notice that is required by this Agreement shall be given as provided hereinbelow. Electronic (i.e., e-mail) notice shall not be sufficient. All notices, demands, or requests to be given under this Agreement shall be given in writing and shall be conclusively deemed received as follows:
 - on the date delivered if delivered personally;
- (2) on the third (3rd) business day after the deposit thereof in the United States mail, postage prepaid, and addressed as hereinafter provided;
- on the date received if sent by facsimile transmission or overnight mail (such as, but not limited to, UPS, Fed Ex, or other similarly reputable private of public express carriers); and
 - (4) on the date it is accepted or rejected if sent by certified mail.
- **14.1.2** Addresses. All notices, demands or requests required by this Agreement shall be addressed to the parties as follows:

To County at:

Riverside County Facilities Management Project Management Office 3450 14th Street, 2nd Floor Riverside, CA 92501 Attention: Rebecca McCray, Deputy Director

With additional copies to: Riverside County Facilities Management Project Management Office 44-199 Monroe Street, Suite B Indio, CA 92201 Attention: Anna Rodriguez, Project Manager

To Architect at:

Prest Vuksic Greenwood Architects, Inc. 44530 San Pablo Avenue, Suite 200 Palm Desert, CA 92260 Attention: John Greenwood

With additional copies to:

N/A	
Attention:	

14.2 CHANGE OF ADDRESS

In event of any change of address, the moving party is obligated to notify the other party of the change of address in writing. Each party may amend, supplement and update the notice list to add, delete or replace any listed individuals by notice to the other party in writing.

ARTICLE 15 **EXHIBITS**

The following exhibits are attached hereto and incorporated in this Agreement by this reference as part of the terms of this Agreement:

Exhibit "A"	-	Property Description
Exhibit "B"	-	Description of Basic, Additional and Excluded Services
Exhibit "C"	-	Initial Program
Exhibit "D"	-	Master Project Schedule
Exhibit "E"	-	Key Personnel List
Exhibit "F"	-	Subconsultants and County Consultants List
Exhibit "G"	-	Reimbursable Expenses Schedule
Exhibit "H"	-	Hourly Rates Schedule
Exhibit "I"	-	Payment Schedule
Exhibit "J"	-	Reference Documents List
Exhibit "K"	-	General Conditions of the Standard Form Construction Contract
		Between County and Contractor
Exhibit "L"	-	Additional Insureds List
Exhibit "M"	-	Release Forms
Exhibit "N"	-	Invoice for Payment Form
Exhibit "O"	-	Construction Document Deliverables
Exhibit "P"	-	Subconsultant Insurance Requirements
Exhibit "Q"	-	LEED's Design Credits (N/A)

In the event of a conflict between the provisions of a conditions of the Agreement, the latter shall control.	ny of the above-listed exhibits and the terms and
day of, 2023 [to be filled in by Clerk of the I	
"COUNTY"	"ARCHITECT"
COUNTY OF RIVERSIDE By: Chairman, Board of Supervisors KEVIN JEFFRIES	Prest Vuksic Greenwood Architects, Inc. (sign on line above) By: John Greenwood (type name)
	Title: Secretary
	The following information must be provided concerning the Architect:
ATTEST:	State whether Architect is corporation, individual, partnership, joint venture or other:
KIMBERLY RECTOR Clerk of the Board	Corporation
By: Mulk Mill	If "other", enter legal form of business:
(SEAL)	Enter address: 44530 San Pablo Avenue Suite 200 Palm Desert, CA 92260
	Telephone: (760) 779-5393 Facsimile: N/A E-mail: johng@pvgach.com
APPROVED AS TO FORM: MINH C. TRAN County Cou	Employer State Tax ID #: 330754774 License #: C-38012
By: XIII	If Architect is not an individual or corporation, list names
Kristine Bell-Valdez	4 representatives who have authority to bind Architect:N/A
Deputy County Counsel	

52 of 53

	If Architect is a corporation, state:
	Name of President: <u>John Vuksic</u>
	Name of Secretary: _John Greenwood
	State of Incorporation: California
	State of incorporation. Camornia

EXHIBIT "A"

PROPERTY DESCRIPTION

Address: 47915 Oasis Street, Indio, CA 92201 APN: 614-150-033 S 1/2, SEC. 26, T.5S., R.7E POR. City of Indio

Edition: November 29, 2021 INITIALS JG

EXHIBIT "B"

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

2.1.1	PR	ROFESSIONAL SERVICES		
	.1	Excluded Services. The following constitute Excluded Services:		
are r	.2 marked	Basic Services, Additional Services. Those of the following professional services, which "Yes," shall be provided by Architect as part of Basic Services. Those of the following services, which are marked "No" shall be deemed to be Additional Services.		
Yes	<u>No</u>			
X		Architectural		
0	x	Civil		
	X	Structural		
X	_	Mechanical		
x	_	Plumbing		
x		Electrical		
x	_	Energy calculations		
.	x	California Environmental Quality Act (CEQA) compliance		
)	Х	National Environmental Policy Act (NEPA) compliance (of applicable)		
		Page 1 of 18		
Edition	i: Nove	ember 29, 2021 INITIALS JG		

EXHIBIT "B"

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

	x	National Pollution Discharge Elimination System compliance (NPDES) - SWPPP
_	x	National Pollution Discharge Elimination System compliance (NPDES) - WQMP
	x	National Pollution Discharge Elimination System compliance (NPDES) - FPPP
_	x	Landscape (on-Site plantings)
0	X	Landscape (off-Site plantings)
	X	Utility coordination
_	x	Fire/life safety If required, design will be done by others and coordinated by the Architect.
0	x	Acoustic
0	x	Elevator, Escalator
	X	Parking, Striping
0	x	Security
	x	Communications
	x	Signage/graphics
Editio	n: Nove	Page 2 of 18
COITIC	INOVE	mber 29, 2021 INITIALS JG

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

	x	Waterproofing
	x	Lighting design
	x	Interior design (furniture, furnishings and artwork)
	x	Space planning for tenant or rental spaces
	x	Future (i.e., post-construction) facilities, systems and equipment
0	x	Quantity surveys and inventories
0	x	Appraisal of existing facilities
0	x	Analyses of owning and operating costs
D	x	Financial feasibility or other special financial studies
0	x	On-Site traffic flow and traffic mitigation
0	x	Measured drawings of Site
-	x	Measured drawings of Existing Improvements
		Page 3 of 18

Edition: November 29, 2021

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

2.1.2 PLANNING/PROGRAMMING PHASE

The Planning/Programming Phase commences upon execution of the Agreement and terminates

upon approval by County of the Final Program for the Project and receipt by County of the other Design Documents required by this <u>Paragraph 2.1.2</u> to be prepared by Architect as part of the Planning/Programming Phase.
Yes No
□ X .1 Alternate Design Approaches. Review and recommend alternative approaches to the design of the Project, including, without limitation, development of conceptual alternatives based on different approaches to building location, massing and height.
☐ X .2 Alternative Structural Approaches. Review and recommend alternative structural approaches for each of the different approaches recommended by the Architect for building location massing and height, including, without limitation, an analysis of how adjacent structures affect the structural design of the new structures.
□ X .3 Alternative Contracting Approaches. Recommend the method of contracting best suited to the Program and Master Project Schedule.
□ X .4 Version Control Plan. In addition to Architect's obligation for version control, if any, assumed under Subparagraph 2.1.7.17, Subpart (1) of this Exhibit "B", prepare for County's review and approval a comprehensive written plan setting forth systematic procedures for periodic and recurring updating of the Drawings and Specifications to systematically, comprehensively, efficiently and clearly incorporate and consolidate changes, additions and deletions to the Project.
Under the Agreement, prepare for County's review and approval a written comprehensive plan setting form systematic procedures during all Phases for periodic and recurring review of the Design Documents by employees of Architect who, acting independent of the Architect's employees involved in the preparation of the Design Documents, will check the Drawings and Specifications for (1) sufficiency, completeness, accuracy, constructability and coordination, and (2) conformance with the Design Coordination Standards and who will personally certify with each submission of Design Documents during the Schematic, Design Development and Final Construction Documents Phases that the Design Documents being submitted have been checked for compliance with the Design Coordination Standards.
Electronic Document Management Manual. Prepare a written management manual, developed in conjunction with County and incorporating software systems compatible with County's software, setting forth: (1) a systematic procedure for standardizing (utilizing AutoCAD for Drawings and MS Word for Specifications and other written documents) the requirements to be followed by other Project Team members for electronic preparation and delivery of drawings, plans, specifications and other documentation, including, but not limited to, Record Documents; and (2) procedures for implementation of County's electronic-based program management system, if any, for the Project.
X .7 Existing Conditions. Based on Architect's inspection of the visible conditions of the Site and Existing Improvements, Architect's functional testing of the Existing Improvements' operating systems (including mechanical, electrical, plumbing, fire protection and low voltage systems) and Architect's eview of documentation that is available from County and current public records (including, without mitation, soils and geotechnical reports, environmental and hazardous materials surveys, as-built ocumentation, and public records pertaining to underground utilities), prepare a written report, data base

Page 4 of 18

Edition: November 29, 2021

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

and topographic map and survey setting forth: (1) a consolidated and coordinated statement of all available information concerning the conditions at the Site and in Existing Improvements; (2) identification of any conflicts between such information or between such information and the visible conditions of the Site or Existing Improvements; (3) confirmation of whether the Existing Improvements comply with Applicable Laws and the operating systems (including mechanical, electric, plumbing and low voltage systems) in Existing Improvements have sufficient capacity to support the proposed new designs; and (4) recommendations by the Architect for further investigation, exploration, inspection or testing by County or County's Consultants or for remediation to, or upgrading of, Existing Improvements found to be obsolete, in violation of Applicable Laws, or inadequate to support the proposed new designs.

8. Community Outreach. (1) Participate, as and when requested, in the County's program for outreach to neighborhood residents and their representatives to assess, evaluate and distribute information with respect to matters of community concern (including, without limitation, traffic, parking, view corridors, pedestrian access, public transportation access and paths of travel) and make recommendations to County for revisions to the Program in order to satisfy the expressed needs of the community. Prepare presentation materials (consisting of hand outs, summaries, charts and other appropriate paper and electronic visual aids) for _____ (___) presentations to community groups designed to develop community consensus on a plan of development that, to the greatest extent possible, satisfies the requirements of the County's updated Program for the Project, as well as the expressed needs of the community. X Initial Program Review. Make recommendations for revisions to the Initial Program, including advice on any impact that acceptance of such revisions by County would have upon the Master Project Schedule, in order to meet the County's most-current, stated goals and objectives and the requirements of Governmental Authorities for the Project. Final Program. In consultation with County and County's Consultants, prepare for County's review and approval revisions to the Initial Program leading to the development of a Final Program, which Final Program when complete shall consist of a comprehensive and complete written statement of the County's Project criteria, objectives and requirements, with reference to, without limitation, the following: (1) planning, space utilization, space flexibility and expandability, function, systems operations and special equipment; (2) typical room layout; (3) blocking and stacking diagrams depicting departmental relationships; (4) staff and visitor ingress, egress and circulation; (57) evaluations and tabulations of key functional spaces by type and location; (6) area calculations, based on net square feet. department gross square feet, building gross square feet and project gross square feet; and (7) a narrative explaining the reasons for any changes or deviations from the Initial Program. .11 Area Calculations. Provide square footage area calculations, based upon the Final Program approved by County, for use by County in planning, demising and allocating spaces. .12 Perspective Drawings. Provide ____ (__) perspective drawings of the Project, as conceptualized in the Final Program approved by the County, representing the appearance of the Project from vantage points selected by County. Massing Model. Provide a study model showing in three dimensions the shape .13 and mass of the building(s) on the Site. X .14 Zoning. Provide materials, such as, but not limited to, sketches and exhibits. Page 5 of 18

INITIALS JG

Edition: November 29, 2021

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

	required to secure applicable zoning and architectural committee approvals and variances necessary for the Project.
	☐ X .15 CEQA Compliance. Prepare, for review and approval by County, California Environmental Quality Act compliance documentation and all pertinent studies required therein to satisfy the requirements of the California Environmental Quality Act.
	2.1.3 SCHEMATIC DESIGN PHASE
	The Schematic Phase commences upon the end of the Planning/Programming Phase and ends upon approval by County of the completed Schematic Design Documents.
	Yes No
	.1 Schematic Design Documents. Prepare, for approval by County, (_) paper copy sets and an electronic version of the following Schematic Design Documents for the Project, consistent with the Final Program and the Architect's Project Schedule:
	□ X (1) Site Plans. Prepare Site plans depicting the Architect's concept for the completed Project, including: (a) preliminary circulation schemes for persons and vehicles and parking; (b) locations of fire hydrants and fire lanes; (c) areas of hardscape and softscape (including, but not limited to, plantings, service drives, loading docks, walkways, ramps, steps, stairs, retaining walls, planters, furnishings, wayfinding elements, plantings, light fixtures and other amenities); (d) Site elevation variations which affect circulation paths and drainage; (e) locations and dimensions of structures (new and existing), property lines, easements, setback requirements; with dimensioning (including, without limitation, at least four (4) dimensions for each side of the building (from fixed and verified landmarks) sufficient to demonstrate that all proposed buildings fit within the property lines, public rights of way, easements and the design space; (f) sections indicating massing and relationships to Existing Improvements and Site conditions; (g) a preliminary Site clearing and demolition plan depicting Existing Improvements and landscaping elements to be removed and to remain; and (h) a depiction of the preliminary routing of utility lines to proposed points of connection.
	□ X (2) Building Grid. Prepare a dimensioned building grid system for use by other Project Team members that provides a common reference for the coordinated documentation of the Work by other Project Team members and that includes (in the case of new construction) all walls and doors and (in the case of renovation) all Existing Improvements to remain.
V (C C a fe	(3) Floor Plans. Prepare preliminary floor plans depicting: (a) layout of walls, partitions, rooms (including room numbers and names), departmental boundaries, core elements (such as, but not limited to, lobbies, corridors, stairs, elevators, toilets, shafts, and chases), columns and other major structural elements, and sufficient dimensioning to demonstrate that all layouts fit within the design space; (b) if the Architect is providing design services related to furniture, furnishings and artwork, a proposed placement of furniture and furnishings in repetitive and typical conditions; and (c) basic criteria or major finish materials, along with a summary of comparative advantages based on cost, durability, ease of maintenance and appearance.
d	X (4) Façade Studies. Prepare alternative façade studies, consisting of different proposed approaches to the design of the building exterior and details for each alternative describing proposed colors, materials and textures.
	X (5) Roof Plans. Prepare preliminary roof plans, shown against a background
,	Page 6 of 18
I	Edition: November 29, 2021 INITIALS JG

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

of structural gridlines and building dimensions, depicting the major elements of the roofed areas, including, without limitation, major penetrations, penthouses and identification of roof mounted equipment.

	, see a see
	D X (6) MEP Studies. Prepare: (a) an outline of alternative mechanical, electrical plumbing, fire protection, and low voltage systems, preliminary sizing of major mechanical, electrical plumbing and fire protection equipment and equipment rooms; (b) single line diagrams of mechanical electrical, plumbing and fire suppression distributions; and (c) identification of special mechanical, electrical plumbing and fire protection requirements (such as, but not limited to, exhaust requirements, emergency power and fire suppression).
	□ X (7) Life Cycle Costing. Prepare an analysis of life cycle costs (including, without limitation, capital costs, operating costs, maintenance costs and simple payback analyses) and comparative performance characteristics of each alternative identified in the County's systems selections.
	X (8) Preliminary Structural Analysis. Prepare (a) a preliminary design of the major structural elements for the Project that is coordinated with the requirements of Governmental Authorities and the recommendations of the County's Consultants in the field of geotechnical engineering and, accompanied by recommendations, if any, for additional studies, reports, exploration or testing of the Site or Existing Improvements; (b) foundation and framing (at each level of the building) design Plans, sufficient to permit preliminary cost estimating; and (c) a preliminary report of the dynamic performance of the structural system suitable for the proposed occupancy.
	□ X (9) Elevator Recommendations. Prepare preliminary recommendations on the number, types, sizes and operational characteristics of elevators.
	☐ X (10) Outline Specification. Prepare an outline Specification covering all technical divisions of the Construction Specifications Institute format, or using such other format acceptable to County.
	□ X (11) Building Code Outline. Prepare documentation (plans and/or narratives) addressing building codes and similar Applicable Laws which apply to the Project, accompanied by a delineation of key design criteria (such as, but not limited to, exits, travel distances, rated walls and corridors, building occupancy, construction type, and fire zones) required to meet such Applicable Laws.
	X (12) Area Calculations. Prepare (a) square footage area calculations, based upon the Schematic Design Documents approved by County, for use by County in planning, demising and allocating spaces and (b) a comparison of areas in the Schematic Phase Documents to the areas set forth in the Final Program.
	X (13) Perspective Drawings. Prepare () perspective drawings of the Project, as conceptualized in the Schematic Design Documents approved by the County, representing the appearance of the Project from vantage points selected by County.
(X (14) Schematic Model. Prepare, and provide County with, a schematic model at a minimum scale of 1/16 th* = 1'0"), constructed of simple materials such as foam board and hand drawn or pasted cut-outs, depicting in three dimensions the appearance of the building façade as shown in the Schematic Design Documents approved by County.
C	X (15) Design Narrative. Prepare a "basis of design" narrative and documentation supporting the design criteria for the structural elements (including structural loading), nechanical, electrical, plumbing, lighting, low voltage systems and other specialized building systems.
	Page 7 of 18

Edition: November 29, 2021

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

.2 **Estimate of Construction Cost.** X (1) Prepare, with the level of due care of a design professional and not a contractor or professional cost estimator, a preliminary written Estimate of Construction Cost based upon the Schematic Design Documents approved by County, including recommendations to County of opportunities for incorporation of design alternatives that will optimize quality and value consistent with the Fixed Limit, Final Program and Master Project Schedule. Review and comment upon the preliminary Estimate of Construction Cost prepared by Contractor or County's Consultant in cost estimating, including recommendations to the County of opportunities for incorporation of design alternatives that will optimize quality and value consistent with the Fixed Limit, Final Program and Master Project Schedule. .3 Community Outreach. Х (1) Participate, as and when requested, in the County's program for outreach to neighborhood residents and their representatives to assess, evaluate and distribute information with respect to matters of community concern (including, without limitation, traffic, parking, view corridors. pedestrian access, public transportation access and paths of travel) and make recommendations to County for revisions to the Program and Schematic Design Documents in order to satisfy the expressed needs of the community. X Prepare presentation materials (consisting of hand outs, summaries. (2) charts and other appropriate paper and electronic visual aids) for ____ (__) presentation(s) to community groups designed to develop community consensus on a plan of development that, to the greatest extent possible, satisfies the requirements of the County's Program for the Project, as well as the expressed needs of the community. Zoning. Prepare for the County materials, such as, but not limited to, sketches and exhibits, required to secure applicable zoning and architectural committee approvals and zoning variances necessary for the Project. X Project Directory. Prepare a Project directory stating complete contact information (name, address, telephone (office, home and cellular), fax, email) for each of the Project Team members. X Preliminary Project Specific Water Quality Management Plan. Prepare for review and approval by the Department of Facilities Management's Design and Construction Division a preliminary project specific Water Quality Management Plan (WQMP) in accordance with the requirements of the applicable and adopted municipal separate storm sewer system (MS4) permit (including, without limitation, any amendments, orders and new permits that the Regional Water Quality Control Boards has issued prior to, or issues after, execution of the Agreement by County and Architect, in connection with preparation of a project-specific WQMP), which permit(s) may include any of the following three (3) MS4

Order No. R8-2002-0011, NPDES No. CAS 61833 adopted by the Santa Ana Regional Water Quality Control Board on October 25, 2002 for the Santa Ana Region.

Order No. R7-2008-0001, NPDES No. CAS 617002 adopted by the Colorado River Basin Regional

Page 8 of 18

Edition: November 29, 2021

permits applicable within the County of Riverside:

INITIALS JG___

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

Water Quality Control Board on May 21, 2008 for the Whitewater River Region.

Order No. R9-2004-001, NPDES No. CAS 108766 adopted by the San Diego Regional Water Quality Control Board on July 14, 2004 for the Santa Margarita Region.

(The aforementioned permits describing the requirements for the WQMP can be provided by the Department of Facilities Management's Design and Construction Division). Architect will prepare the WQMP utilizing the approved template that can be found on Riverside County Flood Control's website at http://www.floodcontrol.co.riverside.ca.us/ or, upon request, such template can be provided by the Department of Facilities Management's Design and Construction Division. Architect shall include all the necessary exhibits, reports, and data required for the preparation of the WQMP. Architect will make all the necessary revisions that are requested by the County to be made to the preliminary WQMP.

County Drainage Area Management Plan (DAMP) and the applicable MS4 permits described in Subparagraph 2.1.3.6, above, Architect shall prepare a Municipal Facilities Pollution Prevention Plan (MFPPP, or 3P). The MFPPP shall be prepared utilizing the approved template found in Appendix J of the Riverside County DAMP. Upon request, the Department of Facilities Management's Design and Construction Division can provide the Architect with the MFPPP template. The MFPPP will include the approved maintenance and inspection requirements of the structural BMPs (e.g. Treatment Control, etc.) identified in the project-specific WQMP. During the preparation of the MFPPP, the Architect, the Department of Facilities Management, and the user department for the County will coordinate inspection, maintenance, and repair responsibility of the BMPs as identified in the MFPPP. A draft MFPPP document will be prepared concurrently with the preliminary project-specific WQMP and will be reviewed by the Department of Facilities Management's Design and Construction Division.

2.1.4 DESIGN DEVELOPMENT PHASE

The Design Development Phase commences upon the end of the Schematic Design Phase and ends upon approval by County of the completed Design Development Documents.

	enus u	роп арр	roval by	County	of the completed Design Development Documents.
	Yes	<u>No</u>			
	substar obtain o Costs fo or diffici	ntially co detailed, or the Pro ulties to	mplete; prelimin oject, an achievin	electron (2) fully ary Con d (b) Ard g compl	n Development Documents. Prepare, for approval by County, Two (2) nic draft of each and all of the Construction Documents, in a form that is: (1) dimensioned; and (3) sufficiently developed so as to enable (a) County to tractor and Subcontractor materials take-offs and estimates of Construction chitect and other Project Team members to identify any significant obstacles etion of the Project within the constraints of the Final Program, Fixed Limit, d Applicable Laws.
		X	.2	Estima	te of Construction Cost.
-	contract Design opportu	Develor nities for	oment D incorpo	ocumer ration of	Prepare, with the level of due care of a design professional and not a stimator, a detailed written Estimate of Construction Cost based upon the its approved by County, including recommendations to the County of design alternatives that will optimize quality and value consistent with the Master Project Schedule.
۰		X d by Co		(2) or Cour	Review and comment upon the detailed Estimate of Construction Cost nty's Consultant in cost estimating based upon the Design Development

Page 9 of 18

Documents approved by County, including recommendations to the County of opportunities for

Edition: November 29, 2021

INITIALS JG ____

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

	AND EXCLUDED CENTION
	incorporation of design alternatives that will optimize quality and value consistent with the Fixed Limit, Final Program and Master Project Schedule.
	□ X .3 Area Calculations. Prepare square footage area calculations, based upon the Design Development Documents approved by County, for use by County in planning, demising and allocating spaces.
	X .4 Initial Palette Presentation. Prepare and conduct an initial, formal presentation to County of Architect's recommendations and alternatives for finish materials and equipment, including acquiring and presenting physical samples of the materials for the exterior building envelope (including, without limitation, cladding, exterior finishes, glazing, trim, ornamental metals and roofing), interior (including, without limitation, finish materials for floors, walls, ceilings, doors, finish hardware, millwork, bathroom fixtures and lighting), wayfinding and signage, and (if the Architect is providing design services related to furniture, furnishings and artwork) furniture, furnishings and artwork.
	□ X .5 Circulation Plan. In consultation with County, prepare a plan for providing the entering, exiting and circulation, as appropriate to the planned uses of the Site during construction, of the County's staff, occupants, visitors and public during construction, including, without limitation, pedestrian flow (including, without limitation, access and paths of travel for persons with disabilities), vehicular flow, parking, delivery paths, fire lanes and fire hydrants.
	□ X .6 Access and Staging Plan. In consultation with County and its Contractor, prepare a plan showing access and exiting for construction equipment and crews and location for on-Site parking, storage and temporary construction facilities, fencing and temporary barricades.
	☐ X .7 Shoring Plans. Prepare Drawings delineating design and details for shoring, underpinning or tie backs.
	X .8 Post-Tension Design. Retain Subconsultant to develop post-tension plans, details, and calculations for the post-tension slab-on-grade foundations.
	X .9 Computer Generated Perspectives. Prepare and present to the County
	X .10 Renderings. Prepare () rendering(s) depicting, both in black and white and color, two dimensionally and pictorially, the elevation and perspective views of the Project from four (4) directions, professionally prepared by a Subconsultant experienced in preparation of design renderings.
	X .11 Presentation Model. Prepare () presentation model(s) professionally prepared by a Subconsultant experienced in display models, in color, and built to a scale of 1/8 th " = 1'0".
-	X .12 Project Directory. Prepare an updated Project directory.
2	2.1.5 FINAL CONSTRUCTION DOCUMENTS PHASE
F	The Final Construction Documents Phase commences upon the end of the Design Development Phase and ends upon approval by County of the Final Construction Documents.
Y	<u>'es No</u>

Page 10 of 18

Edition: November 29, 2021

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

	X
	Documents shall include one (1) One (1) sets of prints of all Drawings, One (1) sets of the Specifications, and one (1) electronic version of all Drawings and Specifications.
	□ X .2 Stair Plans. Prepare Drawings, enlarged to a scale of ¼ th"=1'0", coordinated with structural dimensions, delineating dimensions of risers, treads and landings and that comply with Applicable Laws governing exiting.
	□ X .3 Fixture and Equipment Schedules. Prepare schedules for fixtures and equipment, including, without limitation, bathroom fixtures and accessories.
C	☐ X .4 Tabulated Schedules. Prepare schedules for floor, ceiling and wall finishes, doors and windows and related hardware and openings and fixtures and equipment in tabular form by room number, location number and interior elevation.
- 1-	X .5 Dynamic Analyses. Prepare Specifications for dynamic testing and analysis of the following building elements:
е	X .6 Window Wall. Prepare Drawings and Specifications setting forth all of the essential prescriptive and performance criteria necessary for the Contractor to engineer and build an exterior window wall system.
	X .7 Emergency Water Storage System. Prepare a designed system of water storage or emergency purposes.
lir in	1 X .8 Fire Protection (Suppression and Detection) Systems. Prepare fully ngineered Drawings and Specifications delineating the fire suppression and detection systems that complies with the requirements of the County's insurance carriers and Applicable Laws, including, without mitation, a layout of the heads and the details of the interface with the fire alarm system and complete aformation on head end equipment (including, without limitation, annunciator panels, main and subordinate re alarm annunciator panels, risers, conduit runs, terminal devices and zoning).
ca	Elevator System. Prepare Drawings and Specifications delineating elevator esigns, including, without limitation, sizing of shafts, elevator pits and elevator machine rooms, an overall uilding section of the hoistway, power and lighting requirements for the equipment room and cabs, elevator ab finishes and accessories, and a system for programming operations under normal and emergency anditions (including, without limitation, the interfaces between elevator functions and the fire alarm and eccurity systems).
cri	X .10 Mock-Up Criteria. Prepare Drawings and Specifications delineating the design iteria for construction and testing of mock-ups for the following elements of the completed design:
D Pr	X .11 Energy Plan. Prepare a preliminary energy analysis demonstrating that the oject can meet the requirements of Title 24 of the California Code of Regulations, accompanied by a brief
	Page 11 of 18
E	dition: November 29, 2021 INITIALS JG

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

energy usage report identifying incorporation of any energy savings devices.

place clear	ement of rly indica	furniture tes whet	Furniture, Furnishings, Artwork. Prepare Drawings and Specification based materials, textures and colors for furniture, furnishings and artwork; and (2), furnishings and artwork on architectural floor plans with an identification key which there items will be new, refurbished or reused and cross references for relocated ion prior to and after relocation.
		.13	Estimate of Construction Cost.
to b	nitted witl e subm mmendat	h each o itted by ions to t	(1) Prepare, with the level of due care of a design professional and not a conal cost estimator, detailed written Estimates of Construction Cost based upon and the sets of progressive iterations of the Construction Documents that are required Architect to County pursuant to <u>Subparagraph 2.1.5.1</u> , above, including the County of opportunities for incorporation of design alternatives that will optimize distent with the Fixed Limit, Final Program and Master Project Schedule
of pro	ogressive ity pursu rtunities f	iteration ant to or incorp	(2) Review and comment upon the detailed written Estimates of Construction tractor or County's Consultant in cost estimating that are based upon each of the sets as of the Construction Documents that are required to be submitted by Architect to Subparagraph 2.1.5.1, above, including recommendations to the County of poration of design alternatives that will optimize quality and value consistent with the ram and Master Project Schedule.
□ parkir	X ng counts	.14 s.	Area Tabulations. Prepare tabulations of gross and assignable floor areas and
samp finishe for flo signag	les of the es, glazin ors, walls ge, and (materia g, trim, c s, ceiling (if the Ar	Final Palette Presentation. Prepare and conduct a final, formal presentation to of final finish materials and equipment, including acquiring and presenting physical als for the exterior building envelope (including, without limitation, cladding, exterior branamental metals and roofing), interior (including, without limitation, finish materials s, doors, finish hardware, millwork, bathroom fixtures and lighting), wayfinding and rehitect is providing design services related to furniture, furnishings and artwork) and artwork.
☐ of the Phase		.16 on Plan	Updated Circulation, Access and Staging Plans. Prepare an updated version and Access and Staging Plan initially submitted during the Design Development
	X	.17	Project Directory. Prepare an updated Project directory.
of Cal Quality The Ar (NOI) a to sam Quality activitie other t	ntion Plar ifornia G y Order 9 rchitect's and Notice to the y Control es (which ypes of s	n (SWPP) seneral F 9-08-DW enginee se of Terr County's Board or n may incode	Stormwater Pollution Prevention Plan. Prepare a Stormwater Pollution PP) and associated monitoring plan in accordance with the requirements of the State Permit for Stormwater Discharges Associated with Construction Activity (Water /Q) and subsequent amendments or orders for construction activities as applicable, ring Subconsultant responsible for such matters shall prepare the Notice of Intent mination (NOT) for the applicable Regional Water Quality Control Board and submit is project manager. County will mail all Notices to the applicable Regional Water the State Water Resource Control Board. In addition and prior to ground disturbing clude, but are not limited to, clearing, grubbing, weed abatement, trenching, and rbance including grading), such engineering Subconsultant shall ensure that the gement Practices are installed on-Site in accordance with the approved SWPPP
			Danie 40 of 40

Page 12 of 18

Edition: November 29, 2021

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

that was prepared for the Project. A copy of the approved SWPPP shall be retained on the Site during the duration of construction. A draft copy of the SWPPP shall be submitted by Architect for review and approval by the County's Department of Facilities Management's Design and Construction Division prior to submittal of the NOI. Any changes requested by the Department of Facilities Management's Design and Construction

Division will be implemented by Architect as revisions to the draft SWPPP and shall be re-submitted as a revised draft SWPPP. Upon final approval of the SWPPP, the Department of Facilities Management's Design and Construction Division will mail the NOI.
□ X .19 Final WQMP. Prepare for review and approval by County's Department of Facilities Management, a final WQMP, including any changes to the preliminary WQMP as County may deem necessary or appropriate and provide two hardcopies of the final WQMP approved by County, along with a digital submission.
□ X .20 Final MFPP. Prepare for review and approval by County's Department of Facilities Management's Design and Construction Division, a final MFPPP, including any changes to the preliminary MFPPP as County may deem necessary or appropriate and provide two hardcopies of the final MGPPP approved by County, along with a digital submission.
2.1.6 BIDDING PHASE
The Bidding Phase shall commence upon direction by County for issuance of the Construction Documents to bidders or proposed Contractors or Separate Contractors and ends upon execution of the contracts between County and Contractor and the Separate Contractors for performance of the Work of the entire Project.
Yes No
□ X .1 Prequalification. Recommend prequalification criteria, assist in preparation of the prequalification documents and participate in the evaluation of prequalification submittals by proposed Contractors and Separate Contractors.
□ X .2 Addenda. Prepare and obtain approval by Governmental Authorities of Addenda for issuance to the proposed Contractors and Separate Contractors.
X .3 Alternates. Prepare documentation of alternates, assist County in the ranking of alternates for bidding and assist County in the evaluation of portions of bids or price proposals relating to pricing of alternates.
X .4 Substitutions. Review and evaluate requests by bidders or proposers for substitutions of products or equipment and make recommendations to County for their approval or rejection.
N D .5 Pre-Construction Conferences. Attend, and arrange for necessary Subconsultants to attend, all pre-bid and pre-construction Site visits, tours and conferences conducted for proposed or selected Contractors, Separate Contractors or Subcontractors.
Clarifications. Prepare or secure from the appropriate Project Team member esponses to questions by bidders or proposers and promptly forward each question and response to County for distribution.
Schedule of Values. Review the list of the cost items prepared by the selected Contractor and Separate Contractors to be included in the Schedule of Values to assess whether it appears,
Page 13 of 18

Edition: November 29, 2021

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

based upon Architect's review as a design professional and not a contractor or professional estimator, to represent a fair, balanced and reasonable allocation of costs, overhead and profit. X Plans & Specifications. Coordinate the transmittal of all final and complete Construction Documents to the proposed Contractors and Separate Contractors to ensure that the Construction Documents used for bidding are the same as the Final Construction Documents approved by County and stamped and approved by applicable Governmental Authorities. If any differences exist, report them promptly to County. .9 Bid/Proposal Evaluations. Assist County in the review and evaluation of bids and negotiated proposals for the Work by proposed Contractors, Separate Contractors and Subcontractors. .10 Negotiation of Contracts. Assist County in the preparation and negotiation of the legal terms and conditions of the Construction Contract between County and the Contractor and the contracts between County and Separate Contractors; provided, however, that County shall remain solely responsibility for the legal enforceability and sufficiency of such terms and conditions and provided further, that nothing herein shall be interpreted as authorizing or requiring the Architect to furnish legal advice to County. Multiple and Sequential Bid Packages. Provide services for the implementation .11 of multiple bid packages and sequential bidding. X .12 Long Lead Items. Identify and recommend to the County a schedule for procurement of long lead time items and assist County in implementing such recommendations. X .13 Project Directory. Prepare an updated Project directory. **CONSTRUCTION PHASE** The Construction Phase commences, following execution of the contract between County and Contractor, upon commencement by Contractor of performance of the Work and ends Thirty-Five (35) Days after Final Completion of the entirety of the Work. Yes No General Obligation. Unless and except otherwise indicated in this Paragraph 2.1.7, provide all services required by the Agreement or the General Conditions for the administration of the construction by the Contractor and Separate Contractors. Meeting Minutes. Prepare and distribute preconstruction and construction .2 meeting minutes. Site Observations. Observe construction at the Site as it progresses, as and when appropriate to the stage and progress of the Work and the needs of the Project, but not less frequently than monthly, for the purpose of: (1) ascertaining in general that the character, scope, quality and detail of construction (including workmanship and materials) comply with the Contract Documents, County's directives, approved Submittals, Architect clarifications and requirements of Applicable Laws; (2) evaluating

the progress of the Work; (3) evaluating the suitability of the Project, or any portion designated by County, for use or occupancy; (4) investigating and responding to design and constructibility issues or questions of concern to any Project Team member, or as noted in any inspection reports furnished to Architect; (5)

Page 14 of 18 Edition: November 29, 2021

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

observing the overall quality of Contractor's performance; (6) reviewing specifically and in detail the visible condition of the construction of structural components, building systems and other crucial components of the Work; and (7) observing the performance of specified or directed tests significant to the acceptability of components of the Work. Such observations shall be separate from any inspections which may be provided by inspectors retained by County, Contractor, Separate Contractors or others and such inspections shall not relieve Architect of its responsibilities under the Agreement. Any Subconsultant who has prepared designs or specifications shall be responsible to conduct such observations, in accordance with the provisions of this Subparagraph, only as to those portions of the Work that he/she has so designed or specified.

designs or specifications shall be responsible to conduct such observations, in accordance with the provisions of this Subparagraph, only as to those portions of the Work that he/she has so designed a specified.
X
☐ X .5 Fabricated Materials. Check fabricated materials and equipment located on o outside the Site when such checks are required by the Contract Documents.
Applications for Payment. If and when requested by County, evaluate Applications for Payment by the Contractor and Separate Contractors and the data comprising Applications for Payment and provide recommendations (and, if requested by County, issue certifications using the current Forms G702 and G703 published by the American Institute of Architects or other forms satisfactory to County) as to whether the Work has progressed to the point indicated in such Applications for Payment and whether, to the best of Architect's knowledge based on its periodic observations conducted at the Site, the Work is in accordance with the Contract Documents; provided, however, that any of Architect's recommendations or certifications in that regard are subject to: (1) an evaluation of the Work upon Substantial Completion; (2) the results of subsequent tests and inspections; (3) minor deviations from the Contract Documents correctable prior to Final Completion; and (4) specific qualifications expressed by Architect that are consistent with its obligations under this Agreement.
X
X B Interpretations. If requested by County, issue interpretations and clarifications, in narrative form or in the form of revised Drawings or Specifications, as appropriate. Unless otherwise directed by County, such clarifications and interpretations shall be transmitted to the County in writing for distribution, as County deems appropriate, to other designated Project Team members and shall not be communicated to any other Project Team member (other than County) by Architect or its Subconsultants unless approved by County in writing.
Requests for Information. Architect shall provide responses to requests for information to the County within Five (5) Days after Architect receives such request from Contractor or a Separate Contractor, except that responses to request for information that the Contractor or a Separate Contractor has labeled as "critical" shall be provided within Two (2) Days after receipt by Architect. Unless otherwise directed by County, such responses shall be transmitted to the County in writing for distribution, as County deems appropriate, to other designated Project Team members and shall not be communicated to any other Project Team member (other than County) by Architect or its Subconsultants unless approved by County in writing.

Page 15 of 18

Edition: November 29, 2021

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

X .10 Design Changes and Clarifications. Prepare Drawings and Specifications in connection with the issuance of Construction Change Directives, Change Orders and responses to requests for information by Contractor and Separate Contractors, and where required secure approvals thereof by Governmental Authorities in accordance with Applicable Laws.
□ X .11 Assistance in Evaluation of Changes, Claims. Provide advice and support to County in evaluating the Contractor's and Separate Contractors' entitlement to additional costs, time extensions and claims and assist County in the resolution thereof.
☐ X .12 Tracking of Changes, Claims. Log, track and issue such reminders to other Project Team members as necessary to expedite processing of Change Orders, Construction Change Directives, responses to requests for information and claims.
X
X Change Orders, Construction Change Directives. Provide services for the preparation of revisions in Design Documents in connection with the issuance of Construction Change Directives and Change Orders.
X Ontractor or a Separate Contractor; provided, however, that nothing herein shall be interpreted as obligating the County to compensate Architect where the claim or any defense to the claim is based, in whole or in part, on an allegation asserted, in good faith, that Architect or a Subconsultant was negligent or that Architect has violated an obligation under this Agreement.
☐ X .16 Destruction of Work. Provide consultation and services concerning replacement of Work damaged during construction by fire or other cause.
.17 Record Documents.
Construction Documents to incorporate changes to the Work reflected in Change Orders, Construction Change Directives, Submittal revisions requested by Architect or its Subconsultants and responses by Architect or its Subconsultants to requests for information or clarification from Contractor or Separate Contractors. Such updating shall be accomplished electronically, using AutoCAD, Adobe Acrobat or other software satisfactory to the County. Architect shall transmit the updated Construction Documents to the Contractor or Separate Contractors on a frequent and regular basis, no less frequently than monthly or as often as required to maintain the progress of the Work, to allow for prompt pricing of Changes Orders and

Page 16 of 18
Edition: November 29, 2021 INITIALS JG

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

to facilitate performance by Contractor and Separate Contractors of their obligations with respect to posting and maintaining Record Drawings and Specifications.

X		
Contractors to determine if the information contained therein is internally consistent and corresponds to the condition of the visible, non-concealed conditions at the Site at the time of Final Completion. If a discrepancy is noted by Architect in the course of its review, return the Record Documents to County with a notation of the discrepancy, who shall coordinate further review and correction by the Contractor or Separate Contractors. X		Separate Contractors to determine if the assembly is complete and in the form required by the Contract
Architect's review of the visible, non-concealed conditions at the Site and other available as-built information, the Record Drawings and Specifications accurate depict the actual, as-built condition of the Project. X		Contractors to determine if the information contained therein is internally consistent and corresponds to the condition of the visible, non-concealed conditions at the Site at the time of Final Completion. If a discrepancy is noted by Architect in the course of its review, return the Record Documents to County with a notation of the discrepancy, who shall coordinate further review and correction by the Contractor or
Separate Contractors and any other information available to Architect concerning the as-built conditions of the Work (including, without limitation, any conditions noted by Architect, Subconsultants or County Consultants during the commissioning), Architect shall within Two (2) month(s) after Final Completion, convert the Contractor's and Separate Contractors' complete and corrected Record Documents to a fully consolidated and coordinated final electronic set (using CAD) of Record Documents. Such Record Documents shall be prepared by Architect's neatly inserting, accurately annotating and thoroughly cross-referencing, in a single, record set of electronically prepared Drawings and Specifications, all of the information available concerning the as-built condition of the Project. Architect shall, upon completion thereof mark each page of the Record Documents"; and (2) the Architect's manual signature certifying that: "Based upon Architect's review of the visible, non-concealed conditions at the Site and other available as-built information, these Drawings and Specifications have been approved by Architect as accurately depicting the actual, as-built condition of the Project." X		Architect's review of the visible, non-concealed conditions at the Site and other available as-built information, the Record Drawings and Specifications accurate depict the actual, as-built condition of the
calculations after Final Completion of the Project for use by County in planning, demising and allocating spaces for sale or lease. X .19 Full-Time Site Representative. Provide the services of a full-time representative at the Site. X .20 Site Office. Provide and maintain a temporary office trailer at the Site. X .21 Mitigation Monitoring and Reporting Program. In compliance with the California Environmental Quality Act (CEQA), ensure that the Mitigation Monitoring and Reporting Program (MMRP) as part of the CEQA compliance process noted previously in Paragraph 2.1.2.15, above, is implemented prior to and during ground disturbing activities (ground disturbing activities may include, but are not limited to, grading, clearing, grubbing, weed abatement, trenching, equipment staging, and other types of soil		Separate Contractors and any other information available to Architect concerning the as-built conditions of the Work (including, without limitation, any conditions noted by Architect, Subconsultants or County Consultants during the commissioning), Architect shall within Two (2) month(s) after Final Completion, convert the Contractor's and Separate Contractors' complete and corrected Record Documents to a fully consolidated and coordinated final electronic set (using CAD) of Record Documents. Such Record Documents shall be prepared by Architect's neatly inserting, accurately annotating and thoroughly cross-referencing, in a single, record set of electronically prepared Drawings and Specifications, all of the information available concerning the as-built condition of the Project. Architect shall, upon completion thereof mark each page of the Record Drawings and the cover page of the Record Specifications prominently with: (1) the words "Record Documents"; and (2) the Architect's manual signature certifying that: "Based upon Architect's review of the visible, non-concealed conditions at the Site and other available as-built information, these Drawings and Specifications have been approved by Architect as accurately
at the Site. X 20 Site Office. Provide and maintain a temporary office trailer at the Site. X 21 Mitigation Monitoring and Reporting Program. In compliance with the California Environmental Quality Act (CEQA), ensure that the Mitigation Monitoring and Reporting Program (MMRP) as part of the CEQA compliance process noted previously in Paragraph 2.1.2.15, above, is implemented prior to and during ground disturbing activities (ground disturbing activities may include, but are not limited to, grading, clearing, grubbing, weed abatement, trenching, equipment staging, and other types of soil		calculations after Final Completion of the Project for use by County in planning, demising and allocating
X .21 Mitigation Monitoring and Reporting Program. In compliance with the California Environmental Quality Act (CEQA), ensure that the Mitigation Monitoring and Reporting Program (MMRP) as part of the CEQA compliance process noted previously in Paragraph 2.1.2.15 , above, is implemented prior to and during ground disturbing activities (ground disturbing activities may include, but are not limited to, grading, clearing, grubbing, weed abatement, trenching, equipment staging, and other types of soil		X .19 Full-Time Site Representative. Provide the services of a full-time representative at the Site.
Environmental Quality Act (CEQA), ensure that the Mitigation Monitoring and Reporting Program (MMRP) as part of the CEQA compliance process noted previously in <u>Paragraph 2.1.2.15</u> , above, is implemented prior to and during ground disturbing activities (ground disturbing activities may include, but are not limited to, grading, clearing, grubbing, weed abatement, trenching, equipment staging, and other types of soil		X .20 Site Office. Provide and maintain a temporary office trailer at the Site.
	1	Environmental Quality Act (CEQA), ensure that the Mitigation Monitoring and Reporting Program (MMRP) as part of the CEQA compliance process noted previously in <u>Paragraph 2.1.2.15</u> , above, is implemented prior to and during ground disturbing activities (ground disturbing activities may include, but are not limited to, grading, clearing, grubbing, weed abatement, trenching, equipment staging, and other types of soil

Page 17 of 18

Edition: November 29, 2021 INITIALS JG

DESCRIPTION OF BASIC SERVICES, ADDITIONAL SERVICES AND EXCLUDED SERVICES

2.1.8 POST-COMPLETION PHASE

Yes

No

	X	Conduct, wit	th participation of e	each of the	Subconsultar	nts, a review o	f the Project ten (10
months	after Fi	nal Completion	on for the purpose	e of observi	ng the condit	tion in the Wo	rk. Within thirty (30
Days a	fter comp	pletion of suc	h review, make w	ritten recom	mendations	to County for t	the correction of a	anv
Defecti	ve Work	discovered.	Architect shall be	accompan	ied by Count	v during such	review of the Wo	ark
The nu	mber of	work hours	by Architect and	its Subcon	sultants to c	complete such	review and writt	ten
recomm	nendatio	ns shall not	exceed a total	of () hours. Ho	urs expended	due to condition	ons
encoun	tered that	at are attribu	table to the neglig	gence of A	rchitect or its	Subconsultar	nts or the failure	hv
Archite	ct to com	ply with this	Agreement shall i	not be inclu	ded in the af	orementioned	calculation of hor	urs
and sha	all be pro	vided at Arcl	hitect's Own Exper	nse. Hours	expended fo	r such service	s that are in exce	200
of the n	naximum	number of he	ours stated herein	shall, if and	to the extent	approved in a	dvance in writing	hv
			Additional Services		is in contoni	approved in a	avance in writing	IJ

Page 18 of 18 Edition: November 29, 2021

IN	ľ	Γl/	٩LS	JG	

INITIAL PROGRAM

A.	BUILDING ELEMENTS			
	1. 2. 3.	n/a 		
В.	AME	INITIES		
	1. 2. 3.	New Restraint Beds		
C.	SPE	CIAL FEATURES		
	1. 2. 3.	Security Screws for Ceiling T-Bar System		
D.	D. SYSTEMS PERFORMANCE CRITERIA			
	1. 2. 3.	n/a 		
E.	MATE	ERIALS		
	1. 2. 3.	Ligature Resistant Mechanical Vents Tamper Resistant Electrical Outlets Restroom fixtures (dispensers)		
F,	FINIS	HES		
	1. 2. 3.	Door Hardware (hinges, handles, closers) Safety Film for Windows Plumbing fixtures		
Э.	SPEC	IAL EQUIPMENT		
	1. 2. 3.	Rails for Drinking Fountains Anti-Ligature Camera Mounts Anti-Ligature Fire Sprinklers and Alarms		

Edition: November 29, 2021 INITIALS JG ____

MASTER PROJECT SCHEDULE

Description	Date
1. Commencement of Planning/Programming Phase	
2. Completion of Planning/Programming Phase	
3. Commencement of Schematic Design Phase	Per BOS Approval – Assuming by May 9, 2023
Completion of Schematic Design Phase	June 23, 2023
5. Commencement of Design Development Phase	June 30, 2023
6. Completion of Design Development Phase	August 25, 2023
7. Commencement of Final Construction Documents Phase	September 1, 2023
50% Completion Submission Date	October 20, 2023
75% Completion Submission Date	N/A
Submission to Governmental Authorities Date	November 24, 2023
Final Construction Documents Completion Date (HCAI Approval)	May 17, 2024
8. Commencement of Bidding/Negotiation Phase	Per approval of HCAI – Assuming May 17, 2024
9. Completion of Bidding/Negotiation Phase	June 21, 2024
10. Commencement of Construction (24 Weeks)	June 28, 2024
 Construction Period (Estimated only. Actual construction period will be established when the Construction Contract is executed by Contractor and County) 	6 months

Edition: November 29, 2021 INITIALS JG

KEY PERSONNEL LIST

Name: Job Title:

John Greenwood Architect / Principal PVG Architects

Johng@pvgarchitects.com 760-779-5393

Steve Corner PVG Architects stevec@pvgarchitects.com 760-779-5393

Project Manager

Edition: November 29, 2021

INITIALS

JG

SUBCONSULTANTS AND COUNTY CONSULTANTS LIST

Subconsultants retained by Consultant:

Edition: November 29, 2021

Mechanical	Electrical	
Name: Marc Anderson Linwood Engineering Address: 2301 Dupont Dr. Suite Irvine, CA 92612 Telephone: 949-214-4302 Facsimile:		Name: Address: a, CA, 91739 Telephone: Facsimile:
Name:	Name:	Name:
Address:	Address:	Address:
Telephone: Facsimile:	Telephone: Facsimile:	Telephone: Facsimile:
County Consultants retained County:	l by	
Personal filtration and the design of the control o	Exercise (activity) and the last as a subsequence of the last as a subsequ	
Name:	Name:	Name:
Address:	Address:	Address:
Telephone: Facsimile:	Telephone: Facsimile:	Telephone: Facsimile:
Name:	Name:	Name:
Address:	Address:	Address:
Telephone: Facsimile:	Telephone: Facsimile:	Telephone: Facsimile:
Name:	Name:	Name:
Address:	Address:	Address:
Telephone:	Telephone: Facsimile:	Telephone: Facsimile:

REIMBURSABLE EXPENSES SCHEDULE

Description	Unit Price	Unit of Measure
Agreed Rates for Reimbursable Expenses:		
Copies	\$0.15	/Sheet
Blueprints	\$0.68	/S.F.
Vellums	\$0.00	/S.F.
Sepias	\$0.00	/S.F.
Disks	\$0.00	EA.
Others:	\$0.00	
Vehicle Mileage	\$0.56	mile
Agreed Markups:		
Markup on Direct Costs	10%	Actual Cost
Markup on Subconsultant Costs	%	Actual Cost
Subconsultant Markup on Subconsultant Costs	%	Actual Cost
Reimbursable Expenses Not-to-Exceed Amount (individual):	\$500	0.00
Reimbursable Expenses Not-to-Exceed Amount (aggregate):	\$800	0.00

Edition: November 29, 2021

HOURLY RATES SCHEDULE

 Clerical
 \$75.00
 /Hr.

 CAD
 \$130.00
 /Hr.

 Principal
 \$265.00
 /Hr.

 Project Manager
 \$210.00
 /Hr.

Edition: November 29, 2021

PAYMENT SCHEDULE

Phase:

Payment Percentage:

Planning/Programming Phase	NA
Schematic Design Phase	10 %
Design Development Phase	25%
Final Construction Documents Phase	45%
Bidding Phase	0%
Construction Phase	20%
Post-Completion Phase	0%

Edition: November 29, 2021

EXHIBIT "J "

REFERENCE DOCUMENTS LIST

The Architect shall take into consideration in its preparation of the Design Documents the constraints, requirements and recommendations contained in the existing studies, reports and documents provided by the County or otherwise available as a matter of public record, including but not limited to the following Reference Documents:

Reference Document:	Prepared By: Dated:
1. ALTA Survey	NA
2. Boundary Survey	NA
3. Topographic Map	NA
4. Soils/Geotechnical Report	NA
5. Master Plan	NA
6. Zoning Plan	NA
7. CC&R's	NA
8. Traffic Report	NA
9. Acoustical Report	NA
10. As-Built Documents for Existing Improvements	TO BE PROVIDED TO ARCHITECT
11. Environmental Impact Report	NA
12. Hazardous Materials Survey(s)	TO BE PROVIDED TO ARCHITECT
a. Phase I Environmental Site Assessment (ESA)	NA
b. Phase II (if applicable)	NA
c. Abatement Reports (if applicable)	TO BE PROVIDED TO ARCHITECT
13.CEQA compliance documentation and associated Mitigation Monitoring and Reporting Program (which may include, but not be limited to, other technical studies such as Air Quality, Biological Resources, and Cultural Resources)	NA
13 [Others]	NA

Edition: November 29 2021 INITIALS JG

STANDARD FORM OF GENERAL CONDITIONS

(Attach General Conditions of the Standard Form of Construction Contract Between County and Contractor)

Design Only Contract for the Subject Project

Edition: November 29, 2021

ADDITIONAL INSUREDS LIST

County of Riverside shall be named as Additional Insured. County of Riverside herein refers to County of Riverside, its Agencies, Districts, Special Districts, and Departments, their respective directors, officers, Board of Supervisors, employees, elected or appointed officials, agents or representatives.

Edition: November 29, 2021 INITIALS JG

EXHIBIT M RELEASE FORMS

NA

Edition: November 29, 2021

CONDITIONAL WAIVER AND RELEASE ON PROGRESS PAYMENT

(CA Civil Code §8132)

NOTICE: THIS DOCUMENT WAIVES THE CLAIMANT'S LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS EFFECTIVE ON RECEIPT OF PAYMENT. A PERSON SHOULD NOT RELY ON THIS DOCUMENT UNLESS SATISFIED THAT THE CLAIMANT HAS RECEIVED PAYMENT.

	Identifying Information:
Name	e of Claimant:
Name	e of Customer:
Job L	ocation:
Owne	er:
Throu	gh Date:
	Conditional Waiver and Release
has for through or ma prior to unless	locument waives and releases lien, stop payment notice, and payment bond rights the claimant or labor and service provided, and equipment and material delivered, to the customer on this job in the Through Date of this document. Rights based upon labor or service provided, or equipment terial delivered, pursuant to a written change order that has been fully executed by the parties of the date that this document is signed by the claimant, are waived and released by this document, is listed as an Exception below. This document is effective only on the claimant's receipt of the tent from the financial institution on which the following check is drawn:
	Maker of Check:
	Amount of Check: \$
	Check payable to:
	Exceptions
This do	ocument does not affect any of the following:
(1) (2) (3)	Retentions. Extras for which the claimant has not receive payment. The following progress payments for which the claimant has previously given a conditional waiver and release but has not received payment:
	Date(s) of waiver and release: Amount(s) of unpaid progress payment(s): \$
(4)	Contract rights, including (A) a right based on rescission, abandonment, or breach of contract, and (B) the right to recover compensation for work not compensated by the payment.
	Signature
Dated:	Claimant's Signature
	Claimant's Title

Edition: November 29, 2021

UNCONDITIONAL WAIVER AND RELEASE ON PROGRESS PAYMENT (CA Civil Code §8134)

NOTICE TO CLAIMANT: THIS DOCUMENT WAIVES AND RELEASES LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS UNCONDITIONALLY AND STATES THAT YOU HAVE BEEN PAID FOR GIVING UP THOSE RIGHTS. THIS DOCUMENT IS ENFORCEABLE AGAINST YOU IF YOU SIGN IT, EVEN IF YOU HAVE NOT BEEN PAID. IF YOU HAVE NOT BEEN PAID, USE A CONDITIONAL WAIVER AND RELEASE FORM.

	identifying information:		
Name	of Claimant:		
Name	of Customer:		
	ocation:		
	r:		
	gh Date:		
	Unconditional Waiver and Release		
This document waives and releases lien, stop payment notice, and payment bond rights the claimant has for labor and service provided, and equipment and material delivered, to the customer on this job through the Through Date of this document. Rights based upon labor or service provided, or equipment or material delivered, pursuant to a written change order that has been fully executed by the parties prior to the date that this document is signed by the claimant, are waived and released by this document, unless listed as an Exception below. The claimant has received the following progress payment:			
	\$		
	Exceptions		
This do	ocument does not affect any of the following:		
(1) (2) (3)	Retentions. Extras for which the claimant has not received payment. Contract rights, including (A) a right based on rescission, abandonment, or breach of contract, and (B) the right to recover compensation for work not compensated by the payment.		
	Signature		
Dated:	Claimant's Signature		
	Claimant's Title		

Edition: November 29, 2021

CONDITIONAL WAIVER AND RELEASE ON FINAL PAYMENT

(CA Civil Code §8136)

NOTICE: THIS DOCUMENT WAIVES THE CLAIMANT'S LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS EFFECTIVE ON RECEIPT OF PAYMENT. A PERSON SHOULD NOT RELY ON THIS DOCUMENT UNLESS SATISFIED THAT THE CLAIMANT HAS RECEIVED PAYMENT.

Identifying Information:
Name of Claimant:
Name of Customer:
Job Location:
Owner:
Conditional Waiver and Release
This document waives and releases lien, stop payment notice, and payment bond rights the claimant has for labor and service provided, and equipment and material delivered, to the customer on this job Rights based upon labor or service provided, or equipment or material delivered, pursuant to a written change order that has been fully executed by the parties prior to the date that this document is signed by the claimant, are waived and released by this document, unless listed as an Exception below. This document is effective only on the claimant's receipt of payment from the financial institution on which the following check is drawn: Maker of Check: Maker of Check:
Amount of Check: \$
Check payable to:
Exceptions
This document does not affect any of the following:
Disputed claims for extras in the amount of: \$
Signature
Dated: Claimant's Signature
Claimant's Title

Edition: November 29, 2021

UNCONDITIONAL WAIVER AND RELEASE UPON FINAL PAYMENT

(CA Civil Code §8138)

NOTICE TO CLAIMANT: THIS DOCUMENT WAIVES AND RELEASES LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS UNCONDITIONALLY AND STATES THAT YOU HAVE BEEN PAID FOR GIVING UP THOSE RIGHTS. THIS DOCUMENT IS ENFORCEABLE AGAINST YOU IF YOU SIGN IT, EVEN IF YOU HAVE NOT BEEN PAID. IF YOU HAVE NOT BEEN PAID, USE A CONDITIONAL WAIVER AND RELEASE FORM.

Identifying Information:

Name of Claimant:
Name of Customer:
Job Location:
Owner:
Unconditional Waiver and Release
This document waives and releases lien, stop payment notice, and payment bond rights the claimant has for all labor and service provided, and equipment and material delivered, to the customer on this job. Rights based upon labor or service provided, or equipment or material delivered, pursuant to a written change order that has been fully executed by the parties prior to the date that this document is signed by the claimant, are waived and released by this document, unless listed as an Exception below. The claimant has been paid in full.
Exceptions
This document does not affect any of the following:
Disputed claims for extra in the amount of: \$
Signature
Dated: Claimant's Signature
Claimant's Title

Edition: November 29, 2021

INVOICE FOR PAYMENT FORM

Edition: November 29, 2021

то:	Facili	a Rodriguez lities Management-PMO		INVOICE DATE:			
	44199 Monroe Street, Suite B Indio, CA 92201			INVOICE NUMBER:			
FROM: (760) 863-2537 Prest Vuksic Greenwood 44530 San Pablo Avenu Suite 200 Palm Desert, CA 92260			nue	DJECT NUMBER	₹:	FM054100	011587
PROJECT:	Indio I Comp 47915	779-5393 RUHS ITF Anti-Li liance Project 6 Oasis Street CA 92201	gature SER	EVICES THROUG	GH:		
DESCRIPTION		CONTRACT	PERCENT COMPLETE	COMPLETE TO DATE	PREVIOUSLY BILLED		CURRENTLY DUE
Schematics Design Documents Construction Documents Construction Administration Reimbursable Expenses		\$17,730.00 \$43,325.00 \$78,785.00 \$35,460.00 \$800.00					301
TOTAL		\$176,100.00					***************************************
		Printed					
oignature:		Printed	Name:			Date:	
Accounting Department Batch #: MM/YY: JE: Credit GL #: Debit GL #:		l Use		Facilities Department Use GL Business Unit: Project ID #: Activity Code: Resource Code: Category Code:		e	
Entered by: Approved by:	_	and the same of th		Description:		***	
				Preparer's Nar	ne:	[Preparer]	

CONSTRUCTION DOCUMENT DELIVERABLES

THE CONSTRUCTION DOCUMENTS PREPARED AND DELIVERED BY ARCHITECT AS PART OF BASIC SERVICES DURING THE DESIGN DEVELOPMENT PHASE AND FINAL CONSTRUCTION DOCUMENTS PHASE SHALL INCLUDE, WITHOUT LIMITATION TO THE OTHER PROVISIONS OF THE AGREEMENT AND ITS EXHIBITS, THE CONSTRUCTION DOCUMENTS SET FORTH BELOW.

Yes	No

- X 2.5.1 Off-Site Improvements. Drawings delineating off-Site improvements (including, without limitation, road and infrastructure improvements adjacent to proposed facility) such as, but not limited to, the following: (1) Work area limits; (2) existing street plan with stationing fully dimensioned, including, but not limited to, roadway profiles, paving sections, curb and gutter, sidewalk with all utility pull boxes in sidewalk, utility pull boxes in landscape areas, curb inlets, fire hydrants, light poles, bus stops, drive approaches, medians, striping, signage, signalization, water, electric (underground and overhead), sewer, storm drain, gas and cable services in street; (3) new street plan with stationing fully dimensioned, including, but not limited to, roadway profiles, paying sections, curb and gutter, sidewalk with all utility pull boxes in sidewalk, utility pull boxes in landscape areas, vaults, curb inlets, fire hydrants, light poles, bus stops, drive approaches, medians, striping, signage, signalization, water, electric (underground and overhead), sewer, storm drain, gas and cable services in street; (4) locations, sizes and dimensions (including, without limitation, lay-out points, elevations and curve data) of hardscape and softscape, including but not limited to new and existing curb and gutter, sidewalk with all utility pull boxes in sidewalk, utility pull boxes in landscape areas, vaults, curb inlets, fire hydrants, light poles, bus stops, drive approaches, medians, striping, signage, signalization, water, electric (underground and overhead), sewer, storm drain, gas and cable services in streets; (5) routing, sizing, locations and points of connection for utilities (including but not limited to gas, water and electrical) storm sewer and sanitary sewer systems; (6) designs and details for structural features, including, without limitation, paying section with base, bus pads, curb and gutter, retaining structures, light and signal pole bases, traffic signal equipment bases, guardrail, drive approaches, catch basins and storm drain structures; and (7) traffic control and Work phasing plans approved by local street authority.
- Z.5.2 Site. Drawings delineating (1) Work area limits; (2) horizontal and vertical controls and final building locations and dimensions; (3) locations and dimensions of property lines, easements and setback requirements; (4) locations, sizes and dimensions (including, without limitation, layout points, elevations and curve data) of hardscape and softscape (including, but not limited to, storm drainage systems, service drives, loading docks, walkways, ramps, steps, stairs, patios, courtyards, roadways, entrances, exits, seat walls, retaining walls, planters, furnishings, signage and other wayfinding elements, plantings, catch basins, clean outs, manholes, fixtures, equipment and vaults); (5) routing, sizing, locations and points of connection for utilities (including, but not limited to, gas, water and electrical), low voltage systems and storm, sewer and sanitary sewer systems; (6) designs and details for structural Site features, such as, but not limited to, canopies, retaining walls and planters; and (7) water control measures, including, without limitation, weirs and retention basins.
- □ X 2.5.3 Demolition. Drawings, prepared in a manner that clearly and graphically distinguishes between existing improvements to be removed and those to remain in place, delineating: (1) Site clearance and Site demolition; (2) general and selective architectural, structural, mechanical, electrical, plumbing, and fire protection demolition; (3) means for existing building systems affected by demolition to remain operable; and (4) exterior elevations showing the extent of any building envelope to be removed.

Page 1 of 4

Edition: November 29, 2021

CONSTRUCTION DOCUMENT DELIVERABLES

Yes No	
elevations (building sys in casework low voltage	2.5.4 Architectural, Structural. Drawings (for an entire floor, for all segments and enlarged for selected rooms), fully dimensioned, delineating the locations and (above finish floor) of all architectural and structural elements fully coordinated with stems designs, including without limitation, building grid, walls, partitions, columns, builts, finish floor patterns and layouts, ceiling material patterns, reveals, soffits and accents, devices, mechanical diffusers, registers, grilles, fixtures, occupancy and fire and room and department identifications.
equipment p	2.5.5 Roof. Drawings delineating locations, sizes and dimensions of all s and interruptions (major and minor), roof access, penthouses, equipment screens, pads, roof mounted equipment, slopes, ridges, valleys and crickets (including slope walking pads, roof drains and vents and roofing materials.
and roof ele	2.5.6 Exterior Elevations. Drawings, fully dimensioned, delineating building notuding, without limitation, elevations of all exterior surfaces and materials, finish floor exations, exterior wall penetrations and building mounted fixtures and equipment and mmodations for inter-story drift.
Site improve	2.5.7 Building Section(s). Drawing(s) delineating the cross sections of: (1) the the relationship among the levels of the building and the building and grade levels; (3) ements, roadways, property lines and adjacent structures; and (4) exterior walls without limitation, exterior wall materials, structural support and building façade
□ X without limita signage, and	2.5.8 Signage . Drawings delineating signage and other wayfinding, including, ation, signage required by Applicable Laws, directional signage, room identification a project sign including such information as directed and approved by County.
□ X windows and floor, wall and	2.5.9 Equipment, Fixture and Materials Schedules. Schedules for doors and I related openings and hardware (cross-referenced to frame installation details) and d ceiling finish materials (including, without limitation, bases and wainscots).
sizes, assemi and thermal i	2.5.10 Partition Details. Drawings delineating partition details, including, without standards for fire rating, acoustic and thermal requirements, structural materials and bly and connection details, enclosed materials (such as, but not limited to, acoustic insulation) and backing for equipment; (2) identification of each partition by partition enlarged partition cross sections.
casework sch representation	2.5.11 Interior Details . Drawings delineating, without limitation,: (1) interior cluding millwork and casework cross-referenced to the architectural floor plans; (2) redule; (3) casework sections and millwork details; (4) dimensioned details; and (5) n of the terminal devices for engineered systems, including mechanical, electrical, protection and low voltage systems.
isolation detail	2.5.12 Exterior Details . Drawings referenced to roof plan and building exterior cluding, without limitation, roof assemblies, exterior surface penetrations, vibration is, seismic separation joint details, inter-story drift provisions, flashings and transition erial and plane changes.

Page 2 of 4

Edition: November 29, 2021

INITIALS JG

CONSTRUCTION DOCUMENT DELIVERABLES

<u>Yes</u>	<u>No</u>	
of Site	element	2.5.13 Landscaping. Drawings delineating, without limitation: (1) plantings; (2) ns; (3) Site signage and wayfinding elements; (4) Site illumination; and (5) details (including, without limitation, detailed and integrated architectural and structural structures, planters, retaining walls, walks and other hardscape).
□ delinea (founda	X ting four ations, co	2.5.14 Foundation and Framing . Drawings (fully dimensioned) and schedules addition and framing for each level of structure, sizing of all structural elements dumns, horizontal members) and design of vertical and horizontal diaphragms.
□ elemen	X ts comply	2.5.15 Structural Calculations. Calculations demonstrating that all structural y with the requirements of Governmental Authorities and Applicable Laws.
does no been pro such ce	braced frot follow the tested ertification	2.5.16 Structural Frame Details. Drawings delineating: (1) moment frame ame elevations and connection details; and (2) if the proposed structural system the criteria of Applicable Laws, written certification that all such connections have and pre-approved by the applicable Governmental Authorities or, in the event that is not possible, a proposed comprehensive testing program which meets all Governmental Authorities.
without	limitation	2.5.17 Bracing Details. Drawings delineating details for bracing, including, n, bracing of partitions, ceilings, mechanical, electrical, plumbing and other g, ductwork, conduit, backing for casework, furniture, furnishings, and artwork.
mechani	ical, elec ion on ec	2.5.18 MEP Schedules . Schedules (cross-referenced to single line diagrams) of strical, plumbing, fire protection and low voltage equipment and devices and full quipment parameters relative to sizing, connections, mountings, construction and
complete schemat	e archited ic routing	2.5.19 MEP Distribution . Single line Drawings (floors and roof) set over a ctural background delineating, on a floor-by-floor basis, complete information on g, equipment and distribution locations and sizing and points of connection (above y ground) for:
	X and spec equireme	ial exhaust requirements, dedicated equipment and ductwork for rooms with
panels ar boards, b equipmer	g, withound switch ous ducts out and lig	basic (and, if required by Applicable Laws, emergency) electrical systems at limitation, wiring, raceways, conduit, lighting, main service and distribution agear, motor control centers, generators, transformers, transfer switches, panel is, locations and types of all convenience outlets, lighting fixtures and switches, phting to be connected to the emergency power system, and locations of power nical, plumbing and fire protection equipment and low voltage devices);
Xoof and nechanic generator	site stor	plumbing system (including, without limitation, domestic hot/cold water, and drainage, sanitary waste and vent, sanitary, hot and chilled water for the em, water treatment (water softening), natural gas and fuel for auxiliary

Page 3 of 4
Edition: November 29, 2021
INITIALS JG

CONSTRUCTION DOCUMENT DELIVERABLES

Yes No
☐ X .4 fire protection detection system information (locations for signaling at alarm devices and main equipment), fire suppression system information (locations for points connection and main equipment) and a design-build Specification for fire detection and suppression systems; and,
☐ X .5 low voltage systems, including, without limitation, main equipment distribution, and terminal devices (wayfinding and warning).
□ X 2.5.20 MEP Equipment Rooms. Drawings delineating the layout and sizing equipment rooms and penthouses, including, without limitation, complete information and sizing fe equipment, ductwork, piping, bracing, vibration isolation, supports, attachments, chases and paddenic process.
☐ X 2.5.21 MEP Shaft Layout. Drawings delineating all of the ductwork, electric conduits and piping in shared shafts.
□ X 2.5.22 Existing MEP Capacities. A program to validate that the proposed desig for new mechanical, electrical, plumbing and fire protection systems will be within the capacity of the existing building systems.
X D 2.5.23 Energy Calculations. Final energy calculations demonstrating that the design complies with the requirements of Title 24 of the California Code of Regulations.
X D 2.5.24 Code Requirements. Drawings dedicated to demonstrating to Governmental Authorities compliance of new Work (and, in the case of renovation, Existing Improvements) with Applicable Laws pertaining to fire and life/safety, including, without limitation exits, exit paths, travel distances, rated corridors and other rated walls, occupancy classifications occupancy loads, construction classifications/separations, fire zones/smoke compartments and fire protection.
X D 2.5.25 Specifications. A set of detailed Specifications, customized for the Project, following the Construction Specification Institute format, or such other format that is acceptable to County.
X ☐ 2.5.26 Index. A detailed index of the Construction Documents, including, without limitation, the title and sheet number of every Drawing sheet.
X D 2.5.27 Legends. A comprehensive legend of symbols, notes and abbreviations.
X 2.5.28 Stormwater Pollution Prevention Plan. A Construction Document that specifies Best Management Practices (BMPs) that will prevent all construction pollutants from contacting storm water and with the intent of keeping all products of erosion from moving off Site nto receiving waters.

Page 4 of 4

Edition: November 29, 2021

SUBCONSULTANT INSURANCE REQUIREMENTS

Subconsultants are required to maintain insurance on the same terms and with the same coverages as required of Architect under <u>Section 10.1</u> of the Agreement, except that specific coverage amounts for Commercial General Liability and Professional Liability Insurance are hereby modified for the following Subconsultants:

Linwood Engineering

Edition: _____, 2021

			_
	1.	Com	mercial General Liability:
		00,000 00,000	General Aggregate (Other Than Products-Completed Operations) Products-Completed Operations Aggregate Limit for a period of five (5) years
		00,000 00,000	following Final Completion and Acceptance of the Project Personal and Advertising Injury Limit Per Occurrence Limit
	.2	Profe	ssional Liability \$1,000,000 per claim and \$2,000,000 in the annual aggregate
RASC	C Engine	eering	
	1.	Comn	nercial General Liability:
	\$1,00 \$1,00		General Aggregate (Other Than Products-Completed Operations) Products-Completed Operations Aggregate Limit for a period of five (5) years following Final Completion and Acceptance of the Project
	\$1,000 \$1,000		Personal and Advertising Injury Limit Per Occurrence Limit
	.2	Profes	ssional Liability \$1,000,000 per claim and \$2,000,000 in the annual aggregate.
		[Nai	me of Subconsultant]
	1.	Comm	nercial General Liability:
	\$1,000 \$1,000		General Aggregate (Other Than Products-Completed Operations) Products-Completed Operations Aggregate Limit for a period of five (5) years following Final Completion and Acceptance of the Project
	\$1,000 \$1,000		Personal and Advertising Injury Limit Per Occurrence Limit
	.2	Profes	sional Liability \$1,000,000 per claim and \$2,000,000 in the annual aggregate.
***		[Nan	ne of Subconsultant]
	1.	Comm	ercial General Liability:
	\$1,000, \$1,000,		General Aggregate (Other Than Products-Completed Operations) Products-Completed Operations Aggregate Limit for a period of five (5) years following Final Completion and Acceptance of the Project
	\$1,000, \$1,000,		Personal and Advertising Injury Limit Per Occurrence Limit
	.2	Profess	sional Liability \$1,000,000 per claim and \$2,000,000 in the annual aggregate.

INITIALS ____